DEVELOPMENT OF WALL PUTTY FOR TROPICS USING DRINKING WATER TREATMENT PLANT WASTE ALUM SLUDGE

Galkanda Arachchige Himahansi Hasinthara Galkanda

188060U

Degree of Doctor of Philosophy in Civil Engineering

Department of Civil Engineering

University of Moratuwa Sri Lanka

September 2022

DEVELOPMENT OF WALL PUTTY FOR TROPICS USING DRINKING WATER TREATMENT PLANT WASTE ALUM SLUDGE

Galkanda Arachchige Himahansi Hasinthara Galkanda

188060U

Thesis submitted in partial fulfillment of the requirements for the Degree of Doctor of Philosophy in Civil Engineering

Department of Civil Engineering

University of Moratuwa Sri Lanka

September 2022

DECLARATION OF CANDIDATE AND SUPERVISOR

"I declare that this is my own work and this thesis does not incorporate without acknowledgment any material previously submitted for a Degree or Diploma in any other University or institute of higher learning and to the best of my knowledge and belief it does not contain any material previously published or written by another person except where the acknowledgment is made in the text.

Also, I hereby grant to the University of Moratuwa the non-exclusive right to reproduce and distribute my thesis, in whole or in part in print, electronic or other media. I retain the right to use this content in whole or part in future works (such as articles or books)."

Signature of Candidate:

Date:

The above candidate has carried out research for the Ph.D. thesis under my supervision.

.....

Prof. R. U. Halwatura Department of Civil Engineering University of Moratuwa

Date:

.....

Prof. A. H. L. R. Nilmini Faculty of Technology University of Sri Jayewardenepura

Date:

ABSTRACT

Water is the main source of life; therefore, a sufficient amount of safe water consumption is essential for public health. It is one of the responsibilities of a country, to ensure the access of its citizens to consume sufficient safe water. In Sri Lanka, the national water supply and drainage board (NWSDB) is the responsible authority for drinking water purification and distribution. NWSDB owned 323 water supply schemes. In drinking water treatment plants, surface water is collected and treated to drinking quality removing impurities dissolved in surface water. In the purification process aluminium sulphate $(Al_2(SO_4)_3)$ (Alum) is used as the coagulant and generated alum sludge at the end. The sludge disposal to surface water bodies creates the undesirable formation of mud deposits according to the activation of alum. Direct discharge of sludge into water bodies creates damage to its' creatures and ecosystems. Therefore, direct disposal of alum sludge in open lands and water bodies is prohibited by legislation. Dewatered alum sludge is disposed at landfills and rock blasting wells. However, the increasing daily generation of sludge is creating an urgent necessity for a sustainable solution. The increasing amount of daily alum sludge production has considerable environmental and economic concerns in most countries. Therefore, the world's attention turned to finding a sustainable way to reuse or recycle DWTP alum sludge. This research aimed to address the issues mentioned, by developing a wall putty using waste alum sludge generated in drinking water treatment plants (DWTP) in Sri Lanka. Properties of the DWTP waste sludge differ according to the climatic conditions, geographical conditions, water treatment process and raw water quality. The research was conducted after the identification of the properties of DWTP waste alum sludge of different plants. Laboratory experiments were conducted to study the properties of sludge samples collected from DWTPs in Ambathale, Biyagama and Kandana. Biyagama DWTP was selected to collect sludge for the study due to the low moisture content and high solid content compared to other samples. Waste sludge is discharged at the end of the water treatment process, in semisolid form with high moisture content and it is dewatered through a sludge treatment process in Biyagama DWTP. Dewatered sludge generation of the plant is estimated at 10m³ per day. Properties of DWTP waste alum sludge were studied. According to the results, moisture content variation, volumetric shrinkage variation, chemical composition and heavy metal analysis of the sludge was analyzed. In the first phase of the study, experiments were conducted to develop a wall putty mix using wet alum sludge. Test results reveal that volumetric shrinkage can be reduced with physical additives and adhesiveness can be improved with binders, but a wall putty mix cannot be developed with wet sludge by mixing additives and a binder, due to the high moisture content, high shrinkage and alum activation. In the second phase of the study, experiments were conducted to develop a dry powder from wet sludge overcoming the alum activation. To that thermal alterations of DWTP waste alum sludge were studied. Colour and density variations of the burned sludge at different temperatures were studied. Sludge becomes harder when burned, due to the alum activation and none of the processes that exist in the world, to produce dry powder from DWTP alum sludge. Alum activation of the sludge can be overcome by burning sludge with a lubricant. According to the experiments, coconut oil is identified as an effective lubricant. The density of burned sludge with oil is lower than that of burned sludge without oil at each temperature. Finally, a process was developed to produce dry powder from DWTP alum sludge. In the final phase of the study putty properties of the developed dry powder were analyzed and optimized and the performances of the developed wall putty mix were analyzed compared to existing wall finishes. And the real scale performances of the putty were tested. Finally, it was concluded that the developed dry powder is applicable as a wall finisher successfully on both interior and exterior walls. And also new research areas were identified for further studies from this research.

Keywords: Alum sludge, Drinking water treatment plant, Sludge putty, Tropical climate

ACKNOWLEDGMENT

I wish to express my first and deepest gratitude to Prof. R.U. Halwathura, Faculty of Engineering, University of Moratuwa for his kind corporation, assistance, supervision and technical knowledge given throughout the research study as the supervisor to complete this research successfully. And also, I like to acknowledge the valuable support and guidance provided by Prof. A.H.L.R Nilmini, Faculty of Technology, University of Sri Jayewardenepura as the co-supervisor.

I express my heartfelt thanks to Dr. N.G.R. Perera (Senior Lecturer, Dept. of Architecture), Dr. N. Wijayaratna (Research Coordinator, Department of Civil Engineering) and former research coordinators of the Department of Civil Engineering for the continuous guidance and comments provided as examiners of the progress review panel. And I like to acknowledge the support provided by Prof. C. Jayasinghe (Head, Department of Civil Engineering) and all the academic and non-academic staff of the Department of Civil Engineering throughout the research period.

I offer my special thanks to all the Technical Officers and Lab Attendants of the Department of Civil Engineering and Department of Materials Science and Engineering, University of Moratuwa for the support provided to complete the laboratory experiments of this research. And I offer my sincere thanks to Ms. Priyantha, Ms. Rukma and Naween in the Construction Management Division for the support provided during the research period.

I offer my sincere thanks to all the members of the Pro-Green laboratory for the valuable support provided to complete this research.

I would like to thank, Mr. R. Wijekoon (Manager, Biyagama Water Treatment Plant), Mr. Disanayaka (NWSDB), Mr. Dhammika (NWSDB) and all the staff members of the National Water Supply and Drainage Board Sri Lanka for granting permission and support to collect material and data required for the research. And also, I offer my special thanks to all the staff members of the Biyagama Drinking Water Treatment Plant for the support provided to collect sludge samples throughout the research.

I would like to thank all who contributed to successfully complete this research.

Finally, I like to offer my heartfelt thanks to my loving husband, beloved parents and two sisters for being with me throughout this research period and providing encouragement and valuable contributions.

TABLE OF CONTENTS

DECLARATION OF CANDIDATE AND SUPERVISOR	i
ABSTRACT	ii
ACKNOWLEDGMENT	iii
LIST OF FIGURES	viii
LIST OF TABLES	xi
LIST OF APPENDICES	xii
LIST OF ABBREVIATIONS	xiii
CHAPTER 01: INTRODUCTION	1
1.1. General	1
1.2. Research Gap	3
1.3. Aim and Objectives	4
1.4. Methodology	5
1.5. Main Findings	6
1.6. Organization of the Thesis	8
CHAPTER 02: LITERATURE REVIEW	9
2.1. General	9
2.2. Drinking Water Treatment	9
2.2.1. Drinking water treatment process	10
2.2.2. Drinking water treatment in Sri Lanka	12
2.3. DWTP Waste Sludge	14
2.3.1. DWTP sludge disposal methods	14
2.3.2. DWTP sludge applications	16
2.4. Sustainable Building Construction and Materials	19
2.4.1. Sustainable building constructions	20
2.4.2. Sustainable construction materials	21
2.4.3. Walling materials	22
2.5. Wall Finishes and Technologies	24
2.5.1. Historical evidence of earthen wall finishes in the world	24
2.5.2. Historical evidence of earthen wall finishes in Sri Lanka	26
2.5.3. Present wall finishes and technologies	28
2.5.4. Wall care putty	28
2.6. Soil-based Wall Finishes	30
2.6.1. Research conducted on soil-based wall finishes	30
2.6.2. Test standards for soil-based wall finishes	34
2.7. Characteristics of Wall Care Putty Products	35

2.7.1. Durability of wall finishes	35
2.7.2. Tropical climate and rain surface erosion	36
2.7.3. Bio-receptivity of wall finishes	37
2.7.4. Fungus growth on building walls	37
2.8. Summary	39
CHAPTER 03: SELECTION OF DWTP TO COLLECT SLUDGE	42
3.1. General	42
3.2. Selection of a DWTP	42
3.2.1. Ambathale DWTP	44
3.2.2. Kandana DWTP	44
3.2.3. Biyagama DWTP	44
3.3. Experimental Analysis of Sludge Properties	46
3.3.1. Moisture and solid content analysis	46
3.3.2. Wet Sieve Analysis	48
3.4. Summary	50
CHAPTER 04: ANALYSING PUTTY PROPERTIES OF DWTP WASTE ALUM SLUDGE	51
WASTE ALOM SECOCE	
4.1. General	51
4.2. Chemical Properties of DWTP Waste Alum Sludge	51
4.2.1. XRF Analysis	51
4.2.2. XRD Analysis	53
4.2.3. Heavy Metal Analysis	54
4.3. Observational Shrinkage of DWTP Waste Alum Sludge	58
4.4. Physical Properties of DWTP Waste Alum Sludge	61
4.4.1. Shrinkage variation with chemicals and climatic factors	62
4.4.2. Moisture content variation with chemicals and climatic factors	66
4.5. Summary	69
CHAPTER 05: OPTIMIZING PUTTY PROPERTIES OF DWPT WET ALUM SLUDGE	70
	70
5.1. General	70
5.2. Selection of Additives	70
5.3.1 Analyzing the affectiveness of send	71
5.3.2. Volumetric shrinkage variation of wet alum sludge	71
with sand	12
5.3.3. Analyzing optimum sand/wet alum sludge mixing ratio (Ball drop test)	/4
5.3.4. Analyzing the effectiveness of natural fibers	76

5.3.5. Volumetric shrinkage variation of wet alum sludge	77
5.4 Binders to Improve Putty Properties	79
5.4.1. Analyzing putty properties variation of wet alum	81
sludge with binders	0.1
5.4.2. Moisture content variation of wet sludge with binders	81
5.4.3. Observational shrinkage variation of wet alum sludge with binders	82
5.4.4. Analyzing fungus growth of wet alum sludge with binders	83
5.5. Analyzing Putty Properties of Wet Alum Sludge Mixture	85
5.5.1. Volumetric shrinkage variation with additives	86
5.5.2. Observational shrinkage variation with additives	87
5.5.3. Wet alum sludge mixture with optimum putty properties	88
5.6. Summary	89
CHAPTER 06: DEVELOPMENT OF DRY POWDERED ALUM SLUDGE	91
6.1. General	91
6.2. Thermal Properties of DWTP Waste Alum Sludge	92
6.2.1. Color Variation	93
6.2.2. Density Variation	94
6.3. Overcome Alum Activation of DWTP Waste Alum Sludge	96
6.3.1. Selection of a lubricant	97
6.3.2. Dry powdered sludge production with coconut oil	98
6.3.3. Identification of effectiveness of coconut oil as a lubricant	99
6.3.4. Identification of optimum temperature	100
6.3.5. Identification of optimum coconut oil mixing ratio	101
6.4. Putty properties of dry powdered alum sludge	103
6.4.1. Analyzing observational shrinkage	104
6.4.2. Analyzing effectiveness of stabilizer	105
6.5. Summary	108
CHAPTER 07: DEVELOPMENT OF WALL PUTTY AND REAL- SCALE APPLICATION	110
7.1. General	110
7.2. Selection of Stabilizer	111
7.2.1. Bonding strength variation with stabilizers	112
7.2.2. Bio receptivity variation with stabilizers	117
7.2.3. Durability variation with stabilizers	120

7.3. Comparison with Existing Wall Finishes	124
7.3.1. Material selection	125
7.3.2. Bonding strength comparison of different wall finishes	127
7.3.3. Bio receptivity comparison of different wall finishes	129
7.3.4. Durability comparison of different wall finishes	131
7.4. Real Scale Application	133
7.4.1. Sludge dry powder production	134
7.4.2. Wall preparation	135
7.4.3. Material mixing	135
7.4.4. Putty application	136
7.4.5. Sanding putty surface	137
7.4.6. Application of top coat	137
7.4.7. Real-scale performance analysis	138
7.5. Summary	141
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS	142
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations	142 142
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research	142 142 149
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES	142 142 149 150
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors	142142149150167
CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage	 142 142 149 150 167 169
 CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage APPENDIX – III: Regression analysis of moisture content 	142 142 149 150 167 169 170
 CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage APPENDIX – III: Regression analysis of moisture content APPENDIX – IV: One-way ANOVA: Bonding strength of putty mixtures with stabilizers 	 142 142 149 150 167 169 170 171
 CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage APPENDIX – III: Regression analysis of moisture content APPENDIX – IV: One-way ANOVA: Bonding strength of putty mixtures with stabilizers APPENDIX – V: One-way ANOVA: SOF of putty mixtures with stabilizers 	142 142 149 150 167 169 170 171 172
 CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage APPENDIX – III: Regression analysis of moisture content APPENDIX – IV: One-way ANOVA: Bonding strength of putty mixtures with stabilizers APPENDIX – V: One-way ANOVA: SOF of putty mixtures with stabilizers APPENDIX – VI: Properties of existing wall putty products 	142 142 149 150 167 169 170 171 172 173
 CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS 8.1. Conclusions and Recommendations 8.2. Future Research REFERENCES APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors APPENDIX – II: Regression analysis of shrinkage APPENDIX – II: Regression analysis of moisture content APPENDIX – IV: One-way ANOVA: Bonding strength of putty mixtures with stabilizers APPENDIX – V: One-way ANOVA: SOF of putty mixtures with stabilizers APPENDIX – VI: Properties of existing wall putty products APPENDIX – VII: One-way ANOVA: Bonding strength of putty mixtures 	 142 142 149 150 167 169 170 171 172 173 174

LIST OF FIGURES

Figure 1.1	Wall finishing system; (a) conventional wall finishes (b) proposed wall finishes	5				
Figure 2.1	Population without access to an improved water source, 2020	10				
Figure 2.2	Estimated quantities of DWTP sludge in different countries and					
U	estimated cost of sludge treatment and disposal [31]	11				
Figure 2.3	Existing water supply schemes of NWSDB Sri Lanka [33]	12				
Figure 2.4	Historical earthen plasters in the world; (a)Bhaja caves – India;					
-	(b)Tomb paintings – Egypt; (c)Minoan wall paintings; (d)Mud					
	plasters – Mali; (e)Tung-oil-lime putty – China; (f)Blood lime					
	plaster - Mexico	26				
Figure 2.5	Soil-based plasters in Sigiriya Sri Lanka; (a) Mirror wall (b)					
	Sigiriya fresco	27				
Figure 2.6	Stupa – earth-based historical constructions in Sri Lanka;					
	(a)Thuparamaya; (b)Jethawanaramaya; (c) Abhayagiriya	27				
Figure 2.7	Observational shrinkage test for clay plaster	33				
Figure 2.8	Possible crack patterns in bonding strength test and analysis of					
	nature of the bond	33				
Figure 2.9	Possible observations in the ball drop test	33				
Figure 3.1	Locations of selected DWTPs; (a)Biyagama DWTP					
	(b)Ambathale DWTP (c)Kandana DWTP	43				
Figure 3.2	Collected sludge samples; (a)Raw sludge from Ambathale DWTP					
	(b)Sludge from sludge drying beds in Kandana DWTP					
	(c)Dewatered sludge from Biyagama DWTP	45				
Figure 3.3	Moisture and Solid content measuring test	47				
Figure 3.4	Particle size distribution of sludge samples; (a)Ambathale DWTP	10				
F' 4.1	sludge (b)Biyagama DW IP sludge (c)Kandana DW IP sludge	49				
Figure 4.1	XRF analysis results	52				
Figure 4.2	(a) VDD test report	51				
Eiguna 12	(C)AKD lest report	54				
Figure 4.5	Heavy metal concentration of studge and maximum permissible	57				
Figure 1 1	IIIIIIIS Maasuring shrinkaga [112], (a)Volumatria shrinkaga;	57				
Figure 4.4	(b)Observational shrinkage	58				
Figure 4.5	Sand cement plaster panel casting for observational shrinkage test	50				
Figure 4.6	Observational shrinkage test results	60				
Figure 4.0	Observational shrinkage with different application areas	60				
Figure 4.8	Sludge samples collected form DWTP	62				
Figure 4.9	Casting cubes for volumetric shrinkage test	63				
Figure 4.10	Volumetric shrinkage test: (a) Initial volume: (b)Volume after	05				
inguie into	completely dry	63				
Figure 4.11	Volumetric shrinkage test results	64				
Figure 4.12	Scatterplot of sludge shrinkage Vs humidity	65				
Figure 4.13	Scatterplots of sludge shrinkage Vs chemicals and other climatic	50				
0	factors	66				
Figure 4.14	Moisture content test results	67				

Figure 4.15	Scatterplots of sludge moisture content Vs chemicals and climatic	
	factors	68
Figure 5.1	Sieved sand for the test	72
Figure 5.2	Mixing with mechanical mixer and cube casting for volumetric	
	shrinkage test	72
Figure 5.3	Volumetric shrinkage variation with sand	73
Figure 5.4	Ball drop test results demonstration [113]	75
Figure 5.5	Ball drop test results	75
Figure 5.6	Natural fibers selected for the study	76
Figure 5.7	Cut natural fiber samples; (a)Laboratory cutting mill SM 200;	
	(b)Cut fiber samples with 10mm	77
Figure 5.8	Volumetric shrinkage variation with natural fiber type	79
Figure 5.9	Binders selected in the study	80
Figure 5.10	Moisture content variation of wet sludge mixtures with binders	81
Figure 5.11	Observational shrinkage of wet sludge mixtures with binders	82
Figure 5.12	Sample preparation and fungus growth test	83
Figure 5.13	Fungus growth test results	84
Figure 5.14	Mixtures prepared in a mechanical mixer	85
Figure 5.15	Volumetric shrinkage variation of wet alum sludge with additives	86
Figure 5.16	Observational shrinkage variation of wet alum sludge with	
-	additives	87
Figure 5.17	Wet sludge mix with optimum putty properties	88
Figure 6.1	Wet and dry sludge	91
Figure 6.2	Thermal alterations of soil at different temperatures [169]	92
Figure 6.3	Heating sludge for determination of thermal alterations	93
Figure 6.4	Color variation of sludge with heat	93
Figure 6.5	Powdered dried sludge preparation for density test	94
Figure 6.6	Density measuring test of dried sludge	94
Figure 6.7	Density variation of dried sludge with heating temperature	95
Figure 6.8	Coagulation process	96
Figure 6.9	Coconut oil as the lubricant	97
Figure 6.10	The production process of dry powdered sludge with coconut oil	98
Figure 6.11	Comparison of burned sludge samples with and without coconut	
	oil	99
Figure 6.12	Density variation of sludge burned with coconut oil	101
Figure 6.13	Dry powdered sludge production variation with coconut oil	
	percentage	102
Figure 6.14	Burned sludge samples with coconut oil	103
Figure 6.15	Sludge dry powder production process	103
Figure 6.16	Dry powder mix preparation for observational shrinkage test	104
Figure 6.17	Results of observational shrinkage test	105
Figure 6.18	Selected stabilizers	106
Figure 6.19	Stabilization of dry powdered sludge	106
Figure 6.20	Observational shrinkage with stabilizers	107
Figure 7.1	Stabilizers selected for the experiments	112
Figure 7.2	Sample preparation for bonding strength test of sludge putty	
	mixtures with stabilizers	113

Figure 7.3	Bonding strength test of sludge putty mixtures with stabilizers	114			
Figure 7.4	Bonding strength variation of putty mixtures with stabilizers 11				
Figure 7.5	Crack patterns variation of putty mixtures with stabilizers	116			
Figure 7.6	Sample preparation for fungus test of sludge putty mixtures with				
-	stabilizers	118			
Figure 7.7	Fungus culture preparation	119			
Figure 7.8	Fungus test chamber preparation	119			
Figure 7.9	Fungus growth test of sludge putty mixtures with stabilizers	120			
Figure 7.10	Fungus growth comparison of sludge putty mixtures with stabilizers	120			
Figure 7.11	Sample preparation for spray erosion test of sludge putty mixtures with stabilizers	121			
Figure 7.12	Spray erosion test of sludge putty mixtures with stabilizers	122			
Figure 7.13	Spray erosion test results of sludge putty mixtures with stabilizers	123			
Figure 7.14	Hardware stores selected to purchase commercially available wall putty products	125			
Figure 7 15	Wall finishes selected for the comparison	125			
Figure 7.16	Sample preparation for bonding strength comparison of different	120			
11guie 7.10	wall finishes	127			
Figure 7.17	Bonding strength test conducted for comparison of different wall				
8	finishes	128			
Figure 7.18	Bonding strength variation of different wall finishes	129			
Figure 7.19	Sample preparation for comparison of fungus growth of different				
C	wall finishes	130			
Figure 7.20	Surface appearance of different wall finishes before and after the				
C	fungus test	130			
Figure 7.21	Sample preparation and spray erosion test for comparison of				
	different wall finishes	131			
Figure 7.22	SOF variation of different wall finishes	132			
Figure 7.23	Real-scale applications of the sludge putty	133			
Figure 7.24	Sludge putty application procedure	134			
Figure 7.25	Sludge dry powder production for real-scale application	134			
Figure 7.26	Wall preparation for putty application	135			
Figure 7.27	Mixing the sludge putty mixture	136			
Figure 7.28	Sludge putty application on wall plaster	136			
Figure 7.29	Putty application tools	136			
Figure 7.30	Sanding the putty surface with sandpapers	137			
Figure 7.31	Applying sealer on sludge putty surface as a top coat	138			
Figure 7.32	Selected sludge putty applied walls to analyze performances; (a)				
	Interior wall; (b)exterior wall; (c)exterior wall with shade	139			
Figure 7.33	Color variation of the sludge putty applied to an interior wall	139			
Figure 7.34	Color variation of sludge putty applied on exterior wall	140			
Figure 7.35	Color variation of sludge putty applied on exterior wall with				
	shade	140			
Figure 7.36	Observations of real scale performances of the developed wall				
	putty	140			

LIST OF TABLES

Table 2.1	Properties of DWTP sludge types	14
Table 2.2	Increasing waste sludge generation in other countries	15
Table 2.3	Experiments conducted to incorporate DWTP sludge in material productions	17
Table 2.4	Materials and tests conducted in research studies to develop soil- based wall finishes	32
Table 2.5	Test standards used in previous research studies	34
Table 3.1	Weight basis Moisture and Solid content of sludge samples	47
Table 3.2	Particle size distribution of DWTP sludge	49
Table 4.1	Chemical elements of DWTP Sludge	53
Table 4.2	Chemical compounds of DWTP sludge	54
Table 4.3	Contaminated DWTP sludge with heavy metals and identified	
	solutions in the literature	56
Table 4.4	Heavy metal concentration of DWTP sludge	57
Table 4.5	Volumetric shrinkage of DWTP waste alum sludge	64
Table 5.1	Additives in historical soil-based plasters	71
Table 5.2	Mixing ratios of sand and sludge	73
Table 5.3	Natural fiber inventory	76
Table 5.4	Mixing ratios of natural fibers and sludge	78
Table 5.5	Mix design for wet sludge mixtures with natural binders	81
Table 5.6	Fungus growth rating system – ASTM D 5590-00 [115]	84
Table 5.7	Additives and binder selected to improve optimum wet sludge	
	mix	85
Table 5.8	Mix Design	86
Table 7.1	Mix design for sludge putty mixtures with stabilizers	112
Table 7.2	Analyzing the nature of bonds between putty mix and plaster	
	according to the crack pattern [116]	114
Table 7.3	Analysis of the bond of putty mixtures variation with stabilizers	117
Table 7.4	Properties of selected commercial putty products	126
Table 7.5	Mix design for preparation of putty mixtures	127
Table 7.6	Product details developed with skilled labor feedback and	
	observations	138

LIST OF APPENDICES

Appendix	Description	Page
APPENDIX – I	Recorded data of physical properties, chemicals and climatic factors	167
APPENDIX – II	Regression analysis of shrinkage	169
APPENDIX – III	Regression analysis of moisture content	170
APPENDIX – IV	One-way ANOVA: Bonding strength of putty mixtures with stabilizers	171
APPENDIX – V	One-way ANOVA: SOF of putty mixtures with stabilizers	172
APPENDIX – VI	Properties of existing wall putty products	173
APPENDIX – VII	One-way ANOVA: Bonding strength of putty mixtures	174
APPENDIX – VIII	One-way ANOVA: SOF of putty mixtures	175

LIST OF ABBREVIATIONS

Abbreviation	Description
DWTP	Drinking Water Treatment Plant
NWSDB	National Water Supply and Drainage Board
SOF	Scaled Off Factor
XRD	X-Ray Diffraction
XRF	X-Ray Fluorescence

CHAPTER 01: INTRODUCTION

1.1. General

Water is one of the main requirements of humans and other living creatures. Life cannot be sustained without water, Therefore, ensuring the access of citizens to consume sufficient safe water is one of the responsibilities of a country. A sufficient amount of safe water consumption is essential for public health. A higher portion of the total population of most of the countries in the world has no access to a sufficient amount of safe drinking water; due to the lack of safe natural water resources. Therefore, they provide safe water to the residents after purifying surface water through DWTPs. Most of the residents in Sri Lanka have access to natural water resources; compared to the other countries in the region [1]. The National Water Supply and Drainage Board (NWSDB) is the responsible authority for drinking water purification and distribution. NWSDB has consisted of DWTPs located all over the country and the total drinking water production is estimated at 30 million m³ per month [1].

DWTP process consists of the steps of coagulation, flocculation, sedimentation, filtration and disinfection [2]. Chemical coagulants are added in the coagulation process to remove suspended soil particles; such as sand, silt, clay and humus, and other contaminants. Alum (Al₂SO₄) is the coagulant used in DWTPs in Sri Lanka. Lime (Ca(OH)₂) and chemical polymers are added to the treatment process. Alum sludge is settled in the flocculation and sedimentation processes; it is removed by filtration. Chlorine (Cl) is added to disinfect the treated water in the disinfection process. DWTP is made with organic and suspended matter, coagulant products, microorganisms and chemical elements. Sludge is an unavoidable waste product of DWTPs [3]. The total solid sludge generation of DWTPs in Sri Lanka is estimated at 1,250m³ per month. Properties of the sludge depend on raw water quality, suspended solids, coagulant type, chemicals added in the process, geographical location and climatic conditions. Direct discharge of DWTP sludge into surface water that affects the quality of the water and aquatic ecosystems. Direct discharge of DWTP sludge is

prohibited by Sri Lankan legislation; as DWTP sludge belongs to the non-hazardous industrial waste category [4],[5]. In Sri Lanka sludge is dewatered and sent for land fillings and to fill rock blasting wells. Therefore, DWTPs practice different methods to dewater the sludge; such as sludge lagoons, sludge treatment plants, sedimentation tanks and sludge drying beds.

There is a sludge treatment plant located in Biyagama DWTP. The sludge treatment plant is consisting of two wash water recovery tanks, a sludge balancing tank, a sludge thickener tank, a thickener sludge tank and two sludge decanters. Embedded water in the DWTP sludge is recovered in the sludge treatment process and fed back to the DWTP and the dewatered solid sludge is removed from the plant. The daily dewatered solid sludge production has been quantified as 10m³ [6]. The dewatered solid sludge is disposed at landfills and rock blasting wells, but the increase in waste sludge generation now becoming a problem for the DWTP. Incorporation of the DWTP sludge in a material production process or utilization in material production processes will be a workable solution for the waste sludge. DWTP sludge is a fine material that is made with mud. A soil-based wall putty can be developed using water treatment plant sludge as a sustainable solution for the adverse environmental impacts resulting from the direct disposal of DWTP sludge.

Shelter is one of the main requirements of human which provide protection. Present building constructions account for around 40% of the global energy consumption. And also, contributes around 30% of the global CO₂ emissions which results in adverse environmental impacts such as climate change and global warming [7],[8]. Therefore, the present building constructions are focused on sustainable and green concepts; such as replacing conventional construction materials with sustainable materials [9],[10]. Therefore soil based walling materials are becoming the new trend in the present construction industry [11],[12]. Soil-based walling materials were introduced as green materials and acted as a moisture buffer and a thermal insulator to create thermal comfort. And also, soil-based walling materials balance the relative humidity and indoor air quality of the building [13],[14]. Walls constructed with soil-based walling materials required protective coating for both exterior and interior surfaces. The

present cement-based wall finishers are inadequate for soil-based walls, as they do not support the exchange of water vapor in the soil-based wall [15]. As the wall finisher does not compatible with the walling material, the finisher detached from the material surface. Cement-based finishers are too stiff and do not adapt to the natural movements of the soil-based wall movements. Therefore, cracks formed on the wall surface allowing moisture to enter and be trapped inside due to the low porosity. A soil-based wall finisher should apply to the soil-based wall to gain the full performance of the soil-based walling materials [16],[17]. Wall putty is a common wall finisher used in the present building sector; therefore, soil-based wall putty developed using a waste product will be a sustainable and a green solution.

Wall putty is a wall plastering material used as a wall finisher and filler. Putty is a powder made with calcium carbonate (whiting) and acid-refined linseed oil. Calcium carbonate is a white mineral (calcite) that is naturally included in chalk, limestone and marbles [18]. Extraction of calcium carbonate contributes to natural resource depletion while degrading the environment. Wall putty production contributes to the environmental impacts directly through volatile organic compounds emission and indirectly through cement production and calcium carbonate extraction [19]. A soil-based wall putty developed with a waste material can eliminate the environmental impacts of raw material extraction for the existing wall putty production process. Therefore, this research was conducted to identify the properties of DWTPs' sludge and to develop wall putty using waste alum sludge while providing a sustainable material to the civil construction industry.

1.2. Research Gap

Direct discharge of waste alum sludge generated in DWTPs to the environment is prohibited by Sri Lankan legislation due to adverse environmental impacts. DWTP sludge generation is increasing with the increasing demand for treated drinking water. And there is no sustainable solution was found for the DWTP waste alum sludge. DWTP sludge is a fine material that is made with mud and there is a potential to develop soil-based wall putty using the sludge. As far as science is concerned, it has been identified that climate change and global warming are the issues that have to be addressed in near future; where the building construction sector provides a considerably higher contribution. Existing wall finishing materials and techniques result in environmental impacts. Calcium carbonate extraction results in environmental degradation and natural resources depletion; while putty production contributes to the environmental impacts directly through volatile organic compounds emission and indirectly through cement production and calcium carbonate extraction. Sustainable wall putty is required to avoid the negative impacts of the existing wall putty.

Soil-based walling materials were invented for the construction industry as green materials. But conventional wall finishing materials and techniques are not compatible with those materials. With the application of conventional wall finishers to earth-based walling materials performance will be reduced. Therefore, a soil-based wall finisher is required.

This research seeks to address all those issues mentioned by adding value to the waste alum sludge generated in DWTPs. The proposed wall putty in this research is a sustainable innovation that utilizes waste sludge as the raw material and contributes to sustaining the building construction sector.

1.3. Aim and Objectives

This research was conducted with the main aim of developing wall putty; applicable as a wall finisher replacing both primer and top coat of conventional wall finishes (Figure 1.1), for tropics using DWTP waste alum sludge and with a few objectives including,

- 1. Analyzing the properties of DWTP waste alum sludge,
- 2. Analyzing and optimizing putty properties of wet alum sludge,
- 3. Analyzing possible means to produce dry powdered alum sludge,
- 4. Analyzing and optimizing putty properties of developed dry powdered alum sludge,

5. Analyzing characteristics of the developed wall putty compared to existing wall finishes, and



6. Analyzing the real-world performance of the developed wall putty.

Figure 1.1: Wall finishing system; (a) conventional wall finishes (b) proposed wall finishes

1.4. Methodology

A literature review was conducted as a background study and a methodology was developed for this research study as mentioned below;

1. Collect sludge samples from different DWTPs and conduct a series of experiments to analyze the properties of waste alum sludge.

- 2. Analyze the putty properties of wet alum sludge and optimize properties by mixing with different additives and binders to develop wall putty using wet alum sludge.
- 3. Evaluate thermal alterations of waste alum sludge under different temperatures to analyze the thermal properties.
- 4. Develop a process to produce dry powdered alum sludge with putty properties considering the thermal properties of the sludge.
- 5. Select a suitable stabilizer for the developed wall putty considering strength and durability.
- 6. Compare intrinsic properties of the developed wall putty with the existing wall finishes
- 7. Conduct real scale performances analysis of the developed wall putty to identify the feasibility

1.5. Main Findings

This research was conducted to develop a wall putty for the tropics using DWTP waste alum sludge and the main findings can be concluded as,

- Dewatered waste alum sludge collected from Biyagama DWTP, has the lowest moisture content and highest solid content compared to raw sludge sample collected from Ambathale DWTP and sludge sample collected from sludge drying beds in Kandana DWTP.
- 2. Cadmium (Cd) (<0.1ppm), Lead (Pb) (4.3ppm), Mercury (Hg) (<0.1ppm) and Arsenic (As) (0.9ppm) are identified heavy metals includes in DWTP waste alum sludge and available in lower quantities than the maximum permissible level defined for top coatings.
- 3. DWTP waste alum sludge moisture content varies between the values of 62.31% and 84.72% and volumetric shrinkage varies between the values of 59.82% and

93.62%; humidity is the significant climatic factor effect on shrinkage of the sludge.

- 4. Physical additives reduce shrinkage of the DWTP waste alum sludge; the optimum sand mixing ratio was identified as 30-60% to obtain optimum wet sludge mix to develop a wall putty mix and banana fiber is the most effective natural fiber type to reduce shrinkage compared to coir and sugarcane fiber.
- 5. Binders improve the adhesiveness of DWTP waste alum sludge but increase the moisture content of the mix. Drying crack patterns of wet sludge mixture vary with the binder type. Natural binders facilitate surface growth and reduce the durability of the mix.
- 6. The density of burned alum sludge varies with heat and the lowest density is recorded at 500°C temperature.
- 7. The density of burned alum sludge with coconut oil is lower than the density of burned sludge at each temperature. The lowest density of burned sludge with coconut oil was recorded at 700°C temperature. Coconut oil is an effective lubricant to use to produce dry powdered alum sludge overcoming alum activation.
- DWTP wet alum sludge mixed with 8% coconut oil by weight and burning at 700°C temperature are the optimum conditions for the dry powdered sludge production process.
- 9. Styrene acrylic binder is the most workable stabilizer for the developed sludge putty compared to cement and lime.
- 10. Sludge dry powder with 300µm fineness is applicable as a wall putty by mixing dry powder 3: binder 1 on a plastered wall surface. And a sealer is applied on the putty surface as a protective coating. The sludge putty can be recommended for both interior walls and exterior walls.

1.6. Organization of the Thesis

This thesis includes eight chapters and the breakdown of the chapters is as follows;

- Chapter 01 discussed general information about the research background, research gap, objectives and methodology. The main findings of this research are also discussed in this chapter.
- Chapter 02 presents a detailed review of the literature conducted to find out the background of the research identifying possible applications of DWTP waste sludge and conditions of existing wall putty technologies.
- Chapter 03 presents the properties of sludge collected from different DWTPs. Findings of the series of experiments conducted with sludge samples and the selection of DWTP to collect sludge for the study are discussed in this chapter.
- Chapter 04 presents the putty properties of DWTP waste alum sludge. Findings of the series of experiments conducted with DWTP waste alum sludge samples are discussed in this chapter; including physical and chemical properties.
- Chapter 05 presents the effectiveness of different admixtures to optimize the putty properties of DWTP wet alum sludge considering the strength and adhesiveness of the mixture.
- Chapter 06 presents the thermal properties of DWTP waste alum sludge. And also, the developed process to produce dry powdered alum sludge considering thermal alterations of the sludge and selection of stabilizer are discussed in this chapter.
- Chapter 07 presents the real-scale behavior of the developed wall putty and properties in comparison with the existing wall finishes.
- Chapter 08 of this thesis presents conclusions, recommendations and future research related to this research.

CHAPTER 02: LITERATURE REVIEW

2.1. General

This research was conducted with the main aim of the development of wall putty for the tropics using DWTP waste alum sludge. As the first step, a detailed review of the literature was conducted to find out the background of the research identifying possible applications of DWTP waste sludge and conditions of existing wall putty technologies. The findings of the literature review were used to plan and advance this research study. Therefore, the literature review was conducted under the main topics as follows;

- Drinking water treatment
- DWTP waste sludge
- Sustainable building construction and materials
- Wall finishes and technologies
- Durability of wall finishes

2.2. Drinking Water Treatment

Water is one of the main requirements of humans and all living creatures. A human can survive only for three days without drinking water; and the average water requirement of a human depends on biological factors (age, gender and weight) and climatic factors [20]. Average per capita water consumption is high in hot climatic zones compared to temperate climatic zone. The average drinking water requirement of Sri Lanka is high due to tropical climatic conditions. Daily consumption of safe drinking water in sufficient amounts, is essential for better human physiology and psychology as water is important for metabolic activities [21]. Therefore, ensuring the access of citizens to consume sufficient safe water is one of the responsibilities of a country; as a sufficient amount of safe water consumption is essential for public health [22],[23]. A higher portion of the total population of most of the countries in the world has no access to a sufficient amount of safe drinking water (Figure 2.1); due to the lack of safe natural water resources. Therefore, they provide safe water to the residents after purifying surface water through DWTPs. Most of the residents in Sri Lanka have

access to natural water resources more than the population of other countries in the region (Figure 2.1) [1].



Source: WHO/UNICEF Joint Monitoring Programme (JMP) for Water Supply and Sanitation

Figure 2.1: Population without access to an improved water source, 2020

2.2.1. Drinking water treatment process

There are different water treatment processes and technologies available in the world. Aquatic systems, mechanical systems and terrestrial systems are the available conventional and non-conventional wastewater treatment technologies. Most drinking water treatment plants purify surface water up to drinking water quality and there are two common treatment processes; conventional treatment technologies and advanced treatment technologies [24]. Conventional treatment technologies include processes of clarification, filtration, coagulation, flocculation, adsorption and disinfection. Advanced treatment processes consists with ultrafiltration technologies [25],[26]. Surface water dissolved with soil particles; sand, silt, clay and humus, and other contaminants deposited from surface runoff and effluent dispose in urban areas and industries. Surface water quality is considered when selecting a water treatment process [2]. Sludge is an unavoidable waste product of DWTPs and the sludge generating rate

of the water treatment plant was estimated as 1%-3% of purified total raw water amount [27].

DWTP sludge is an unavoidable waste product, generated in water treatment and categorized under industrial waste with no harm [27]. But the composition of the DWTS differed with the geographical location, climatic conditions and raw water quality. Therefore DWTP sludge generated in DWTPs located in different countries contain heavy metals and other contaminants exceeding maximum permissible limits; which has the potential to create adverse impacts on eco-systems and human health [28],[29],[30]. Direct discharge of DWTP sludge into water bodies and land is prohibited by the legislation of most countries considering the potential adverse impacts. Therefore, the DWTP sludge should be treated before disposal. With the increasing demand for drinking water, the estimated sludge generation is increasing and the cost for sludge treatment and disposal increasing as shown in Figure 2.2 [31],[32].



Figure 2.2: Estimated quantities of DWTP sludge in different countries and estimated cost of sludge treatment and disposal [31]

2.2.2. Drinking water treatment in Sri Lanka

Sri Lanka has so many natural resources; natural water resources are one of the valuable resources. And there was access to safe water for a higher portion of the Sri Lankan population. Urbanization and industrialization restricted the accessibility of the urban population to safe water compared to rural areas. Therefore, the government of Sri Lanka has the responsibility to distribute safe drinking water in sufficient quantities to urban areas; and NWSDB is the responsible authority for water purification and distribution [1]. NWSDB operates with 323 major, medium and small-scale water supply schemes, located all over the country as shown in Figure 2.3 [33]. Raw water from surface water bodies was collected in DWTPs and treated to drinking quality before distribution. Because surface water is dissolved with impurities such as sand, silt, clay, humic particles and other contaminants; as a result of surface runoff and waste disposal from urban areas and industries [2].



Figure 2.3: Existing water supply schemes of NWSDB Sri Lanka [33]

DWTPs in Sri Lanka practice conventional water treatment technologies as the surface water contamination is at a lower level in the country than in other countries in the world. Therefore the water purification process in DWTPs in Sri Lanka includes coagulation, flocculation, sedimentation, filtration and disinfection processes [2]. Chemical coagulants are added in the coagulation process to remove suspended impurities such as sand, silt, clay and humic particles. Lime (Ca(OH)₂) and chemical polymers are added to the treatment process. Alum sludge is settled in the flocculation and sedimentation processes; it is removed in by filtration. Chlorine (Cl) is added to disinfect the treated water in the disinfection process. Alum-based waste sludge is generated in DWTPs in Sri Lanka and removed after dewatering [3]. There are different technologies used for sludge dewatering, such as sludge treatment plants, sludge drying beds, sludge lagoons and constructed wetlands.

Biyagama DWTP is located at Kelani River right bank. The plant is designed with a conventional physic-chemical water treatment process to consist of two intake structures, a raw water regulation tank, a mixing chamber, six flocculation/clarifiers, eight filters and a clear water/contact tank [34]. Raw water is collected from the Kelaniya River; and daily release 180,000m³ of treated drinking water from the plant. Biyagama DWTP fulfil drinking water requirement of one million of population at divisional secretariats of Biyagama, Kelaniya, Mahara, Wattala, Ja-ela and Dompe. There is a sludge treatment plant located in Biyagama DWTP. The sludge treatment plant is consisting of two wash water recovery tanks, a sludge balancing tank, a sludge thickener tank, a thickener sludge tank and two sludge decanters. Embedded water in the DWTP sludge is recovered in the sludge treatment plant. The daily dewatered solid sludge production has been quantified as 10m³ [6].

2.3. DWTP Waste Sludge

There are different types of sludge with specific properties. Properties of the sludge differ on different factors; such as the geographical location of the plant, climatic conditions, raw water quality, treated water quality and water treatment process [29],[31]. The water treatment process includes physical processes and chemical treatments. In the treatment process coagulants, polymers, lime, chlorine and other chemicals were added. Coagulants are added to remove suspended particles in raw water which do not settle under gravity due to the light weight. Aluminium salts (Al₂(SO₄)₃.18H₂O), ferric ion salts (FeCl₃.6H₂O), and ferrous iron salts (FeCl₂, FeSO₄.7H₂O) are the coagulants used in DWTPs [35],[36]. DWTP sludge removed from each plant consists of coagulant products added in the treatment process. Concentrations of the coagulants and other chemicals added in the treatment plant depend on the raw water quality and the treated water quality. Therefore, the sludge from different DWTPs consists of different quantities of coagulant products and other chemicals. Three types of DWTP sludge can be identified according to the coagulant type and the chemical composition; 'Alum sludge', 'Ferric sludge' and 'Lime sludge' [31]. Each type of DWTP sludge has unique properties due to the behavior of different chemicals and coagulant products (Table 2.1).

Sludge Type	Moisture content (%)	Density (gcm ⁻³)	Soil classification (USCS)	Loss on ignition (%)	References
Alum sludge	80-550	1.86-2.33	High-plasticity clay	15-57	[21] [27]
Ferric sludge	108	2.26-2.72	High-plasticity clay	24-50	- [31],[37], [39] [30]
Lime sludge	38-72	2.57-2.62	Low-plasticity silt	35-40	- [30],[39]

Table 2.1: Properties of DWTP sludge types

2.3.1. DWTP sludge disposal methods

All the DWTPs in the world released the waste sludge to nearby water bodies and open lands at the beginning. After that DWTP sludge is categorized as a non-hazardous industrial waste [32],[40]. However, it is non-hazardous, disposal of water bodies and open land is prohibited by legislation. Dispose of sludge creates potential disturbances to the natural water reservoirs and aquatic ecosystems due to the unwanted mud deposits [4]. The increasing generation of sludge is becoming a huge problem for most countries; as shown in Table 2.2, due to the high cost engage in sludge treatment, transportation and disposal processes. Dewatered DWTP solid sludge dispose of at landfill sites after treatments by many countries; but some developing countries still release it to water bodies and nearby drains due to the high cost of sludge treatment and disposal practices [2],[41]. According to the literature very few DWTPs reuse or recycle their waste sludge.

Source	DWTP	Country	Sludge Generation	Disposal Method
[2]	Ghaziabad	India	29,700 tons/yr.	Discharged to a nearby drain
[30]	Santa Maria	Brazil	2,100m ³ /day	Discharged to rivers and sea
[42]	From all DWTPs of the country	South Korea	1,159.1 tons/day (2016) 982.7 tons/day (2011)	Disposed in landfills
[43][44]	Metropolitan Waterworks Authority	Thailand	300 tons/day in the dry season 700 tons/day in the wet season	Disposed in landfills
[45]	From all DWTPs of the country	Italy	Dewatered sludge 750,000 tons/yr.	Disposed in landfills
[46]	Fong-Yuan DWTP	Taiwan	6,000 tons/yr.	Disposed in sanitary landfills
[47]	From 29 DWTPs	Taiwan	160,000 tons/yr.	Disposed in landfills
[48]	From 5 DWTPs	Taipei - Taiwan	70,200 tons/yr.	Disposed in landfills
	From 384 DWTPs	France	63,800 tons/yr.	Disposed in landfills
[49]	From 307 DWTPs	Mexico	2.8x10 ⁹ m ³ /yr.	Disposed in landfills
	From 215 DWTPs	Spain	120,000 tons/yr.	Disposed in landfills
[50]	Sent Joan Despí DWTP, Barcelona	Spain	17,000 m ³ /day	Mixed with raw materials in cement production after spray drying.

 Table 2.2: Increasing waste sludge generation in other countries

Dewatered sludge generated in Biyagama DWTP is also disposed at landfills and rock blasting wells, but the increase in waste sludge generation now becoming a problem for the DWTP. Incorporation of the DWTP sludge in a material production process or utilization as a raw material in a material production process will be a sustainable solution for the waste sludge; to eliminate the results of the environmental impact when disposing of DWTP sludge [51].

2.3.2. DWTP sludge applications

Direct discharge of water bodies, open lands and nearby drains was the common DWTP sludge disposal method practiced in the world until the direct discharge of sludge was prohibited by legislation. Therefore, all the countries started looking for a sustainable solution to discharge or recycle DWTP sludge. Some research studies were conducted to improve the efficiency of DWTP processes to reduce the sludge production rate of the plant. But the sludge production was unable to eliminate; as it is an unavoidable waste product of the DWTP process. Therefore, treated DWTP sludge was disposed in landfills. But the high sludge treatment cost and transport cost to dispose in landfills led to fond processes to recycle DWTP sludge.

DWTP sludge is a fine material similar to mud and clay in texture. The sludge is made with coagulant products and chemical elements, organic and suspended matter and microorganisms [13],[15]. Mud and clay are versatile materials used in the construction industry from prehistoric times [52],[53]. Therefore, most of the research was conducted to identify the possible uses of DWTP sludge as an alternative to mud and clay. And also, research studies were conducted to analyze the possibility of DWTP sludge to incorporate into construction material production processes without affecting the properties and performances of the products (Table 2.3).

Source	Year	Country	Product	Findings
[54]	2017	Brazil	Concrete	The addition of DWTP sludge in wet form reduces the strength of concrete. The ideal incorporation level was identified as up to 5%; which effective limit for non- structural concrete.
[44]	2015	Thailand	Brick	DWTP sludge–fly ash geo-polymer bricks have a longer service life compared to clay–cement bricks, which are used commonly in wall constructions in many countries.
[55]	2016	India	Brick	Bricks can be produced by incorporating the DWTP sludge with clay materials. Ferric sludge has better mechanical properties than alum sludge.
[32]	2005	Taiwan	Brick	Adding DWTP sludge increase the sintering temperature requirement. Under the common brick-making condition, DWTP sludge can be incorporated up to 15% with clay to produce first-grade bricks.
[56]	2017	Egypt	Refractory brick	DWTP sludge can incorporate successfully with rice husk ash to produce high-quality insulating refractory bricks for various thermal insulating applications.
[57]	2015	Brazil	Ceramic brick	Ceramic bricks can be produced by incorporating DWTP sludge up to 13.9%.
[58]	2017	Morocco	Portland cement	Incorporation of DWTP sludge with clay in the synthesis of Portland clinker complementarily can reduce the energy demand of Portland cement production.
[59]	2011	Jordan	Tile	Tiles produced by mixing DWTP sludge replacing 50% of cement, comply with the breaking strength of 2.8MPa required for external use tiles.

Table 2.3: Experiments conducted to incorporate DWTP sludge in material productions

According to the previous research studies the main findings can be concluded as below;

- WTP sludge has some unique characteristics to enable its incorporation in construction materials such as bricks, ceramics, cement and geopolymers.
- The main challenges in the application of WTP sludge in construction materials are derived from its high variation in physicochemical properties and relatively high content of organic matter, which will increase the porosity and water absorption therefore potentially adversely affecting products' structural integrity and performance.
- The majority of studies with ceramic bricks showed that the incorporation of alum sludge up to 10% is satisfactory in maintaining the structure performance. For iron-based sludge, an increase in mechanical strength was observed when more significant proportions were added. A similar effect was observed in studies with lightweight aggregates, where Fe content in the sludge had a high impact on water absorption and compressive strength.
- Regarding the production of cement and concrete with the partial addition of DWTP sludge, it is potentially feasible, especially as a pozzolanic addition and replacement of sand in concrete up to 5%.
- The effect of sintering temperature and the ratio of oxides had a great influence on heavy metals leaching from WTP sludge amended products. Sintering above 1000°C is recommended to ensure the full immobilization of heavy metals such as Cd, Cr, Cu and Pb hence preventing further environmental pollution.
- Used as a substitution, WTP sludge can make saving for raw materials and energy, replace landfill as solid waste management, and contribute to sustainable construction materials production.

2.4. Sustainable Building Construction and Materials

The shelter is a main requirement of humans; from prehistoric times shelters were used as a protective structure to protect from extreme environmental conditions. Extreme conditions can be identified as heavy rain, extreme wind, harmful radiation and protection from animal attacks. In the beginning, primitive structures were used as shelters and with civilization building constructions were developed and improved with the technology development. In prehistoric times shelter was a protective structure but present structured buildings are designed with wide verities of activities including living, entertaining and manufacturing [60]. Different building materials used in building constructions can be identified as walling materials, floor finishes, roofing materials, paving materials, wall finishers, etc [61],[62]. Walls and roofs are important building components in the concern of protection; as they are directly exposed to the external environment and protect building occupants from external heat gain, extreme weather conditions and other external effects [61].

More than 400 million tons of materials are annually consumed by the building construction sector; according to the estimations of the Green Building Council of the United Kingdom. Raw material extraction for construction material productions creates adverse environmental impacts; resulting in natural resource depletion, damaging natural ecosystems and affecting natural cycles. And also, the material extraction processes create adverse environmental issues through fuel consumption, chemicals, dust and heat emissions and noise pollution. The construction industry contributes to the high amount of global waste generation and global resource consumption. Building constructions contribute to the consumption of 70% of electricity, 50% of energy and 15% of natural water; while contributing to emissions of 40% of CO_2 and 30% of waste. And the estimated values depend on the country, economic level and lifestyles of the residents [63],[64].

Environmental issues of building constructions run throughout the whole lifecycle of development. Those impacts occur from the beginning of initial works, during the construction period, in the operational stage and up to the final demolition stage [63]. Land clearing and preparation for building constructions change natural land use

patterns resulting destruction of natural habitats and destroying wildlife. The building construction process contributes to global warming, greenhouse gas emissions, air pollution and water pollution; through the processes of material extraction, manufacturing and transportation [65]. After considering all those adverse issues of conventional building construction practices, it was necessary to resort to a sustainable construction process [66]. At the present, all the development practices are compiled with the Sustainable Development Goals (SDGs) adopted by the United Nation in 2015. Therefore building constructions move from conventional building practices toward sustainable building constructions [67],[68].

2.4.1. Sustainable building constructions

Sustainability can be defined as, the fulfillment of the present requirements without compromising the ability to fulfillment of future requirements. Sustainable design refers to sustainable practices in the construction industry; including building designs and constructions. Buildings designed with sustainable concepts; also define as green buildings, have many advantages in the areas of resource efficiency, environmental impacts, and social and economic impacts. The green building concept has been developed to mitigate adverse impacts of conventional construction practices to meet the design intentions of the present while considering effects in the future. The whole lifecycle of the building is considered in the green building concept. It is a continuous process applicable from the site selection stage, material selection stage; considering all the adverse impacts of each step [69].

There are different green building rating systems in the world to assess sustainable building constructions; such as LEED, USGBC, GBCSL, BREAM, etc. [70]. Those rating systems have been developed by allocating credits for several categories, including sustainable site, sustainable design elements, location and transportation, material and resource, building product disclosure and optimization, energy and atmosphere and indoor environmental quality [71]. In those rating tools, a considerable number of points were allocated under sustainable building construction methods. Building construction methods that reduce resource consumption and improve the

usage of locally available sustainable materials were encouraged in sustainable building design and construction.

There are important benefits of sustainable building constructions over conventional constructions; such as upgrading building occupants' health, indoor comfortability, building life cycle cost reduction, reduction of waste generation, energy and water saving, increasing building lifetime and combination renewable energy sources [72],[69]. Green buildings can reduce solid waste generation by 70% and carbon emissions by 35%; while reducing energy usage by 50% and water usage by 40% compared to conventional buildings [73]. The initial cost of green buildings is much higher as concluded in many research findings. Initial cost engaged in green building constructions and green concepts implementation; however, research studies were clearly showing the operational cost reduction and financial benefits which can be achieved considering the life cycle approach of green buildings compared to conventional buildings [74]. Therefore, research studies are conducting to develop low-cost construction materials for sustainable constructions and to develop green features with high efficiency and low cost; in order to reduce the initial cost of green buildings.

2.4.2. Sustainable construction materials

The rapid population growth of the world increases the demand for buildings; and creates a high demand for construction materials [75]. The building material production processes use natural resources, such as clay, sand, stones and lime in large quantities, in developing countries. There is a high demand for clay bricks and clay tiles; produced with clay, concrete blocks; produced with cement and sand, concrete; prepared by mixing aggregates of stones, and cement; made with lime. A considerable amount of energy is consumed in the mining, transportation and burning processes of these raw materials in construction material production processes. And also the production processes result in adverse environmental impacts; such as deforestation due to wood consumption, saltwater intrusion into rivers and loss of soil fertility [76]. Therefore, it is important to replace conventional raw materials in the construction
material production process with alternatives to reduce the burden on natural resources.

There is a considerable high number of points allocated for construction material selection by most of the building sustainability assessment tools in the world. Resource efficiency is considered in the material and resource category of all assessment tools [10]. In LEEDv4, 13 points were allocated for the material and resource category and which is 11.81% of the total point allocation; and in the GREENSL rating tool, 14 points were allocated and it is 14% of the total point allocation [63]. Resource efficiency in building constructions includes the efficient use of energy, natural resources, and materials; and encourages to use of sustainable materials, reuse of materials and use of alternative materials with low environmental impacts [7],[61]. Building constructions contribute to high consumption of energy and material; 50% of the raw materials and 42% of energy are used for building constructions annually in European Union according to the estimations [7].

Cement is one of the common construction materials used in the construction industry around the world; it is estimated that the annual cement production in the world is over 2 billion tons [77]. It was predicted that concrete use reach four times, the use in the year 1990, By 2050, if not found an alternative mixture [77],[78]. It was estimated that 1 ton of cement production results in 1 ton of CO₂ emission; and cement production is responsible for 5% of the annual global CO₂ emissions [79],[80]. Most developed countries use waste products as alternatives to natural resources use for construction material production and there are sustainable construction materials developed from waste products [66],[76].

2.4.3. Walling materials

Building wall is the most important building component. The cost of wall construction contributes to 35% of the total building construction cost in Sri Lanka; while walling materials account for more than 15% of the total building construction cost [61],[53],[81]. There are two types of walling materials in the world; solid walling materials and lightweight walling materials. Solid walling materials are used in

tropical countries and it is the most common walling material type used in Sri Lankan building constructions [80]. There are different solid walling materials used in Sri Lankan buildings; such as burnt clay bricks, concrete blocks, Cabook blocks, cement stabilized earth blocks (CSEB), mud concrete blocks (MCB) and geo-polymerized earth blocks [61]. Durability and structural performance of walling materials is very important as they are directly exposed to outdoor extrema environmental conditions.

The tropical climate zone included countries located near the equator which consists of tropical climatic conditions. Sri Lanka belongs to the tropical wet climatic zone; with the climatic conditions of high daily temperatures vary in between $20^{\circ}C-30^{\circ}C$, year-round uniform high precipitation, over 2000mm total rainfall and humidity vary in between 60%-95% [82]. Therefore, the walling materials of buildings in the tropical zone are exposed to tropical climatic conditions consisting of high humidity, rainfall and temperature [80]. Durability of walling materials is directly affected by the tropical climate conditions. Continuous exposure of walling materials to tropical climate; which consists of high solar radiation, rain, wind and temperature results in the deterioration of walling materials. Climatic conditions consist of high humidity and moisture, which accelerate the biological growth of walling materials. Surface growths of fungus, algae, lichen and moss can be identified in the tropical climate. Surface growth damages the appearance of the building while reducing the walling materials' strength and durability. And also surface growth adversely affects indoor air quality and occupants' health [83].

At the present, different walling materials are used for construction. Walling materials have different intrinsic properties; and have different surface roughness values [80],[61]. According to previous studies it has been identified as walling materials with low surface roughness values are more durable and resistant to mould and moss growth [62],[84]. Wall finishes are applied to smooth wall surfaces against the rough surface of the wall. Smooth building walls improve indoor air quality; reduce dust level, increase the interior light level and evade fungus and moss growth [85].

2.5. Wall Finishes and Technologies

Walling materials have different surface roughness values. With the increase in surface roughness durability of walling materials is reduced. To overcome these issues and to increase the durability wall paster is applied on walling material [62],[86]. wall plaster provides a smooth and finished surface while providing a protective coating acting as a moisture barrier for walling material. And also wall plaster added a little extra strength support to walling material. Wall plaster increases the appearance of the building while creating occupant comfort in the building [85],[87].

The most common wall plaster type is cement plaster; where the main ingredient is cement. Rough cement plaster is prepared by mixing cement and sand. And also there are gypsum plasters; where gypsum is the main ingredient [88]. In Sri Lankan buildings, cement and rough cement plaster are widely used. There is a high demand for cement and sand in the building construction sector in Sri Lanka. Cement production and application result in some adverse environmental issues; such as environmental pollution, GHG emissions, natural resource depletion and dust emission [89],[90]. Similarly, cement production creates health problems including respiratory problems [91]. The demand for sand also increases as sand is the main ingredient of rough cement plaster. River sand mining creates many adverse environmental issues in Sri Lanka. And also the high price of sand contributes significantly to increasing the building construction cost [92],[62]. Green building concept encourages to reduction of cement and sand usage considering adverse environmental impacts of both materials. And research studies are conducted to find out environmentally friendly alternatives for sand and sustainable processes to reduce the environmental impacts of cement production and usage.

2.5.1. Historical evidence of earthen wall finishes in the world

From prehistoric times wall finishing techniques were practiced in decorative and communication perspectives [93]. The prehistoric man used smooth surfaces for cave paintings which were used for decoration and communication [93]. When considering historical evidence, can be seen that materials and techniques used for wall smoothing

and finishing have changed with time. In prehistoric times man used plasters to make the base for their cave paintings. For those plasters, they used soil as the main ingredient and added animal fat, plant saps, balsams, milk or blood as the binding medium [94]. There is so much evidence of historical earth-based constructions constructed thousands of years back, which can be found all over the world including Sri Lanka.

Bhaja Caves – India: A durable earthen plaster used for exterior and interior plastering is found in Bhaja caves in India and dated back from the 1^{st} century BC to the 3^{rd} century AD (Figure 2.4 (a)). The plaster was made with clay, mixed with raw soil, slaked lime, rice husk and proteinaceous adhesive; to give strength and stop cracking during the drying [95].

Tomb paintings – Egypt and Minoan wall paintings: Paintings were painted on properly prepared plasters in the tomb in Egypt and Minoan walls which dated back to the 4th and 5th century BC (Figure 2.4 (b) & (c)). They used sand and lime plaster with different thicknesses in different places [94].

Mud plasters – Mali: In Mali, wet mud fermented with rice husks was used as wall plaster (Figure 2.4 (d)) and some African countries used proteins to have enzyme reactions, therefore, they mixed milk and sometimes animal blood into the plaster mixture [96].

Tung-oil-lime putty – *China:* Tung-oil–lime putty is a special organic-inorganic binding material that was widely used in ancient China. It is mainly composed of boiled tung oil and lime. To reinforce its cracking resistance and save raw materials, plant fibers were also added. Due to its good water-resistivity, adhesive property, mothproofing and stability, it was used in numerous fields. According to ancient records and archaeological discoveries, the application of putty dated from about AD 581–AD 907 (Figure 2.4 (e)) [97],[98].

Blood lime plaster – Mexico: Blood lime plaster is used in ancient Europe, China and Mexico (Figure 2.4 (f)). These traditional oriental blood lime plasters have high bonding strength, water resistance, weather resistance and smoothness than the other

historical plasters [99]. The blood proteins affect the crystal formation of the plaster and develop a compact layer that improves the surface properties of the plaster [100].



Figure 2.4: Historical earthen plasters in the world; (a)Bhaja caves – India; (b)Tomb paintings – Egypt; (c)Minoan wall paintings; (d)Mud plasters – Mali; (e)Tung-oil-lime putty – China; (f)Blood lime plaster - Mexico

2.5.2. Historical evidence of earthen wall finishes in Sri Lanka

Sri Lanka has a great written history with shreds of evidence compared to other countries in the world. Historical monumental constructions lasting for thousands of years can be found all over the country. Sigiriya is one of the UNESCO world heritages located in Sri Lanka; constructed by King Kashyapa in the 5th century AD (more than 1500 years old). The Sigiriya is popular all over the world because of its mirror wall and the fresco (Figure 2.5). The mirror wall is created with highly polished soil-based plaster on the internal surface of the parapet wall [101]. The world-famous fresco of Sigiriya is painted on a soil-based plaster which is made with few layers. The reddishbrown color ground layer of the plaster (half an inch in thickness) was made with tempered earth and kaolin and coated with at least two layers of white chunam (one-fourth to half an inch thickness). The clay base of the plaster was strengthened by the admixture of paddy husk and shreds of coconut fiber [102].



Figure 2.5: Soil-based plasters in Sigiriya Sri Lanka; (a) Mirror wall (b) Sigiriya fresco

Sri Lanka has a great proven history of earth-based constructions. Stupas are monumental structures built to Honour Lord Buddha; they are architecturally important Buddhist structures. The very first stupa constructed in Sri Lanka is Thuparama (Figure 2.6 (a)); which was built by King Devanampiyatissa (250-210 BC) in the capital city of Anuradhapura. Jetavana is the largest structure made of brick in the world [103] (Figure 2.6 (b)). Stupas are solid structures, built with burnt clay bricks and un-burnt sun-dried clay bricks. The bricks were laid with a thin mortar; made with clay slurry, called 'butter clay', and the outer surface was protected with a thick plaster mixed with plant-based adhesives. Ingredients used to prepare mortar in stupas were identified as finely crushed dolomitic lime, sand and clay; after studying the plaster samples of Abhayagiri (Figure 2.6 (c)) (constructed in the first century BC) and Jetavana (constructed in the third century AD) stupas. This evidence are proving that lime mortar technology had developed throughout the centuries [103].



Figure 2.6: Stupa – earth-based historical constructions in Sri Lanka; (a)Thuparamaya; (b)Jethawanaramaya; (c)Abhayagiriya

2.5.3. Present wall finishes and technologies

Present building constructions use different wall finishing materials; such as cement, wall care putty, lime slurry, etc, and different techniques; such as plastering and cladding are used to smooth and finish wall surfaces [104], [105]. But Cement and sand are the main ingredients of the wall plaster. Cement production contributes to 5% of global annual CO₂ emission [78],[77], and sand mining results in adverse environmental impacts such as environmental degradation. And also sand price is increasing gradually at present while increasing the construction cost of buildings [106],[107]. Cement and sand usage in the building construction industry can be reduced by avoiding cement plasters used as wall finishes. The reduction of cement and sand usage can reduce building construction costs and the global annual CO₂ emission. Wall care putty is a thick mixture of finely powdered calcium carbonate (whiting) and acid-refined linseed oil [18]. Calcium carbonate extraction results in environmental degradation and natural resource depletion. Wall putty production contributes to the environmental impacts directly through volatile organic compounds emission and indirectly through cement production and calcium carbonate extraction [19]. On the other hand, these conventional wall finishing materials and techniques are not compatible with soil-based walling materials.

2.5.4. Wall care putty

With the increase in building construction, the demand for wall finishes is increasing. Wall care putty is the most commonly applied material as the primer layer for both interior and exterior wall surfaces. Wall care putty is applied as a leveling material on the wall plaster and acts as a protective layer for the wall plaster while providing a strong and durable base for the top coat. And also the putty materials are developed with the ability to fill dents, and hair cracks to level the rough surface of the plaster [108]. Wall putty materials are developed with the characteristics of;

- High bonding strength
- Moisture barrier
- High durability
- Low bio-receptivity

There are different types of wall putty products available in the market; such as lime, white cement putty, acrylic putty and Plaster of Paris (POP).

Lime putty: this is a conventional wall putty product applied in traditional constructions. And also, it was the widely applied wall putty in past constructions. Lime putty developed with calcium carbonate extracted from lime depositions; such as limestones, chalk, corals and marbles.

White cement wall putty: is a putty product commercially available in powder form; and it is developed with a mineral, white cement and polymer. The powder is applied after mixing with water. There are different white cement-based wall putty products are available in the market from different commercial brands. Each brand has different properties.

Acrylic wall putty: is a ready-use putty mix, developed with acrylic polymers and other chemicals. POP is a powder material developed with gypsum and the putty mix is prepared by mixing the powder with water [109]. Talcum, quartz, gypsum and cement are the main raw materials for most commercial acrylic wall putty materials and the material should consist of the engineering properties of high bonding strength, crack resistance, resistance to water, low shrinkage and enough flexibility [108].

Most common wall care putty products in the market are white. Wall putty is applied as the primer layer of the wall providing a uniform base for the top coating. Paint is applied as the top coat and to avoid impact on the selected color of the paint wall putty is applied in white color. Therefore, there is high demand for white color wall putty products; but in the present market colored wall putty products are available. Colored putty products are developed by adding color pigments to conventional white commercial wall putty products. Colored wall care putty products are applied as the primer layer with a protective top coat eliminating the application of a paint layer on walls. According to architectural purposes and client requirements, colored wall putty products are applied in buildings. However, the increasing consumption of natural resources has led to a strong depreciation alarming the need for sustainable alternatives for existing wall care putty materials.

2.6. Soil-based Wall Finishes

Soil-based walling materials are becoming the new trend of the present construction industry as an eco-sustainability material; such as mud concrete blocks, cement stabilized earth blocks and rammed earth walls [13], [15], [110]. Soil-based walling materials perform as a moisture buffer; to control the humidity of the indoor environment; a thermal insulator; provide thermal comfort, and improve the indoor air quality of buildings [15],[17]. Walls constructed with soil-based walling materials require protective coating for both exterior and interior surfaces. Cement-based plasters are not compatible with soil-based walling materials. Cement-based plasters are highly rigid materials and do not facilitate moisture transmission and natural wall movements of soil-based building walls. As result, the detachment of cement-based plaster from the building wall reduces the optimum performance of soil-based walling materials [13]. Cement-based pastes are not suitable for repairs of soil-based walls; soil-based walls repaired with cement-based mixtures were cracked easily and allowed trapped rainwater inside the wall. A soil-based wall finisher and a mixture; for repairs of soil-based building walls, are required to obtain optimum performances of soilbased constructions [15],[111].

Mud and clay are versatile natural materials used in the construction industry. There are available walling materials, mortars, plasters, roofing materials and flooring materials made with mud and clay. Mud and clay-based materials are universally accepted as environment-friendly and user-friendly materials for building constructions [16]. There is a very long history of mud and clay-based plasters and renders; historical evidence is available in all regions of the world. Mud and clay-based finishes are optimum and ideal for soil-based building walls, but the most important fact is that soil-based wall finishes are compatible with building walls constructed with any walling material [112].

2.6.1. Research conducted on soil-based wall finishes

With the adaptation to sustainable building construction practices, demand for sustainable materials were increasing at present. Most countries in the world conducting research studies to develop sustainable building construction methods and introduce sustainable construction materials. To that soil-based walling materials and wall construction methods were introduced to the present construction industry. Although soil-based walling materials and construction methods were used from ancient times in most countries; present research studies were conducted to develop the ancient soil-based technologies to make them compatible with present building construction technologies. Research studies were conducted to develop soil-based wall finishes using mud and clay with the aims of;

- To use for repairs of ancient soil-based structures and constructions,
- To develop a wall finishing material compatible with present soil-based construction technologies
- To develop eco-sustainable construction material

Advantages of the soil-based wall finish [113],[114] compared to conventional wall finishes were identified as;

- Soil-based wall finishes are successfully applied to different building wall types; such as, rammed earth walls, adobe walls, concrete walls, wood walls, brick walls and straw bale walls.
- Soil base wall finishes perform as an insulator for heat, moisture, sound, odor and noise.
- Soil-based wall finishes create thermal comfort for occupants against the temperature variation of the outdoor environment while maintaining optimum indoor humidity.
- Soil-based wall finishes contribute to improving indoor air quality compared to conventional wall finishes.

Soil-based wall plasters were developed by mixing different clay types with different admixtures. Research studies were conducted to develop ready-mixed clay plasters by mixing clay sand in different ratios; also clay plasters were developed by mixing different clay types with different binders and additives; as shown in Table 2.4. in the studies mixtures were prepared by mixing using a hand mixer or a mechanical mixer until a proper consistency was observed [113],[114].

Source	Year	Product	Materials	Tests conducted
[113]	2015	Ready-mixed clay plasters	Red clay Yellow clay Sand	 Volumetric shrinkage Observational shrinkage Ball drop test Compressive strength Moisture content XRF analysis XRD analysis
[114]	2014	Ready-mixed clay plaster	Clayey soil Siliceous aggregates (0/4 mm sand) Cement Lime	 Linear shrinkage Compressive strength Flexural strength Adhesive strength Density XRD analysis
[115]	2017	Eco-efficient earth-based plaster	Clayish earth Siliceous sand (0-2mm) Cut out fibers (1-2cm long)	 Particle size distribution Density Biological susceptibility XRD analysis

Table 2.4: Materials and tests conducted in research studies to develop soil-based wall finishes

According to those research studies the main findings can be concluded as below;

- Different clay types have different physical and chemical properties and the effectiveness of clay plaster depends on the clay type.
- Shrinkage and bonding strength are the main properties of the clay that should consider when developing clay plaster or putty.
- Shrinkage of clay plasters can be measured by conducting a volumetric shrinkage test and observational shrinkage test (Figure 2.7). Shrinkage of clay plaster can be reduced by adding fibers.
- Surface crack formation of clay plasters increases, with the increased moisture content of the mix; and the optimum water mixing ratio was identified as 30%-40%.
- Bonding strength decreases, with the increased clay content of the mix while increasing drying cracks according to the observational shrinkage test.
- Bonding strength can be improved by adding binders and the standard bonding strength of wall putty products is ≥0.6Mpa. The nature of the bond can be analyzed

after observing the bond-breaking pattern of the sample according to the ASTM D5868-014 test standards (Figure 2.8) [116].

- The standard density of the soil to develop effective soil-based plasters was identified as 2.6-2.7gcm⁻³ and the standard particle size distribution was identified as 18%-22% of clay, 40%-42% of silt and 30%-40% of sand according to IS:13077-1991 standards.
- Plaster properties of the clay mix can be improved by mixing sand and the optimum clay sand mixing ratio can be identified by conducting a ball drop test (Figure 2.9).



Figure 2.7: Observational shrinkage test for clay plaster



Figure 2.8: Possible crack patterns in bonding strength test and analysis of nature of the bond



(a) High clay content

(b) Poor clay content

(c) Optimum clay content

Figure 2.9: Possible observations in a ball drop test

2.6.2. Test standards for soil-based wall finishes

According to the literature, there is no developed standard for the characterization of soil-based wall finishes [113]. Therefore, standards developed for commercial wall putty products, cement and lime-based plasters, soil-based block productions, and mud mortar, were followed in relevant research studies; as shown in Table 2.5.

Source	Product	Test	Standard	Standard value
[117]	Mud plaster	Particle size distribution	IS:13077- 1991	18%-22% of clay 40%-42% of silt 30%-40% of sand
		Density	EN 1097-3	2.6-2.7gcm ⁻³
		Density	EN 1097-3	2.6-2.7gcm ⁻³
[115]	Eco-efficient earth-based plasters	XRD analysis for mineralogical analysis	DIN 18947,2013	-
		Fungus growth test	ASTM D 5590-00	0%
	Nous outorion	Bond strength	ASTM D5868-014	≥0.6Mpa
[108]	wall putty	Sanding property	JGT-157- 2009 GB/T9278	Easy
		XRF analysis to determine the chemical composition		-
[113]	Ready-mixed clay plaster	XRD analysis for mineralogical analysis	DIN 18947,2013	-
[113]	with clay and sand	Ball drop test	(German Standard)	Optimum clay/sand proportions are semi- shiny and exhibit only a small number of surface cracks, with no shattering
		Volumetric shrinkage		<10%

Table 2.5: Test standards used in previous research studies

2.7. Characteristics of Wall Care Putty Products

Wall care putty is applied as a leveling material on the wall plaster and acts as a protective layer for the wall plaster while providing a strong and durable base for the top coat. And also the putty materials are developed with the ability to fill dents, and hair cracks to level the rough surface of the plaster [108]. Wall putty materials are developed with the characteristics of;

- High durability
- Low bio-receptivity
- High bonding strength
- Moisture barrier

2.7.1. Durability of wall finishes

Building components such as building walls and roofs are constantly being designed and manufactured to satisfy the ever-growing needs of society. Many of these walls and roofs will be subjected to adversative conditions over their life and these conditions will affect their intended performance. Durability measures that building component will perform their functions under the expected conditions. The durability of building components can be defined as, the safe performance of a structure or a portion of a structure for its expected design life, according to the ASTM E241 standard. Therefore selected materials should be durable enough to perform optimum conditions throughout the whole lifecycle of the building [118].

Building walls are directly exposed to the external environment; therefore, the durability of the wall finishers under humid tropical climates should be concerned. Whole life cycle cost includes component replacement interval, maintenance activities and frequency; therefore, life cycle cost is influenced by durability. There are different stress factors effected on the durability of wall finishes; such as mechanical, thermal, environmental and biological factors. High solar radiation, heavy monsoonal rains and high wind conditions can occur in tropical climatic zones; and surface erosion, mass loss, pitting, cracking and surface growth can result when exposed to tropical climatic conditions for a longer time duration [118]. The ability of a wall finisher to resist when exposed to heavy rain conditions should be measured. Exposure to stress factors results

in strength loss of buildings and adversely affect the appearance of the building; therefore, measuring the durability of a wall finisher is important when selecting or developing a wall finisher to apply in the tropical climatic zone [62],[119]. Commercial wall putty products available in the market are subjected to scale-off, fungus growth, color deterioration and water leakages after long-term exposure to tropical climatic conditions [84],[86],[120].

2.7.2. Tropical climate and rain surface erosion

There are different climatic zones in the world; the tropical climatic zone is one of the major climatic zones with a wide spread. The tropical climatic zone consists of countries located near the equator. Sri Lanka belongs to the tropical wet climatic zone and the climatic zone consists of high daily temperature; vary in between $20^{\circ}C-30^{\circ}C$, year-round uniform precipitation, over 2000mm total rainfall and humidity between 60% - 95% [82]. When designing a building, it is important to consider the climatic conditions of the climatic zone where the building is located. Because the durability of the building directly depends on the exposed climatic conditions. Buildings in the tropical zone must be designed considering the tropical climatic conditions. But at present, building designers use European and American construction concepts for buildings in the tropical zone. Building designed without considering the climatic zone can result in thermal discomfort; which contributes to increased demand for artificial heating, cooling and ventilation and increases building energy demand [121].

Building walls and roofs are directly exposed to tropical climatic conditions [122],[123]. However, tropical climatic conditions are the nastiest climatic conditions for building materials; due to the high heating and continuous wetting conditions. These adverse climatic conditions accelerate weather degradation, scaling off, natural decay, mould and moss growth on building walls. Sri Lanka is an equatorial country with a tropical climate; consisting of high humid and moist conditions throughout the year [84]. These climatic conditions accelerate the mould and moss growth on building walls while reducing the aesthetic value of the building. Similarly, it affects the durability and strength of the building structure [62],[124].

2.7.3. Bio-receptivity of wall finishes

The ability of a material to be colonized by a living organism is defined as the 'Bioreceptivity' [125]. Bio-receptivity of the walling material depends on the intrinsic properties of the material [126],[127]. High humid and moist conditions facilitate the bio-receptivity of building walls. Water is the main requirement of organisms' metabolic activities; walling materials absorb water during rain or when exposed to flood conditions or high humidity environments. Building walls get wet when expose to heavy rain conditions under tropical climatic conditions; and wetted wall finishing materials increase the material bio-receptivity [128],[129]. Fungus growth on wall putty under wet conditions can be identified as the most common effect when exposed to highly humid environments and rain.

A fungus is a microscopic level living creature that is not visible to the human eye. But the fungus is visible to human eyes, only when they are colonized. Fungus belongs to the kingdom of fungi. They obtain nutrients from the attached surface degrading the material [130],[131]. The fungus can colonize most of the habitats in the world; including land, underground, soil, surface waters and oceans. Fungus growth is accelerated by high humid and moist conditions [131],[130]. Therefore, the fungus can successfully colonize building surfaces under highly humid and moist conditions.

2.7.4. Fungus growth on building walls

There are a few factors affecting fungus growth; such as the nature of the surface, exposure time, environmental temperature, moisture, humidity and fungus species. Tropical climatic conditions, including high humidity and moisture, are favorable for fungus growth. Therefore, the buildings in the tropical zone are more vulnerable to fungus growth [128],[132]. *Aspergillus niger* and other *Aspergillus spp* are the identified fungus species on building walls [128]. Fungus growth damages walling materials and results in health problems for the building occupants. Deterioration of the wall finishes due to fungus results in aging, strength reduction and discoloration; which reduce the aesthetic value of the building. And also, fungus growth adversely

affects indoor air quality. The fungus produces mycotoxins, spores and fragments which are toxic to human and results in health problems [128],[132].

Building walls should be protected from fungus growth to improve durability and occupants' health. Fungus growth facilitating factors should be identified and they should be eliminated to avoid fungus growth [132]. The most common fungus treatment method was applying a sealer or coating system on the wall surface to act as a moisture barrier to avoid moisture absorption. And also, chemicals are commonly used to control the existing fungus on walls. Sodium polyborate, dichlofluanid and iodopropinyl-butyl-carbamate (IPBC) are popular anti fungus additives [128]. Chemical treatments for fungus are not a sustainable and safe prevention method and green building concepts do not encourage chemical usage in sustainable building constructions and operations [133].

2.8. Summary

Water is one of the main requirements of humans and other living creatures. Life cannot be sustained without water, Therefore, ensuring the access of citizens to consume sufficient safe water is one of the responsibilities of a country. A sufficient amount of safe water consumption is essential for public health. A higher portion of the total population of most of the countries in the world has no access to a sufficient amount of safe drinking water; due to the lack of safe natural water resources. Therefore, they provide safe water to the residents after purifying surface water through DWTPs. Most of the residents in Sri Lanka have access to natural water resources; compared to the other countries in the region. The NWSDB is the responsible authority for drinking water purification and distribution. NWSDB consists of DWTPs located all over the country and the total drinking water production is estimated at 30 million m³ per month.

DWTP process consists of the steps of coagulation, flocculation, sedimentation, filtration and disinfection. Sludge is an unavoidable waste product generated in DWTPs. Total solid sludge production of DWTPs in Sri Lanka can be estimated at 1250m³ per month. The nature of sludge depends on suspended solids of raw water quality, coagulant type and chemicals added in the treatment process. Direct disposal of sludge into open lands and water bodies results in contamination of surface water and groundwater; reducing water quality and damaging aquatic ecosystems. Direct discharge of water treatment plant sludge is prohibited by Sri Lankan legislation; as sludge is classified under industrial waste. Different DWTPs practice different sludge treatment methods; such as sludge lagoons, sludge treatment plants, sedimentation tanks, etc.

DWTP sludge is a fine material similar to mud and clay in texture. The sludge is made with coagulant products and chemical elements, organic and suspended matter and microorganisms. Mud and clay are versatile materials that have been used since prehistoric times in the construction industry. Therefore, most of the research was conducted to identify the possible uses of DWTP sludge as an alternative to mud and clay. And also, research studies were conducted to analyze the possibility of DWTP sludge to incorporate into construction material production processes without affecting the properties and performances of the products. According to the research studies the main findings can be concluded as below;

- Sludge in different DWTPs has unique physical and chemical properties; the properties should be identified before incorporate into the material production process.
- DWTP sludge can incorporate in some construction material production processes up to an effective percentage; beyond the effective limit, products failed in standard performances.
- The effectiveness of DWTP sludge incorporation depends on the type of sludge (alum sludge or ferric sludge or lime sludge).
- A production process was not developed to utilize DWTP sludge as the main raw material of a product.

Soil-based walling materials are becoming the new trend of the present construction industry as an eco-sustainability material; such as mud concrete blocks, cement stabilized earth blocks and rammed earth walls. Soil-based walling materials perform as a moisture buffer; to control the humidity of the indoor environment; a thermal insulator; to provide thermal comfort, and improve the indoor air quality of the building. Walls constructed with soil-based walling materials require protective coating for both exterior and interior surfaces. Cement-based plasters are not compatible with soil-based walling materials. Cement-based plasters are highly rigid materials and do not facilitate moisture transmission and natural wall movements of soil-based building walls. As a result, detachment of cement-based plaster from the building wall reduces the optimum performance of soil-based walling materials. Cement-based pastes are not suitable for repairs of soil-based walls; soil-based walls repaired with cement-based mixtures were cracked easily and allowed trapped rainwater inside the wall. A soil-based wall finisher and a mixture; for repairs of soilbased building walls, are required to obtain optimum performances of soil-based constructions. Wall putty is a common wall finisher used in the present building sector; therefore, soil-based wall putty will be a sustainable and green solution.

Wall putty is a wall plastering material used as a wall finisher and filler. Putty is a thick mix of finely powdered calcium carbonate (whiting) and acid-refined linseed oil. Calcium carbonate is normally found as a white mineral (calcite) that occurs naturally in chalks, limestone and marbles. Calcium carbonate extraction results in environmental degradation and natural resource depletion. Advantages were identified as what can be obtained with soil-based wall finishers compared to conventional wall finishes. Soil-based wall finishes are applied to different building wall types; such as, rammed earth walls, adobe walls, concrete walls, wood walls, brick walls and straw bale walls. Soil base wall finishes perform as an insulator for heat, moisture, sound, odor and noise. And also, soil-based wall finishes creating thermal comfort for occupants against the temperature variation of the outdoor environment while maintaining optimum indoor humidity. Soil-based wall finishes. With the knowledge gathered from the literature, this research was conducted to develop a wall putty mix using DWTP waste alum sludge.

CHAPTER 03: SELECTION OF DWTP TO COLLECT SLUDGE

3.1. General

DWTP sludge is identified as a fine material similar to mud and clay in texture and made with organic and suspended matter, coagulant products, microorganisms and chemical elements. The properties of the sludge depend on;

- The geographical location of the plant
- Climatic conditions
- Raw water quality
- Suspended solids
- Treated water quality
- Water treatment process
- Coagulant type
- Chemical added in the treatment process
- Sludge treatment process

Therefore DWTP sludge of each plant consists of unique properties [3],[29] and the properties of the sludge in each DWTP should be studied separately. According to that, sludge samples from three different DWTPs; where practices three different sludge treatment processes, were collected; and a series of experiments were carried out to identify the properties. The physical properties of each sample were identified and the suitable DWTP to collect sludge for further studies was selected considering solid content and moisture content of the samples. Further physical and chemical properties of the selected sludge sample were studied after that. Selection of DWTP for collecting sludge will be discussed in this chapter.

3.2. Selection of a DWTP

In Sri Lanka, NWSDB is responsible for drinking water purification and supply throughout the country. NWSDB is consist of DWTPs located all over the country to secure a safe and sufficient water supply. This research study was conducted with the collaboration of NWSDB and required permissions were approved at the beginning of the study for DWTP visits, data collection and sample collections.

There are different sludge treatment methods available in the world such as sludge treatment plants, sludge settling tanks, sludge lagoons and constructed wetlands. According to NWSDB, sludge treatment plants, sludge lagoon and sludge drying beds are available in DWTPs in Sri Lanka. In this research three DWTPs were selected for sludge sample collection. Three DWTPs located in Ambathale, Biyagama and Kandana were selected and the plant locations are shown in Figure 3.1.



Figure 3.1: Locations of selected DWTPs; (a) Biyagama DWTP (b) Ambathale DWTP (c) Kandana DWTP

3.2.1. Ambathale DWTP

Ambathale DWTP is located at Kelani River bank and it is one of the major water sources of NWSDB with a capacity of 549,000m³ per day. The plant produces 518,000m³ of water per day for Colombo city. The plant consists of three water intakes and two storage reservoirs. A raw sludge sample was collected from the Ambathale plant as shown in Figure 3.2 (a).

3.2.2. Kandana DWTP

Kandana DWTP is located at Kandana, Horana and the plant is constructed under the Kalu Ganga water supply project. The plant is designed with a capacity of 60,000m³ per day and around 300,000 people who live in the southern part of Colombo benefit. There are large tanks called 'sludge lagoons', which store the raw sludge of the plant. In the lagoons' wet sludge is left to sun drying and dried sludge cake is produced. Dried sludge is sent for land fillings. Wet sludge samples from sludge lagoons were collected from Kandana DWTP (Figure 3.2 (b)).

3.2.3. Biyagama DWTP

Biyagama DWTP is located at Kelani River right bank and providing safe drinking water to cities covering the Northern part of the Western Province in Sri Lanka, including, Biyagama, Kelaniya, Kiribathgoda, Kadawatha, Ragama, Wattala, Kandana, Ja-Ela, Seeduwa and Ganemulla. The plant is designed to produce drinking water of 180,000m³ per day at its' full capacity. Raw water for the plant is extracted from the Kelani River; and the treated water is produced with a conversion rate of 95% and the 5% loss happened during raw water transmission, sludge dewatering and backwashing.

Biyagama DWTP is designed with a conventional treatment process; a physicchemical process, consisting of the major processes as, raw water extraction, pumping, screening, mixing chamber, flocculator, lamella clarifier, sand filter, clean water tank and distribution network. There is a sludge treatment plant in Biyagama DWTP. The sludge, generated in sedimentation and backwashing processes, is thickened through a sludge thickener. Sludge is dewatered in the sludge decanter and produces 'sludge cake' to send to landfill sites. Samples of the sludge cake were collected from the Biyagama DWTP (Figure 3.2 (c)).



Figure 3.2: Collected sludge samples; (a) Raw sludge from Ambathale DWTP (b) Sludge from sludge drying beds in Kandana DWTP (c) Dewatered sludge from Biyagama DWTP

3.3. Experimental Analysis of Sludge Properties

Sludge samples were collected from selected three DWTPs and the experiments were conducted in University of Moratuwa laboratories. Initial tests were conducted to identify the basic physical properties of the sludge samples to select the most suitable sludge sample to develop a wall putty. To that wet sieve analysis and moisture content were tested.

3.3.1. Moisture and solid content analysis

The workability of a soil-based mixture depends on the moisture content and the effectiveness of the mix depends on the solid content; therefore, the moisture and solid content of each sludge sample were tested. According to the literature weight basis, oven dry method was conducted to analyze the moisture content of soil-based mixtures according to the ASTM D 2216-98 standard [134][113]. The moisture content and solid content were analyzed according to the weight basis oven dry method [135],[136].

Methodology:

Three replicates of each plant sludge sample were taken and the wet weight (Ww) of each sample was measured and recorded. After that, the samples were oven dried, at 110°C temperature for 24 hours to remove moisture completely (Figure 3.3). Dried samples were weighted until the weight become constant. And then the dry weight (Dw) of each sample was measured and recorded. With the collected data, moisture content (MC) and the solid content (SC) of each sample was calculated according to equation 3.1 and 3.2. And the average value of the three replicates was calculated and the results are shown in Table 3.1.

MC(0) = Ww - Dw	Equation 2.1
$MC(\%) = \frac{W_W}{W_W}$	

SC(%) = 100 - MCEa	juation	3.2
--------------------	---------	-----



Figure 3.3: Moisture and Solid content measuring test

						_
Tahle 3.1+	Weight hasis	Moisture 91	nd Solid (content of	cludge e	amnlee
1 and 5.1.	Weight Dasis	monsture a	lu bonu (content or	siuuge s	ampics

Sample	Moisture Content (%)	Solid Content (%)
Ambathale DWTP	97.70	2.30
Biyagama DWTP	68.38	31.62
Kandana DWTP	86.32	13.68

Results and Discussion:

According to the test results, the Ambathale DWTP sludge sample has the highest moisture content (97.7%) and lowest solid content (2.3%). Kandana DWTP sludge sample has the second highest moisture content (86.32%) and second lowest solid content (13.62%). Biyagama DWTP sludge has the lowest moisture content (68.38%) and the highest solid content (31.62%).

According to the literature, bonding strength and shrinkage are directly affected by moisture content. When the moisture content increase bonding strength decreases while shrinkage increases [113],[114]. However, those properties depend on the clay content also; therefore, to analyze the particle size distribution of the solid fraction of each sludge sample, a wet sieve analysis was conducted.

3.3.2. Wet Sieve Analysis

The clay content of the sludge directly affects the properties of the sludge. According to the literature, increasing the clay content of a mix increase the bonding strength reducing shrinkage [137],[114]. Therefore, the clay sand ratio must be optimized to develop a clay plaster [113]. To that, analyzing the particle size distribution of the sludge is important. Wet sieve analysis was conducted to analyze the particle size distribution of each sludge sample according to IS:13077-1991.

Methodology:

Equal portions (1kg) of each wet sludge sample were measured and three replicates from each sample were analyzed. Measured sludge samples were put into a 75 μ m sieve. The wet samples were sieved continually stirring by hand and adding water. The sludge portion retained in the sieve was oven dried at 110^oC for 24 hours and measured the dry weight (Ws). The effluent was collected and sundried to reduce the water amount. The sundried sample was oven dried to obtain dry weight (Wc). With the recorded data, the sand percentage and clay and silt percentage of each sludge sample were calculated and the average of the three replicates was taken. The results are shown in Figure 3.4.

Results and Discussion:

According to the test results, silt and clay content is highest in Ambathale DWTP sludge and lowest in Biyagama DWTP sludge. The sand content is highest in Biyagama DWTP sludge and lowest in Ambathale DWTP sludge. When comparing the results with the standard values; as shown in Table 3.2, the particle size distribution of each sample is not in the standard range defined to develop mud plaster. Therefore, further improvements should be done to the sludge mix to optimize the particle size distribution to develop a wall putty.

Sludge collected from Ambathale and Kandana contains high moisture content and is required to remove moisture before use in the study. Therefore, Biyagama DWTP was selected to collect sludge considering the lowest moisture content and the highest solid content.



Figure 3.4: Particle size distribution of sludge samples; (a) Ambathale DWTP sludge (b) Biyagama DWTP sludge (c) Kandana DWTP sludge

Table 3.2:	Particle size	distribution	of DWTP	sludge
------------	---------------	--------------	---------	--------

	Sand	Silt + Clay	Reference
The standard for mud plaster - IS:13077-1991	30%-40%	58%-64%	[117]
Mud plaster	52%	48%	[11/]
Clay from quarries of Spain	43.4%	56.6%	
DWTP sludge from Sant Joan Despí DWTP in Spain	10.6%	89.4%	[50]
Ambathale DWT sludge	2.46%	97.54%	
Biyagama DWTP sludge	4.63%	95.37%	
Kandana DWTP sludge	3.14%	96.86%	

3.4. Summary

DWTPs practice different sludge removal processes and removing sludge has different properties identical to the sludge removal process. Therefore, DWTP should be selected to collect sludge for experiments considering favorable conditions to develop a wall putty. To that sludge samples were collected from Ambathale, Biyagama and Kandana DWTPs; and each plant practice different sludge removal methods. Laboratory tests were conducted to identify the properties of each plant sludge sample. After a comparison of the properties of samples, a DWTP was selected to collect sludge.

According to the test results, the Ambathale DWTP sludge sample has the highest moisture content and the lowest solid content compared to the other two samples. Biyagama DWTP sludge; which was collected from the sludge treatment plant, has the lowest moisture content and highest solid content; therefore, Biyagama DWTP was selected to collect sludge for the study. According to the particle size distribution of the sample, it was identified that further improvements to the sludge are needed to develop a wall putty. After that, laboratory experiments were conducted to identify the chemical and physical properties of Biyagama DWTP sludge as discussed in the next chapter.

CHAPTER 04: ANALYSING PUTTY PROPERTIES OF DWTP WASTE ALUM SLUDGE

4.1. General

According to the results of the initial tests; discussed in the previous chapter, Biyagama DWTP was selected to collect sludge for this research. Identical properties of the material should be studied to proceed with further experiments. Biyagama DWTP sludge was not utilized as the main raw material in the material production process. Therefore, standard experiments proceeded to identify the chemical and physical properties of DWTP sludge. Experimental procedures and the results will be discussed in this chapter.

4.2. Chemical Properties of DWTP Waste Alum Sludge

According to the literature, sludge properties vary with the sludge type (Table 2.1). And the sludge type depends on the chemical composition [31],[37]. Therefore, the identification of the chemical composition of DWTP sludge is important. To identify the chemical properties of DWTP sludge, X-Ray Fluorescence (XRF) analysis, X-Ray Diffraction (XRD) analysis and inductively coupled plasma mass spectrometry (ICP-MS) analysis were conducted. Chemical elements of the sludge sample were analyzed by conducting an XRF analysis. XRD analysis was performed to identify the minerology of DWTP sludge. Heavy metal identification was conducted by performing an ICP-MS analysis.

4.2.1. XRF Analysis

XRF analysis was conducted with the collaboration of the Sri Lanka Institute of Nano Technology (SLINTEC). The sludge sample was sun-dried and powdered. The powdered sample was sieved through a 100µm sieve to prepare a sample for the test. The test was conducted using HORIBA Scientific XGT- 5200 X-ray Analytical Microscope. The sample was pasted on the sample stage using double tapes and six different places were analyzed. The test was conducted with the parameters of, 200s live time; 100µm XGT Diameter; 50kV X-ray tube volume and P2 processing time.

Results and Discussion:

Chemical elements of the sludge sample were identified through XRF analysis [138] and the results are shown in Figure 4.1. According to the XRF test results Al, Si, S, K, Ca, Ti, Mn and Fe were detected. And Si (42.23%) is detected in the highest percentage, and Mn (0.25%) is detected in the lowest percentage in DWTP sludge. According to the literature, chemical elements of clay samples and sludge collected from different DWTPs were identified by conducting XRF analysis as shown in Table 4.1. When comparing the chemical composition of DWTP sludge in this study with the results of the previous research studies; it can be concluded that the percentages of each chemical element are in the range of chemical elements of clay and DWTP sludge samples analyzed in previous research studies.



4. Results

Sample 01

Element	Spot 1	Spot	t 2	Spot 3	Spot	4	Spot 5	Spot	6
13Al	27.09	25 0	98	22.99	26.4	7	21.84	24 6	50
14Si	40.78	39.8	85	41.79	37.2	0	45.01	45.1	.0
165	0.49	0.4	14	0.37	0.5	2	0.39	0.4	2
19K	2.45	4.4	4.45 7.91 1.53 1.3		1.85	2.5	7		
20Ca	5.52	4.8	84	4.27	5.5	5.58 4.75		5.8	9
22Ti	1.44	1.6	50	1.52	1.6	6	1.58	1.1	.8
25Mn	0.18	0.2	21	0.20	0.4	0	0.20	0.1	.7
26Fe	22.07	22.6	64	20.95	26.6	4	24.38	20.0)7
Chemical E	lement	Al	Si	S	K	Ca	Ti	Mn	Fe
Aass (%)		24.43	42.23	0.44	2.35	5.34	1.48	0.25	23.47

Figure 4.1: XRF analysis results

Table 4.1:	Chemical	elements	of I)W	ГР	sludge
------------	----------	----------	------	----	----	--------

Sample	Al (%)	Si (%)	S (%)	K (%)	Ca (%)	Ti (%)	Mn (%)	Fe (%)	Mg (%)	Na (%)	Reference
Red clay from Turkey	19.2	60.6	-	2.8	0.4	1.2	0.2	7.7	0.8	0.3	[113]
Yellow clay from Turkey	16.9	63.7	-	2.4	0.6	1.1	0.1	6.5	1.0	0.4	[113]
Clay from Brazil	33.4	52.4	-	0.3	-	2.1	0.1	0.7	0.1	0.1	[139]
Clay from quarries of Spain	17.3	59.4	-	2.9	2.6	1.0	0.1	6.8	1.5	1.1	[50]
Sludge from St. Joan Despi DWTP in Spain	15.8	53.7	-	3.2	14.4	0.7	0.1	5.0	3.6	0.4	[50]
Sludge from Urban DWTP in Morocco	40.0	25.0	0.8	1.6	0.8	0.3	0.2	3.1	0.7	0.6	[140]
Sludge from Bangkhen DWTP in Thailand	24.6	58.9	-	1.5	0.6	0.1	-	0.4	1.2	7.3	[141]
Sludge from DWTP from Egypt	16.6	34.4	2.6	0.4	4.5	0.9	0.2	8.2	0.7	0.2	[56]
DWTP sludge in the study	24.4	42.2	0.4	2.3	5.3	1.5	0.3	3.5	-	-	

4.2.2. XRD Analysis

XRD analysis for the DWTP sludge was conducted at the University of Moratuwa. XRD analysis was conducted to identify the chemical compounds of the DWTP sludge [138]. Three sludge samples were sun-dried and powdered. The powdered samples were sieved through a 100µm sieve to prepare a sample for the test (Figure 4.2 (a)). XRD analysis was conducted using the X-ray diffractometer- Eco D8 Advance for Structure Analysis (Figure 4.2 (b)). The test was conducted with the parameters of Co, Cu and Mo radiation, 2theta from 20° to 130°, step size of 0.03°, 10s per step and the 30 min total run time for each sample. Bruker diffraction EVA version 4.2 software package was used to collect data and interpret the diffraction patterns of the samples (Figure 4.2 (c)).



Figure 4.2: XRD analysis; (a) Sample preparation; (b) X-Ray Diffractometer; (c) XRD test report

Results and Discussion:

According to the XRD analysis results, chemical compounds of DWTP sludge were identified as, Kaolinite $(Al_2Si_2O_5(OH)_4)$, Quartz (SiO_2) , Calcite $(CaCO_3)$ and Ettringite $(Ca_6Al_2(SO_4)_3(OH)_{12}(H_2O)_{26})$. Compared to the findings of previous research, it can be concluded that the chemical composition same as other clay and DWTP sludge samples.

Sample	Clayish Earth	Yellow & Red clay	DWTP sludge	Sludge from Giza DWTP	DWTP sludge	Sludge in the study
Country	Portugal	Turkey	Brazil	Egypt	Spain	Sri Lanka
Chemical Compounds	Quartz Illite Kaolinite Hematite Dolomite Calcite	Kaolinite Illite	Kaolinite Quartz	Quartz Albite	Mica Calcite Quartz chlorite	Kaolinite Quartz Calcite Ettringite
Reference	[115]	[113]	[142]	[143]	[144]	

Table 4.2: Chemical compounds of DWTP sludge

4.2.3. Heavy Metal Analysis

Heavy metals are present in nature below the harmful level, when heavy metal concentration increases the harmful level, it creates adverse environmental impacts. Heavy metals do not generate through natural phenomena; heavy metals are added to the natural environment through anthropogenic activities [145]. Surface water

reservoirs in some countries and some locations are contaminated with heavy metals due to industrial disposals and soil contamination. However, when a DWTP collects water from a surface water source that is contaminated with heavy metals, the sludge generated from the plant included heavy metals. Because heavy metals have relatively high density compared to water and are settled with sludge in the purification process [146]. According to the literature, some DWTP sludge contaminated with heavy metals was identified as shown in Table 4.3. Contaminated sludge with heavy metals cannot be directly discharged or incorporated into the material production process [147].

There is a 'maximum permissible limit' define for heavy metals in each application of materials. The maximum permissible limit presents the applicable limit of heavy metals without harm. Exceeding the permissible level creates contamination associated with potential health risks or ecological risks [148]. The maximum permissible limit differs with the country, product, location, climatic conditions and purpose. Therefore, the contaminated sludge with heavy metals should be treated up to an acceptable level before discharge; also, in the literature, it was identified that sintering sludge above 1000°C can immobilize heavy metals in the material. But all these options are high-cost processes [31]. Therefore, it is important to analyze the heavy metal concentration of the DWTP sludge.

ICP-MS Analysis:

ICP-MS analysis was conducted with the collaboration of NWSDB and ITI to analyze the heavy metal concentration of the DWTP sludge. Samples were prepared according to the nitric acid digestion standard method. A 5g portion of the sample was weighed into a PTFE beaker, and 10mL of HNO₃ and 5ml of H₂O₂ were added. The mixture was heated on a hot plate. The residue was then dissolved in 25ml of deionized water. The solution was filtered and diluted into a 50ml and analyzed using NIOX300D inductively coupled plasma mass spectrometry. Test conducted with parameters as; 20-30min stabilization, focus lens and radio frequency power at 1550 W to minimize interference effects and to maximize the signal, carrier gas in auxiliary flow at 0.8 mL/min and nebulizer flow of 1.0 mL/min and 5.3 mL/min.

Reference	Classification	Concentration	Sludge use	Main finding			
[149]	Non- hazardous	Low	Ceramic brick	Slight increase in Cd concentration at the increased firing temperature			
[28]	Hazardous	High As and Fe content	Ceramic brick	Above 3% sludge-mix showed leaching			
[35]	Non- hazardous	High Al content	Lightweight composite concrete	Al concentration above that established by the Brazilian standard			
[31]	Non- hazardous	Low	Lightweight aggregate	The concentration of heavy metals below the regulatory thresholds			
[150]	Non- hazardous	Low	Ceramsite	The concentration of heavy metals below the regulatory thresholds			
[47]	Hazardous	Very high Pb, Zn, Cd, Ni, Cr and Cu content	Eco-cement clinker	The concentration of heavy metals below the regulatory thresholds			
[151]	Non- hazardous	High Cr and Zn content	Cement clinker	The concentration of heavy metals below the regulatory thresholds			
[48]	Non- hazardous	Very low	Concrete mix	The concentration of heavy metals below the regulatory thresholds			
[59]	Non- hazardous	High Fe content	Paving tile	The concentration of heavy metals below the regulatory thresholds			
[152]	Not reported	High Zn, Fe and Cr content	Geopolymer	Slight increase in Zn, Fe and Cr concentration when SiO2/Al2O3 ratio increases			

 Table 4.3: Contaminated DWTP sludge with heavy metals and identified solutions in the literature

Results and Discussion:

According to the results of the ICP-MS test, Cadmium (Cd) (<0.1ppm), Lead (Pb) (4.3ppm), Mercury (Hg) (<0.1ppm) and Arsenic (As) (0.9ppm) were identified as heavy metals in the sample. When comparing the heavy metal concentration with the findings of literature it can be concluded that the sludge has very few heavy metals with very low concentrations; as shown in Table 4.4. To identify the potential impact of the present heavy metals in the wall putty development process, the values were

compared with the maximum permissible limits defined by Sri Lankan legislation for wall top coatings. As observed, the heavy metal concentration of the alum sludge is lower than the maximum permissible limit as shown in Figure 4.3. Therefore, the heavy metal concentration presence of the sludge has no potential harm in the sense of human health and the ecological.

Source	Sample	As	Ba	Cd	Cr	Cu	Hg	Ni	Pb	Zn
[50]	Clay from quarries of Spain (mg/kg)	0.23	499	0.2	81	22.3			36	98
	DWTP sludge from St Joan Despí DWTP in Spain (mg/kg)		495	1.4	93	62	0.08	34	74	195
	Reference value based on uncontaminated soils in Netherland (mg/kg)	29	200	0.8	100	36	0.3	35	85	140
[146]	Agricultural soils from Southwestern Nigeria (mg/kg)	50	·	0.03	200	9.14		15.24	28.37	31.17
	Permissible limit (mg/kg)	2.4		10	81.2	50		100	200	250
[153]	Raw alum sludge from DWTP in Malaysia (mg/l)	0.02	0.06	0.01	0.02	0.47		0.05	0.03	0.21
[154]	Sintering red mud from alumina refining plant in Guizhou, China (ppm)	267.3	212	27.1	537.8	78.2	67.3	984.9	56.6	103.2
	Bayer red mud from alumina refining plant in Guizhou, China (ppm)	246.7	197.2	14.4	416.9	213.6	58.7	578.6	48	76.3
	DWTP sludge in the study (ppm)	0.9	-	<0.1	-	-	<0.1	-	4.3	-
	The maximum permissible limit defines by Sri Lankan legislation (ppm)	1.4	-	2	-	-	0.5	-	6	-
7		10								161
6		6	-2							
5	4.3	_								-
4 ع										
ä 3			-							
2			-						1.4	8.
1	<0.01			<	0.01	5		0.9	T (
Ū	Cadmium (Cd)	ad (Pb)	Heavy	Metal	lercury (H	lg)		Arseni	c (As)	

Table 4.4: Heavy metal concentration of DWTP sludge

Figure 4.3: Heavy metal concentration of sludge and maximum permissible limits

Maximum permisible limit

Presence in alum sludge sludge
4.3. Observational Shrinkage of DWTP Waste Alum Sludge

Shrinkage is one of the elementary properties of clay-based materials. According to the USCS soil classification system, DWTP sludge is categorized under clay. Therefore; the determination of the shrinkage of DWTP sludge is a major requirement before being used as a raw material. Shrinkage can be measured in two ways; volumetric shrinkage and observational shrinkage, as shown in Figure 4.4.



Figure 4.4: Measuring shrinkage [113]; (a) Volumetric shrinkage (b) Observational shrinkage

Volumetric shrinkage measures the loss in volume of clay-based mixtures after drying, and it depends on the clay content, moisture content and particle size [113]. Observational shrinkage is the drying shrinkage, after application of a clay-based mixture in a thin layer. Due to the drying shrinkage of the mixture drying cracks occurred on the surface after drying. By measuring crack formation time and analyzing the severity of the cracks observational shrinkage can be measured and compared.

Observed shrinkage of the mixture can be deduced by comparing the cracks formed on the surface in wet and after dried. Observational shrinkage test results reveal the adhesiveness of the mixture. The crack pattern and intensity of cracking can characterize the nature of the internal bonds of the mixture; and the bonding strength between the mixture and the substrate can characterize the adhesiveness of the mixture [113],[114].

Observational shrinkage test:

An observational shrinkage test was conducted to determine the drying shrinkage of DWTP wet sludge. In the literature, shrinkage was tested according to DIN 18947,2013 standards and the observational shrinkage was conducted by applying the wet wall finishing mix to the compatible base material [113]. In the test, cement sand plaster panels were cast with the dimensions of 1inch×1feet×1feet, to use as the base; as shown in Figure 4.5. Sand and cement were mixed in a 5:1 ratio and mixed with water. Casted samples were unmoulded after 24 hours and dried for seven days. After plaster panel preparation wet sludge was applied as a thin layer with 1mm thickness in an area of 8inches×8inches using a scraper. Three replicates were applied and dried for 24 hours and drying cracks were observed. To analyze drying crack initiation depends on the applying area, in the next step wet sludge was applied with three different areas (3inches×3inches; 4inches×8inches; 8inches; 8inches); as shown in Figure 4.7.



Figure 4.5: Sand cement plaster panel casting for observational shrinkage test

Results and Discussion:

Drying cracks of each sample were observed after 24 hours and observed that observational shrinkage doesn't change with the applied area. Samples applied with different areas, cracked with the same severity. Drying cracks of the wet sludge samples applied on the plaster base with an area of 8inches×8inches were observed to determine the observational shrinkage and the results are shown in Figure 4.6. according to the test results, it was observed that the sludge mix was completely cracked and detached from the plaster surface. According to the literature the wet sludge mix's adhesiveness and binding properties should be improved to develop a wall putty mix.



Figure 4.6: Observational shrinkage test results



Figure 4.7: Observational shrinkage with different application areas

4.4. Physical Properties of DWTP Waste Alum Sludge

According to the literature DWTP sludge's physical properties depends on;

- The geographical location of the plant
- Climatic conditions
- Raw water quality
- Suspended solids
- Treated water quality
- Water treatment process
- Coagulant type
- Chemical added in the treatment process
- Sludge treatment process

Out of these factors, the geographical location of the plant, treated water quality, coagulant type, chemicals, conditions of the water treatment process and sludge treatment process remain constant with time. But the climatic conditions, raw water quality, suspended solids and chemical application dosage change with time. Considering raw water quality and suspended solids, plat daily dosage of chemicals and coagulant is decided. Therefore, physical properties variation of sludge considering those variable factors should be studied. For that sludge, samples were collected daily from the Biyagama DWTP sludge treatment plant for six months duration (from 1st of January to 30th of June 2020) (Figure 4.8). Collected samples in plastic containers were transported to the University of Moratuwa and laboratory tests were conducted. In the DWTP, chemicals (Alum, Lime, Chlorine and Polymer) are added daily for the water purification process. Chemical addition data and climatic data (Highest and Lowest temperatures of the day and Humidity) of each day were collected from Biyagama DWTP data records. Recorded data (Appendix – I) were statistically analyzed to identify the variation of physical properties of sludge with chemicals apply in the plant and climatic factors. Statistical analysis was performed using 'Minitab 21' statistical software.



Figure 4.8: Sludge samples collected from DWTP

4.4.1. Shrinkage variation with chemicals and climatic factors

According to the literature, the volumetric shrinkage test was conducted according to DIN 18947,2013 test standard [113]. The wet sludge sample was mixed using a mechanical mixture to avoid compaction of the sample during transportation. Three replicates from each sludge sample were cast in a mold oil applied metal mold with the dimensions of $70 \times 70 \times 70$ mm³. Casted samples were unmoulded after 7 days and let to dry; as shown in Figure 4.9. Three-dimensional measurements were taken weekly until 35 days; where volume was constant after completely drying. Volumetric shrinkage was measured considering the difference between initial volume and dried volume after completely drying; as shown in Figure 4.10. The volumetric shrinkage of each mixture was calculated according to equation 4.1; where VS is volumetric shrinkage, V1 is the initial volume (70^3 mm³) and V2 is the final volume at the measuring date [17],[6]. The average of the shrinkage values of three replicates was calculated as the volumetric shrinkage of each sludge sample.

$$VS(\%) = \frac{V1 - V2}{V1}$$
....Equation 4.1



Figure 4.9: Casting cubes for volumetric shrinkage test



Figure 4.10: Volumetric Shrinkage Test; (a) Initial volume (b) Volume after completely dry

Results and Discussion:

According to the test results, the volumetric shrinkage variation of DWTP sludge is shown in Figure 4.11. according to the test results, DWTP sludge volumetric shrinkage varies between 59.82% and 93.62%; with an average value of 83.91%. When comparing the shrinkage value with previous findings; as shown in Table 4.5, DWTP sludge shrinkage is very high. According to the literature volumetric shrinkage depends on moisture content and clay content. Shrinkage of DWTP sludge is high due to the high moisture content and clay content.



Figure 4.11: Volumetric shrinkage test results

Table 4.5: Volumetric shrinkage of DWTP waste alum sludge

	Shrinkage (%)	Reference
Yellow Clay from Turkey	14.4	[112]
Red Clay from Turkey	13.7	[115]
Montmorillonite-rich clays from India	12 - 23	[155]
Sludge from Chin-Tan DWTP in Taipei	40 - 50	[149]
DWTP sludge in the study	59.82 - 93.62	

Regression analysis was performed, to statistically analyze the variation of sludge shrinkage, with chemicals applied in the DWTP and climatic conditions. According to the results, the p-value of humidity was lower than 0.05 (P<0.05) and the p-value of other factors was higher than 0.05 (P>0.05) (Appendix – II). Therefore, humidity (Pvalue = 0.006) is the only significant climatic factor effect on shrinkage variation of DWTP sludge, from selected factors. Other climatic factors and chemicals are not significant in the volumetric shrinkage variation of DWTP sludge. There is a positive correlation (Coef = +1.163) between sludge shrinkage and humidity (Figure 4.12). Scatterplots of sludge shrinkage versus chemicals and climatic factors are shown in Figure 4.13. According to the scatterplots regression line of humidity and shrinkage has the highest R^2 value ($R^2 = 0.64$). Humidity is the concentration of water vapor present in the air. Increasing humidity in the air reduces the moisture evaporation from the sludge; therefore; the moisture content of sludge is high on days with high humidity compared to low humid conditions. High moisture content results in high shrinkage of DWTP waste alum sludge. Therefore, increasing humidity increase the volumetric shrinkage of DWTP waste alum sludge.



Figure 4.12: Scatterplot of sludge shrinkage Vs humidity



Figure 4.13: Scatterplots of sludge shrinkage Vs chemicals and other climatic factors

4.4.2. Moisture content variation with chemicals and climatic factors

The moisture content of each sample was tested by conducting the weight basis ovendry method according to the ASTM D 2216-98 standard [135],[136]. Three replicates from each sample were tested and the average was calculated. The wet weight (Ww) of each sample was measured and recorded. After that, the samples were oven dried, at 110°C temperature for 24 hours; until the weight become constant. And then the dry weight (Dw) of each sample was measured and recorded. With the collected data, the moisture content of each sample was calculated according to equation 3.1; and the results are shown in Figure 4.14. According to the test results, the moisture content of DWTP sludge varies between the values of 62.31% and 84.72% with an average of 76.26%.



Figure 4.14: Moisture content test results

Regression analysis was performed to statistically analyze the variation of sludge moisture content with chemicals applied in the DWTP and climatic conditions. According to the results, all the p values were greater than 0.05 (P>0.05) (Appendix – III). Therefore chemicals (Alum, Lime, Chlorine and Polymer) and climatic factors (Highest and Lowest temperatures of the day and Humidity) are not significant factors effect on the moisture content of DWTP waste alum sludge. Scatterplots of sludge moisture content versus chemicals and climatic factors are shown in Figure 4.15. The sludge treatment plant process dewatered sludge; therefore, the moisture content of the dewatered sludge depends on the sludge treatment process conditions.



Figure 4.15: Scatterplots of sludge moisture content Vs chemicals and climatic factors

4.5. Summary

Biyagama DWTP was selected to collect sludge for the experiments. According to the findings of the previous chapter Biyagama DWTP sludge has the lowest moisture content and the highest solid content compared to sludge samples collected from Ambathale and Kandana DWTPs. Experiments in this chapter were conducted to analyze the putty properties of DWTP waste alum sludge. Therefore, laboratory experiments were conducted to study the chemical and physical properties of DWTP sludge. According to the chemical composition analysis (XRF analysis), Si (42.23%) is the main chemical element included in DWTP sludge; and the chemical element variation in the range of values identified in the literature. According to the XRD analysis test results, DWTP sludge chemical composition consists of kaolinite, quartz, calcite and ettringite. And also, the chemical composition is significant to the findings in the literature. According to the heavy metal analysis test results Cadmium (Cd) (<0.1ppm), Lead (Pb) (4.3ppm), Mercury (Hg) (<0.1ppm) and Arsenic (As) (0.9ppm) were identified in the sludge sample as heavy metals; but the available quantities are lower than the maximum permissible level define for wall top coatings in Sri Lanka. Therefore, there is no harmful impact on human health or the environment when developing a wall putty with DWTP waste alum sludge.

Laboratory tests were conducted to identify the shrinkage and moisture content of the sludge. According to the findings of the observational shrinkage test, observational shrinkage is not affected by the applied area and the DWTP sludge layer is completely cracked and detached from the plaster surface. Volumetric shrinkage of DWTP sludge varies between the values of 59.82% and 93.62% with an average of 83.91% according to the volumetric shrinkage test results. Moisture content test results reveal that DWTP sludge moisture content varies between the values of 62.31% and 84.72% with an average of 76.26%. Statistical analysis was performed to identify significant factors that affect on physical properties of the sludge. According to the statistical analysis results, it is revealed that humidity is the significant climatic factor effect on the shrinkage of DWTP sludge with a positive correlation.

CHAPTER 05: OPTIMIZING PUTTY PROPERTIES OF DWTP WET ALUM SLUDGE

5.1. General

After the identification of the putty properties of DWTP sludge, the next step of the research was to optimize the putty properties of raw wet sludge. According to the results of the laboratory experiments, DWTP sludge has high shrinkage; both observational and volumetric shrinkages, and low adhesiveness. To develop the raw sludge to a wall putty observational and volumetric shrinkages must be reduced and adhesiveness should be increased. Therefore, the next steps of the study were performed to enhance the putty properties of the wet raw sludge and the experimental procedures and results will be discussed in this chapter.

5.2. Selection of Additives

According to the literature, research studies were conducted to study the possible usages of DWTP sludge, as a raw material in the production industry. Different construction materials; such as bricks, roof tiles, lightweight aggregates, cement, concrete and geopolymers, were tested for incorporating DWTP sludge [23],[32],[57],[59]. Each research concluded that the integration of DWTP sludge with some products succeeded and did not succeed with some products; the shrinkage of DWTS was identified as the major issue of concern when used in the construction material production process [113].

Shrinkage is one of the elementary properties of the clay-based materials production process. Shrinkage can be measured in two ways; volumetric shrinkage and observational shrinkage [113]. Observational shrinkage is measured as the drying shrinkage after application of clay plasters and the volumetric shrinkage measures loss in volume after drying quantitatively. Shrinkage depends on the clay content, moisture content and particle size. According to historical evidence natural fibers, binders and additives were mixed with clay plasters to overcome shrinkage [95],[102]. As the DWTP sludge is characterized under clay; according to the USCS soil classification, it is required to reduce shrinkage. The effectiveness of different additives to reduce the

shrinkage of DWTP sludge can be studied before being used as a raw material [23],[113].

There is historical evidence available all over the world for soil and mud-based plasters and wall finishes which were lasting for thousands of years. According to the historical evidence, natural additives were added to soil mixtures to overcome the shrinkage as shown in Table 5.1. A detailed inventory of additives was prepared after referring literature. From those, a few additives were selected for the study considering the availability and preferable properties.

Source	Soil Plaster Type	Country	Additives	
[07]	Plaster in Caves;	Bhaja Caves,	Raw soil, Slacked lime, Rice	
[95]	Interior and	India	adhesive	
		Mali and	Wet mud fermented with rice	
[96]	Wall Plaster	African	husks and proteins (milk and	
		Countries	animal blood)	
[30]	Blood Lime	Europe, China,	Animal blood and lime	
[30]	Plaster	and Mexico	Ammai blood and mile	
[102]	Plaster of Sigiriya	Sri Lonko Doddy husk and account fi		
	Fresco	SII Laiika	I addy husk and cocondt hoer	

Table 5.1: Additives in historical soil-based plasters

5.3. Physical Additives to Improve Putty Properties

DWTP sludge should be with low shrinkage and high adhesiveness to apply as a wall putty mix. As the sludge has high shrinkage, the wet mix should be improved to reduce shrinkage. Physical additives are used to reduce shrinkage in historical soil-based plasters. Therefore, the shrinkage reduction with physical additives was quantitatively measured and analyzed in this section. For the test few different physical additives were selected; including sand and three natural fibers.

5.3.1. Analyzing the effectiveness of sand

Sand is the most commonly used aggregate in the construction industry and it is a naturally available material in Sri Lanka [156]. River sand is used in this study and the sand was sun-dried to remove moisture. The dried sand sample is sieved from a 300µm

sieve; to eliminate coarse particles and have a uniform aggregate size distribution in all samples (Figure 5.1).



Figure 5.1: Sieved sand for the test

5.3.2. Volumetric shrinkage variation of wet alum sludge with sand

Nine different mixtures were prepared by mixing sand and wet sludge according to the mix design shown in Table 5.2. Mixtures were prepared by mixing them in a mechanical mixer (Figure 5.2). The volumetric shrinkage test was conducted according to DIN 18947,2013 standards [113]; as described in the previous chapter. Three cubes (three replicates) from each mix were cast to measure volumetric shrinkage. Three-dimensional measurements of each cube were taken after 4weeks (after complete drying of the cubes) and volumetric shrinkage was calculated using Equation 4.1.



Figure 5.2: Mixing with mechanical mixer and cube casting for volumetric shrinkage test

Table 5.2: Mixing ratios of sand and sludge

Mix	Mix 1	Mix 2	Mix 3	Mix 4	Mix 5	Mix 6	Mix 7	Mix 8	Mix 9
Sludge (%)	100	95	90	80	70	60	50	40	30
Sand (%)	0	5	10	20	30	40	50	60	70

Results and Discussion:

Volumetric shrinkage variation with different sand percentages is shown in Figure 5.3. According to the test results, shrinkage becomes 0 at 70% of sand. Possible mixtures to conduct the volumetric shrinkage test could be prepared only up to 70% of sand; cubes required for the volumetric shrinkage test were unable to be cast with mixtures beyond 70% sand. Particle size distribution (clay/sand ratio) of the mix affected the workability and the clay content affect the adhesiveness of clay-based mixtures. Therefore, wet sludge mixtures over 70% of sand are not workable enough to cast cubes as the increasing sand percentage lowers the adhesiveness of the mix [113],[117]. Volumetric shrinkage of wet sludge is reduced with increasing sand but reduces the adhesiveness of the mix. Although sand positively effects to reduce shrinkage it is not effective to improve the adhesiveness of DWTP wet sludge. To determine the optimum sand mixing ratio with wet sludge, a ball drop test was conducted.



Figure 5.3: Volumetric shrinkage variation with sand

5.3.3. Analyzing optimum sand/wet alum sludge mixing ratio (Ball drop test)

Clay/sand ratio of the mixture effect the shrinkage and adhesiveness of the mix. Increasing sand percentage reduce shrinkage and adhesiveness also; therefore, the optimum sand/wet sludge mixing ratio should be identified. The ball drop test is one of the field tests used to predetermine the effective clay sand mixing ratios to develop clay-based plasters [113]. In this test, a ball of the mixture with around 4cm diameter is released freely from a height of 1.5m, onto a flat surface. Various results can be observed in this test, as shown in Figure 5.4. The mixtures with high clay content resulted in a dropped sample with a shiny and smooth surface and slightly cracked with no cracking (Figure 5.4 (a)). If the mixture is poor in clay content, results in a dropped sample with a dull surface and fragments into rubbles (Figure 5.4 (b)). The mixture with optimum clay content results in a dropped sample with a small number of surface cracks and no fragmenting (Figure 5.4 (c)). A mixture with optimum clay and sand ratio is suitable to develop a plaster. A ball drop test was conducted for three replicates of all nine mixtures and the test results are shown in Figure 5.5.

Results and Discussion:

According to the ball drop test results, optimum clay and sand content was observed between the sand percentage of 30% to 60%. Below 30% of sand percentage, the clay content is high in mixtures and beyond 60% clay content is low; and the mixtures were not consistent with suitable putty properties. Therefore, wet sludge mixtures with a sand content of 30% to 60% were suitable for a wall putty mix. However, the identified mixture has optimum shrinkage performance, but the adhesiveness should be improved in the mix to develop a wall putty.



(a) High clay content

(b) Poor clay content

(c) Optimum clay content

Figure 5.4: Ball drop test results demonstration [113]



Figure 5.5: Ball drop test results

5.3.4. Analyzing the effectiveness of natural fibers

To study the effect of physical additives on the volumetric shrinkage of sludge, natural fibers were also tested. According to the literature, it was identified that natural binders were added in historical soil-based plasters to reduce shrinkage. And also, it was identified that fiber type, fiber length and concentration depend on the effectiveness of natural fibers [157][158]. Natural fibers used as additives in previous research studies and historical plasters were studied through a literature survey. A natural fiber inventory was prepared to refer to literature; as shown in Table 5.3, and selected possible three natural fiber types for the test. Three fibers were selected considering availability, cost, production process and intrinsic properties. Sugarcane fiber, Coir fiber and Banana fiber (Figure 5.6) were selected for the test.

Natural fiber	Density (g/cm ³)	Tensile Strength	Young's modulus	Elongation (%)	Reference
Coir	1.15-1.46	95 - 593	2.8 - 6	15 - 51.5	
Kenaf	1.2 - 1.45	223-930	14.5 - 53	1.5 - 6.9	
Flax	1.38- 1.5	343 - 2000	27 - 103	1.2 - 3.3	
Bamboo	0.6 - 1.4	140 - 1000	11 - 35.91	1.4 - 3.7	
Abaca	1.5	400 - 980	6.2 - 20	1 - 10	
Jute	1.23-1.5	187 - 800	8 - 78	1 - 3.1	[170]
Sisal	1.2 - 1.5	363 - 855	9 - 38	1.9 - 7	[159]
Ramie	1 - 1.55	220 - 1000	24.5 - 128	1.2 - 4	[160]
Cotton	1.5 - 1.6	287 - 800	5 - 13	3 - 10	[101]
Banana	1.35	500 - 914	8 - 32	1.5 - 10	
Hemp	1.35-1.5	270 - 1110	23.5 -90	1 - 4.5	
Pineapple	1.5	170 - 1627	1.5 - 82.5	1 - 3	
Henequen	1.2	430 - 570	10.1 - 16.3	3.7 - 5.9	
Bagasse	1.2 - 1.25	20 - 290	17 - 27.1	1 - 3	



Sugarcane fiber

Coir fiber

Banana fiber

Figure 5.6: Natural fibers selected for the study

5.3.5. Volumetric shrinkage variation of wet alum sludge with natural fibers

Selected fibers were collected in dry conditions. The fibers were cut in 10mm lengths to obtain uniform fiber length in the samples. According to the literature 10mm fiber length was concluded as the optimum fiber length for fiber additives in soil-based plasters [162],[155]. Selected fibers were cut using a 'Laboratory Cutting Mill SM 200' (Figure 5.7). Fiber samples were cut into small pieces and fed into the cutting mill. The mill operates at 50Hz speed and the cut fibers were collected after being sieved from the bottom sieve with the size of 10mm.



Figure 5.7: Cut natural fiber samples; (a) Laboratory cutting mill SM 200; (b) Cut fiber samples with 10mm

Ten different mixtures were prepared by mixing fiber and wet sludge in different percentages and three sets of samples were prepared for tree fiber types as shown in Table 5.4; mix 1, was the controller which includes only sludge. Sludge and fiber amounts were measured by weight according to the percentages. A volumetric shrinkage test was conducted according to DIN 18947,2013 standards [113]; as

described in the previous chapter, for three replicates of each mix, an average value was calculated for each mix.

Sugarcane Fiber Coir Fibe			Coir Fiber			Banana F	iber	
Mix	Sludge (%)	Fiber (%)	Mix	Sludge (%)	Fiber (%)	Mix	Sludge (%)	Fiber (%)
1	100	0	1	100	0	1	100	0
2	99	1	2	99	1	2	99	1
3	98	2	3	98	2	3	98	2
4	97	3	4	97	3	4	97	3
5	96	4	5	96	4	5	96	4
6	95	5	6	95	5	6	95	5
7	94	6	7	94	6	7	94	6
8	93	7	8	93	7	8	93	7
9	92	8	9	92	8	9	92	8
10	91	9	10	91	9	10	91	9
11	90	10	11	90	10	11	90	10

Table 5.4: Mixing ratios of natural fibers and sludge

Results and Discussion:

Volumetric shrinkage variation with different fiber types and percentages is shown in Figure 5.8. According to the test results, fiber can be effectively mixed up to 10% by weight with DWTP sludge. Volumetric shrinkage of the sludge is reduced when increasing the fiber percentage of the mixture. Shrinkage reduced gradually up to 8% and shrinkage reduction rate reduced beyond 8% can be observed for every fiber type. And 8% was identified as the optimum fiber mixing ratio. When comparing three fiber types, the lowest shrinkage is recorded with Banana fiber while the highest shrinkage is recorded with Sugarcane fiber. Therefore, Banana fiber is identified as the most effective fiber type to reduce volumetric shrinkage of DWTP sludge when compared to the other two fiber types. But then consider the mass production process and banana and coir fiber is produced as an industrial raw material. Therefore, there is high potential for to use of sugarcane fiber on a mass scale in putty production with wet alum sludge than the other two fiber types.



Figure 5.8: Volumetric shrinkage variation with natural fiber type

5.4. Binders to Improve Putty Properties

Binder is any material or substrate which holds other materials together to form a bond mechanical or chemical, by adhesion or cohesion [158],[163]. Binders can tightly bind clay particles of the sludge to overcome the shrinkage. Binders can be simply classified as;

- Organic bitumen, animal and plant glues, polymers
- Inorganic lime, cement, gypsum, liquid glass

According to the literature it was found that natural binders extracted from plant-based materials were added to soil mixtures to develop soil-based plasters [164]. Historical evidence is available all over the world for soil-based plasters which last for thousands of years. In Sri Lanka, historical evidence was found in the 'Sigiriya' rock castle built by King Kashyapa in the 5th century AD. The mirror wall and the frescoes were soil-based plasters that last for thousands of years [165].

After analyzing literature on the compositions of historical soil-based plasters, three natural binders and one synthetic binder were selected for the study; as shown in Figure 5.9. Natural binders were extracted according to the indigenous technologies and the styrene-acrylic binder was collected from the available market; the binder was selected as the synthetic biner of the study, considering its wide usage in the paint production industry [163].



Figure 5.9: Binders selected in the study

Dawulkurundu leaf extraction: Dawulkurundu leaf extraction was used in the food production process and as a binding material in ancient wall paints according to indigenous knowledge of Sri Lankan society [164]. The binder from Dawulkurundu was extracted from leaves. Dawulkurundu leaves were collected and crushed into small pieces. The crushed leaves were blended with a small amount of water in a mechanical blender and Dawulkurundu leaf extraction was prepared.

Bhoomi bark extraction: Bhoomi is a medicinal plant and the plant extraction is used in ayurvedic medicine; as a binding agent with medicinal properties [166],[167]. Binder from Bhoomi was extracted from tree bark. Bhoomi tree bark was cut off from the tree and it was cut into pieces and blended with a small amount of water to prepare Bhoomi bark extraction. A styrene acrylic binder was purchased from the market.

Bael fruit extraction: Binder of the Bael fruit was used in the historical soil-based plasters, including the Sigiriya mirror wall and plaster of the Sigiriya fresco [164]. Binder of Bael is included in Bael fruit and the binder was extracted from ripened Bael fruits.

5.4.1. Analyzing putty properties variation of wet alum sludge with binders

To analyze the putty properties variation of wet sludge with binders, experiments were conducted to test moisture content, observational shrinkage and fungus growth test. Five sludge mixtures were prepared by mixing DWTP wet sludge with the binders according to the mix design shown in Table 5.5. Mixtures were prepared by mixing them in a mechanical mixer. After the preparation of sludge mixtures, observational shrinkage, moisture content and fungus growth were tested for tree replicates of each mix.

	Mix 1	Mix 2	Mix 3	Mix 4	Mix 5
DWTP Sludge	Х	Х	Х	Х	Х
Styrene Acrylic Binder	0	Х	0	0	0
Dawulkurundu Leafe Extraction	0	0	Х	0	0
Bhoomi Bark Extraction	0	0	0	Х	0
Bael Fruit Extraction	0	0	0	0	Х
x - Added in the mix, o - Not added in the mix					

Table 5.5: Mix design for wet sludge mixtures with natural binders

5.4.2. Moisture content variation of wet alum sludge with binders

The weight basis oven-dry method was used to measure moisture content and the test was conducted according to the ASTM D 2216-98 standard; as mentioned in previous chapters. Test results are shown in Figure 5.10. According to the test results moisture content increase when mixing binders with wet sludge compared to the moisture content of raw sludge.



Figure 5.10: Moisture content variation of wet sludge mixtures with binders

5.4.3. Observational shrinkage variation of wet alum sludge with binders

An observational shrinkage test was conducted to determine the drying shrinkage of DWTP wet sludge with the presence of a binder. The test was conducted according to DIN 18947,2013 standards as mentioned in previous chapters. Mixtures were applied on sand cement plaster panels and dried for 24 hours. After samples were completely dry, drying cracks on the surface were observed. Shrinkage test results are shown in Figure 5.11. According to the observational shrinkage test results cracking patterns differed with the binder type. When the sludge was applied in wet raw condition, it was completely cracked without binding to the panel surface after 24 hours. The rest of the mixtures prepared by mixing with binders were cracked with different crack patterns according to the binder type after 24 hours. However, the mixtures with binders were bound to the panel more tightly than the wet raw sludge. The presence of a binder reduced the severity of the cracks while increasing the adhesiveness of DWTP wet sludge.



Figure 5.11: Observational shrinkage of wet sludge mixtures with binders

5.4.4. Analyzing fungus growth of wet alum sludge with binders

A fungus growth test was conducted according to the ASTM D3273-94 test standards [168],[62]. Cement plaster panels were cast according to the size of 5cmx15cmx2.5cm. Mixtures were applied on one surface of each sample with 1mm thickness and cured the samples until completely dry (Figure 5.12). After 14 days of the incubation period, test samples were vertically hung in the chamber as shown in Figure 5.12. Samples were hanged for four weeks (one month) and after that samples were taken out from the chamber and the putty surfaces were captured. Putty surfaces were compared before and after to identify fungus growth on the putty surface and the fungus growth was rated according to the rating system defined by ASTM D 5590-00 standard; as shown in Table 5.6 [115].



Figure 5.12: Sample preparation and fungus growth test

Scale	Description	Fungus growth Percentage
0	None	0
1	Trace of growth	<10%
2	Light growth	10 - 30%
3	Moderate growth	30 - 60%
4	Heavy growth	>60%

Table 5.6: Fungus growth rating system – ASTM D 5590-00 [115]

Results and Discussion:

The results of the fungus growth test are shown in Figure 5.13. According to the observations, fungus growth of the samples applied mix 1 and 2 were rated on scale 1; as a trace of fungus growth observed in the samples. Fungus growth of the samples applied mix 3,4 and 5 were rated as scale 4; as heavy fungus growth was observed on the samples. According to the observations, it can be concluded that the fungus growth of wet sludge increases with natural binders. Synthetic binder (Styrene acrylic binder) does not facilitate fungus growth as natural binders. Natural binders increase the organic matter content of the mix which facilitates fungus growth. Therefore, natural binders cannot be recommended as an effective additive to develop wall putty mix with DWTP sludge due to the increasing fungus growth.



Figure 5.13: Fungus growth test results

5.5. Analyzing Putty Properties of Wet Alum Sludge Mixture

Experiments were conducted to identify the effectiveness of physical additives and binders to optimize the putty properties of the DWTP wet sludge. Considering the findings, one synthetic binder and two physical additives were selected as shown in Table 5.7.

Synthetic Binder	Physical A	Additives
Styrene Acrylic Binder	Sand	Fiber
 A synthetic binder Does not facilitate fungus growth Reduce drying cracks of wet sludge Improve adhesiveness of wet sludge 	 Dry sand sieved from a 300µm sieve was used Reduce volumetric shrinkage of wet sludge Improve clay/sand ratio of wet sludge mix 	 A waste product in the sugar production process Fiber cut with 10mm length pieces Reduce volumetric shrinkage of wet sludge

Mix Design:

Eight different mixtures were prepared by mixing selected additives with sludge in a mechanical mixer (Figure 5.14), according to the mix design shown in Table 5.8. After the preparation of mixtures, experiments were conducted to test the volumetric shrinkage and observational shrinkage of each sample.



Figure 5.14: Mixtures preparation in a mechanical mixer

Mintune		Ing	gredients		
wiixture	Sludge	Sand	Fiber	Synthetic Binder	
Mix 1	Х	0	0	0	
Mix 2	Х	Х	0	0	
Mix 3	Х	0	Х	0	
Mix 4	Х	Х	Х	0	
Mix 5	Х	0	0	Х	
Mix 6	Х	Х	0	Х	
Mix 7	Х	0	Х	Х	
Mix 8	Х	Х	Х	Х	
x - Added in the mix, o - Not added in the mix					

Table 5.8: Mix Design

5.5.1. Volumetric shrinkage variation with additives

A volumetric shrinkage test was conducted according to DIN 18947,2013 standards [113]; as described in the previous chapter, for three replicates of each mix. Volumetric shrinkage was calculated using equation 4.1. and an average value was calculated for each mix. Volumetric shrinkage test results are shown in Figure 5.15. According to the test results, the lowest volumetric shrinkage was observed with mix 4; which was prepared by mixing both sand and fiber with wet sludge. And the mix 8; which was prepared by mixing the binder with both physical additives has the second lowest shrinkage.



Figure 5.15: Volumetric shrinkage variation of wet alum sludge with additives

5.5.2. Observational shrinkage variation with additives

An observational shrinkage test was conducted to determine the drying shrinkage of each mix. The test was conducted according to DIN 18947,2013 standard as mentioned in previous chapters. Three replicates of each mixture were applied on sand cement plaster panels and dried for 24 hours. After samples were completely dry, drying cracks on the surface were observed. Shrinkage test results are shown in Figure 5.16. according to the test results it can be observed that mix 8; was prepared by mixing both physical additives along with the binder. When comparing the mixtures with and without a binder; mixtures with the binder (Mix 5,6,7 & 8) are more tightly bound to the plaster surface than the mixtures without a binder (Mix 1,2,3 & 4).



Figure 5.16: Observational shrinkage variation of wet alum sludge with additives

5.5.3. Wet alum sludge mixture with optimum putty properties

According to the test results, it was identified that volumetric shrinkage can be reduced by mixing additives with wet sludge. And also, it was identified that the adhesiveness of the wet sludge mix improved with additives. Therefore, the putty properties of the wet sludge can be improved by additives, and an optimum wall putty mix with DWTP wet sludge was developed considering shrinkage (observational and volumetric) test results, as shown in Figure 5.17. The developed wet sludge mixture was tightly bonded with the plaster surface and low with drying cracks. And scale-off was not observed from the dried putty surface. But the mix consists of a few defects; as macro level drying cracks are present on the dried putty surface and fibers pop up from the surface reducing the smoothness of the surface. Existing wall putty mixtures consist of microcracks. And also, existing putty surfaces have a smooth finish. Therefore, the wall putty mix developed with the wet alum sludge is not consistent with potential putty properties. The next section of this research was conducted to develop a dry powder with putty properties, using wet alum sludge.



Figure 5.17: Wet sludge mix with optimum putty properties

5.5. Summary

DWTP sludge properties were identified in the previous chapter. According to the results, shrinkage was identified as the main drawback of wet sludge when developing a wall putty. Therefore, the next experiments were conducted to improve the putty properties of wet sludge. According to the identified properties of the wet sludge, adhesiveness should be increased while reducing the drying shrinkage of the mix. According to the literature and historical evidence, it was found that additives; such as fibers, sand and binders were added to soil-based plasters to reduce shrinkage while increasing adhesiveness. Therefore, different additives were selected and their performances to improve the putty properties of wet sludge were studied.

Sand and natural fibers were selected as physical additives and mixtures were prepared by mixing with wet sludge. Different mixtures were prepared by mixing sand in different percentages and volumetric shrinkage and ball drop tests were conducted for each mix. According to the test results, the addition of sand reduced the volumetric shrinkage and the shrinkage becomes zero at 70% sand percentage, but it is not practically applicable in developing a putty mixture. To identify the optimum sand percentage, a ball drop test was conducted. According to the ball drop test results, it was identified that wet sludge mixtures with a sand percentage of between 30% to 60% have optimum clay and sand contents to develop wall putty mix. Although the mix has optimum shrinkage performances, the adhesiveness of the mix should be developed.

Three natural fibers were selected for the study and mixtures were prepared by mixing fiber with wet sludge according to the developed mix design. A volumetric shrinkage test was conducted for each mix. According to the test results addition of fiber reduced the volumetric shrinkage and banana fiber is the most effective natural fiber type to reduce volumetric shrinkage of wet sludge compared to coir fiber and sugarcane fiber.

After identifying the effectiveness of physical additives, the effectiveness of binders was tested. According to historical evidence, natural binders can be identified when added to historical soil-based plasters. Three different plant-based natural binders were selected for the study. And a synthetic binder was selected to compare the

performances of natural binders. Different wet sludge mixtures were prepared by mixing natural binders and a synthetic binder with wet sludge according to the developed mix design and tests were conducted to measure moisture content, observational shrinkage and fungus growth. According to the test results crack patterns of wet sludge mixture vary with the binder type and the moisture content of the mix increased with the addition of natural binders. But fungus growth in the mix increase with natural binders. Therefore, natural binders are not effective as an additive to develop wall putty with DWTP sludge.

After studying the physical additives and binders, next experiments were conducted to optimize the putty properties of the wet alum sludge. Wet alum sludge was mixed with the synthetic binder and physical additives in different mixing ratios. According to the test result, it was revealed that physical additives reduced drying shrinkage while binders increase the adhesiveness of wet sludge mixtures. According to that the most effective method to enhance the putty properties of wet sludge is adding both synthetic binder and physical additives. With the findings an optimum wet sludge mixture with low volumetric shrinkage, low drying cracks and high adhesiveness was developed; but the mix has few defects as the presence of small surface cracks and fibers pop up from the surface when applied as a wall putty. Therefore, the wall putty mix developed with the wet alum sludge is not consistent with potential putty properties. The next section of this research was conducted to develop a dry powder with putty properties, using wet alum sludge.

CHAPTER 06: DEVELOPMENT OF DRY POWDERED ALUM SLUDGE

6.1. General

In the previous chapter, experiments were conducted to enhance the putty properties of DWTP wet alum sludge to develop wall putty from wet alum sludge. According to the test results, a wet alum sludge mix was developed with optimum putty properties, but some defects were identified. According to the high moisture content and low solid content, the shrinkage of wet alum sludge is considerably high. Therefore, it is difficult to develop wall putty directly with wet alum sludge. When the wet alum sludge dries to remove moisture, it is hard as stone according to the activation of alum (Figure 6.1). Therefore, the sludge cannot be dry directly to produce a dry powder. Therefore, the next step of the study was conducted to develop dry powdered alum sludge applicable as a wall care putty. Thermal properties of DWTP alum sludge were identified and a process was developed to produce dry powdered alum sludge. The experimental procedures and results will be discussed in this chapter.



Figure 6.1: Wet and dry sludge

6.2. Thermal Properties of DWTP Waste Alum Sludge

Heat has a significant influence on soil-based materials such as clay, soil and mud. The exposure to heat with different temperatures for different time durations alters the soil-base material properties significantly as shown in Figure 6.2 [169],[153]. Therefore, identification of the thermal alteration properties of DWTP alum sludge is essential to developing dry powdered alum sludge[170].



Figure 6.2: Thermal alterations of soil at different temperatures [169]

DWTP alum sludge was oven dried at 105°C for 24 hours duration until the weight become constant to remove the moisture (Figure 6.3 (a) & (b)). Dried sludge samples were heated in a muffle furnace at ten set temperatures between 100°C to 1000°C (temperatures as 100°C, 200°C, 300°C, 400°C, 500°C, 600°C, 700°C, 800°C, 900°C and 1000°C) (Figure 6.3 (c) & (d)). The variations of the density and color due to the thermal alterations of the sludge at ten different temperatures were measured and observed.



Figure 6.3: Heating sludge for determination of thermal alterations

6.2.1. Color Variation

Images of sludge samples were taken at three different stages. The first image was taken when collecting wet sludge samples from the Biyagama DWTP. The second image of the sludge sample was taken at the end of heating; when the sample was under dry conditions. After that, the dry samples were submerged in water at room temperature for 24 hours duration and the third image of each sample was taken under wet conditions. After arranging the images according to the heated temperature, color variation was observed (Figure 6.4). According to the observations color of the sludge is variants with the temperature. From 100°C to 400°C the color becomes darker in both dry and wet conditions. Beyond 400°C color become lighter up to 700°C and become constant up to 1000°C.



Figure 6.4: Colour variation of sludge with heat
6.2.2. Density Variation

Density can be defined as the mass of a given volume of substance under given conditions. There are various test methods for density measurements of different materials. Dried sludge is consisting of unevenly shaped solid particles; therefore, the density calculation was conducted according to EN 1097-3 standard [115],[117]. Dried sludge particles were powdered and sieved using a 300µm sieve (Figure 6.5). After that, powdered sludge samples were compacted in a box with a known volume (8cm³) and the weight was measured (Figure 6.6). After recording the weights of each sample, the density was calculated using equation 6.1; where Wb refers to the empty weight (g) of the box, Ws refers to the weight (g) of the sludge compacted box and V refers to the volume (cm^3) of the box. Measurements were taken three times (three replicates) for each sample and the average value was calculated.

Density $(gcm^{-3}) = \frac{Ws - Wb}{V} = \dots$ Equation 6.1



(a) Grinding uneven shapes dried sludge particles

(b) Powdered dried sludge



(d) Powdered dried sludge





(a) Empty box with known volume

(b) Weighing empty box

(c) Sludge powder compacted box

(d) Weighing sludge powder compacted box

Figure 6.6: Density measuring test of dried sludge

Results and Discussion:

After calculating the density values of each heated sludge sample, values were statistically analyzed to study the density variation with the heating temperature. And the results are shown in Figure 6.7. When comparing the density values of heated sludge samples, the highest density was achieved at the temperature of 900°C and the lowest density was achieved at 500°C. According to the graph, density values of the heated sludge samples are gradually reduced from 100°C up to 500°C and beyond the 500°C, density again gradually increases up to 900°C temperature. According to the results, it can be concluded that density increased when applying heat to sludge and the lowest density was recorded at 500°C temperature.



Figure 6.7: Density variation of dried sludge with heating temperature

6.3. Overcome Alum Activation of DWTP Waste Alum Sludge

In DWTPs, raw water from surface water sources is collected and treated to drinking quality. In the surface water sources, negatively charged colloids are suspended and they do not sediment under gravity due to their lightweight. To neutralize the negative charge of the colloids, alum is added to the treatment process as a positively charged coagulant [41],[171]. After adding alum, neutralized colloids are bonding together as shown in Figure 6.8, and the flocculation process will be started. In the flocculation process, micro-flocs are forming and become macro-flocs. In the water treatment process a polymer is added to tightly bind the macro-flocs together and the colloids sediment with the gravity due to the increase of weight in the flocculation process. Sediments of the water treatment process are removed as sludge and when the sludge is dried it is harder due to alum activation [3],[36]. Therefore, alum activation should be overcome to produce dry powdered alum sludge. Next experiments were conducted to identify effective lubricant and optimum processing conditions to develop a dry powder production process.



Figure 6.8: Coagulation process

6.3.1. Selection of a lubricant

When the wet sludge was dried as it is, it harder than a stone; according to the activation of alum, which is very difficult to grind and needed more energy to grind to a powder. Therefore, a method should be developed to dry alum sludge in powder form. According to the literature, there was no available method to produce dry powdered sludge overcoming the alum activation. To overcome the alum activation a lubricant can be added to the sludge before applying heat. When mixing a lubricant with wet sludge, lubricant particles should be attached to the sludge particles. The lubricant should be boiled out letting the sludge particles loosely bound when adding heat. Therefore, the lubricant should be a non-drying oil; to remain liquid phase in the at room temperature without solidifying to mix effectively with the wet sludge, and with a lower boiling point; to remove from the dry powder without influencing the putty properties of the powder [169],[153].

When selecting the lubricant oil petroleum oils were excluded considering the harmful emissions when burning; from the other non-drying oils, coconut oil was selected due to its lower boiling point (177°C - 232°C) compared to other non-drying oils. Coconut oil is locally produced non-drying oil and has no harmful emissions results when burning. Therefore, coconut oil is selected considering the favorable conditions as mentioned in Figure 6.9.



Coconut Oil

- Nondrying oil
- *Boiling point* 177^oC 232^oC
- No harmful emissions
- Locally available

Figure 6.9: Coconut oil as the lubricant

6.3.2. Dry powdered sludge production with coconut oil

As the first step of the study, dry powdered sludge was produced after mixing with coconut oil to study the effectiveness of coconut oil as a lubricant to overcome alum activation of DWTP sludge. In the dry powder production process, wet sludge was mixed with 10% of coconut oil in a mechanical mixture and the dry powdered sludge was produced following the steps shown in Figure 6.10.



Figure 6.10: Production process of dry powdered sludge with coconut oil

In the process oil mixed sludge samples were burned at ten different temperatures (temperatures as 100°C, 200°C, 300°C, 400°C, 500°C, 600°C, 700°C, 800°C, 900°C and 1000°C) in a muffle furnace and ten dry powder samples were prepared as shown in Figure 6.11. Parallel to that, ten samples of dry powdered sludge samples were prepared without mixing coconut oil (Figure 6.11). And the properties of each sample were tested and the results were compared to identify the effectiveness of coconut oil as a lubricant to overcome alum activation.

Temperature	100ºC	200°C	300°C	400°C	500°C
Sludge					
Sludge mixed with oil					540 500
Temperature	600°C	700°C	800°C	900°C	1000°C
Temperature	600°C	700°C	800°C	900°C	1000°C

Figure 6.11: Comparison of burned sludge samples with and without coconut oil

6.3.3. Identification of effectiveness of coconut oil as a lubricant

Coconut oil was selected as the lubricant for the study. The effectiveness of coconut oil to overcome alum activation should be studied next. Therefore, the next experiments were conducted to study the performance of coconut oil as a lubricant. Properties of dried sludge samples; burned with and without coconut oil, were tested and compared with both results. In the study density and easiness to grind to powder were tested. The appearance of the samples was captured and arranged according to the heated temperature; as shown in Figure 6.11, and both samples were compared. According to the test results, sludge samples burned without mixing coconut oil were harder with the increase the temperature. And also, the particles were tightly bound and need extra energy to grind to a powder. When compared the samples with sludge samples; which burned after mixing coconut oil, particles were loosely bound and easy to grind to a powder. The sample dried at 100°C temperature consists of coconut oil; as the boiling point is higher than 100°C, therefore, the powdered sample cannot be mixed with water due to the hydrophobic properties of the containing oil. According to the observations both samples have the same color at each temperature. And also, samples burned beyond 800°C temperature, are harder again and do not easy to grind. Therefore, around 700°C is the optimum temperature to burn sludge mixtures produced in the dry powder production process. According to the test results mixing coconut oil with wet sludge mixture before burning has an effect to reduce the hardness of the sample when increasing the temperature up to 700° C without effect on the color of the sample. And can be concluded that coconut oil is an effective lubricant to overcome alum activation of DWTP sludge.

6.3.4. Identification of optimum temperature

The density of each sludge sample burned with coconut oil was tested and study the variation with temperature to study the optimum temperature for the dry powder production process. Samples were powdered and measurements were taken according to Figure 6.5 and Figure 6.6. After the collection of data, density was calculated according to equation 6.1. The density values of each heated sludge sample with oil were compared with the density value of the heated sludge sample without oil. And the results are shown in Figure 6.12. When compared to the density values, all the values of heated sludge samples with oil are significantly lower than the density values of heated sludge samples without oil. The highest density was achieved at the temperature of 900°C and the lowest density was achieved at 700°C. According to the graph, density values of the heated sludge samples gradually decrease from 100°C up to 700°C and gradually increase from 700°C up to 900°C. According to the results, it

can be concluded that the lowest density was recorded at 700°C temperature which is the optimum temperature to produce dry powdered sludge.





Figure 6.12: Density variation of sludge burned with coconut oil

6.3.5. Identification of optimum coconut oil mixing ratio

After identifying the optimum temperature to burn wet alum sludge after mixing with coconut oil, the next experiment was conducted to identify the optimum mixing ratio of coconut oil for dry powdered alum sludge production. In the test, five mixtures were prepared by mixing oil with wet sludge from 2% oil to 10%. Mixtures were mixed in a mechanical mixer. Mixtures were burned at 700°C and burned sludge particles were ground using a mortar and pestle to powder and the results are shown in Figure 6.13. According to the test results, samples with 2% and 4% of the oil were too hard and unable to grind to a powder using mortar and pestle. The sample with 6% oil was not too hard as samples with 2% and 4% oil and ground but was unable to grind to a fine powder. Samples with 8% and 10% of the oil were not too hard compared to other samples and were easily ground to powder using mortar and pestle.



Figure 6.13: Dry powdered sludge production variation with coconut oil percentage

According to the test results samples with coconut oil percentage beyond 8% can be easily powdered and the easiness to powder increase with increasing oil percentage. Up to 8% of oil percentage is not sufficient to overcome alum activation totally and 8% is the minimum oil percentage to overcome alum activation in the dry powder production process. Furthermore, mixing oil in higher percentages will make the process unproductive. Therefore, the optimum coconut oil mixing percentage can be identified as 8% for a productive dry powder production process.

In the study, a process to overcome alum activation was identified and the optimum conditions of the process were defined according to the results obtained so far. As the next step, the putty properties of the produced dry powder should be studied to develop a wall putty mix. Therefore, the next experiments were conducted to identify the putty properties of the developed dry powdered alum sludge.

6.4. Putty properties of dry powdered alum sludge

Following the developed method coconut oil was mixed in 8% by weight with wet alum sludge and a sludge mixture was prepared. And then the mixture was oven dried at 105°C for 24 hours to remove moisture. The moisture-removed mixture was burnt in a furnace up to the temperature range of 100° C – 800° C. The burnt mixture (Figure 6.14) was ground with mortar and pestle and sieved through a 300µm sieve to produce alum sludge dry powder as shown in Figure 6.15. Eight samples of dry powdered alum sludge were produced by heating with eight different temperatures in the range of 100° C – 800° C. Next experiments were conducted to identify the putty properties of developed sludge powders; to identify optimum dry powder to develop the wall putty. Therefore, an observational shrinkage test was conducted for each sample.







(a) Burned sludge

(b) Grinding burned sludge (c) Sieving powdered sludge Figure 6.15: Sludge dry powder production process



6.4.1. Analysing observational shrinkage

An observational shrinkage test was conducted for each dry powder mix to identify the powder sample with the potential to develop wall putty. The dry powders were mixed with water and apply to sand cement plaster surface using a scraper as shown in Figure 6.16. Sample burned at 100°C cannot be mixed with water, due to the presence of oil; therefore, an observational shrinkage test was conducted for the other seven samples. Samples were applied on plaster panels with an area of 25 cm^2 (5 cm x 5cm) and three replicates from each mix. After application samples, wet appearance was captured and plaster panels were let dry for 24 hours duration and dry appearance was captured. Wet and dry appearances were compared to identify drying shrinkage and the results are shown in Figure 6.17. According to the test results, drying cracks were observed on the samples burned at temperatures from 200°C to 600°C, and no drying cracks were observed on the samples burned at 700°C and 800°C. Therefore, the dry sludge powder sample burned at 700°C temperature was the optimum dry sludge powder sample to develop wall putty. However, the sample dried with zero cracks but it was scaled off from the surface; and the putty properties of the mix should be optimized. Next experiments were conducted to select an effective stabilizer to develop wall care putty from the dry sludge powder mix.



(a) Sludge dry powder

(b) Adding water and mixing

(c) Applying on plaster panel

(d) Drying

Figure 6.16: Dry powder mix preparation for observational shrinkage test



Figure 6.17: Results of observational shrinkage test

6.4.2. Analyzing effectiveness of stabilizers

Optimum dry sludge powder was identified in the previous section, and there are drying cracks observed on the dry surface of the mix. But a small scale off from the surface was observed. Therefore, a stabilizer should be selected to add to the mix to enhance the putty properties of the mix. In this step of the study, two stabilizers were selected and an observational shrinkage test was conducted for each mix mixing both stabilizers. And the effectiveness of the stabilizer was analyzed according to the observations. In the study, two stabilizers; lime and cement, were tested (Figure 6.18). Mixtures were prepared by mixing dry sludge powder with both stabilizers (with the

percentage of 10% by weight) separately (Figure 6.19). Samples were left dry for two days and drying cracks and scale off of the surface were analyzed. The results are shown in Figure 6.20.



Lime

Cement





Figure 6.19: Stabilization of dry powdered sludge



Figure 6.20: Observational shrinkage with stabilizers

According to the test results, drying cracks were observed on all samples burned at temperatures from 200°C to 600°C, and no drying cracks were observed on the samples burned at 700°C and 800°C. And also, the dry sludge powder samples burned at 700°C and 800°C temperatures were tightly bound to the plaster surface and did not scale off. Therefore, both stabilizers effectively enhanced the putty properties of the dry sludge powder sample. Furthermore, the samples mixed with lime have lighter colors compared to other samples, and samples mixed with cement have darker colors compared to other samples. Both stabilizers affect the color of the mix while enhancing putty properties. According to the results of the test, it can be concluded that a wall putty mix can be developed with the sludge dry powder with a stabilizer (lime or cement).

6.5. Summary

In the previous chapter, experiments were conducted to develop a wet sludge mix with putty properties. With the experiments, a wet sludge mix was developed but the mix was with few defects and has no potential to apply as a wall putty mix. Therefore, a process was required to produce dry powdered sludge overcoming alum activation of wet sludge. In this chapter, experiments were conducted to develop a dry powdered sludge with putty properties. DWTP wet alum sludge is harder when dry, due to the activation of alum and it is difficult to powder.

In the first step of this chapter, the thermal properties of DWTP sludge were studied. In the test, ten dried sludge samples were prepared by burning at ten different temperatures between 100°C and 1000°C. Color variation and density of each sample were measured. According to the observations color of the sludge is variants with the temperature and from 100°C to 400°C the color becomes darker in both dry and wet conditions. Beyond 400°C color become lighter up to 700°C and become constant up to 1000°C. When comparing the density values, the highest density was achieved at the temperature of 900°C and the lowest density was achieved at 500°C. Density values of the heated sludge samples are gradually reduced from 100°C up to 500°C and beyond the 500°C, density again gradually increases up to 900°C temperature. According to the results, it can be concluded that density increased when applying heat to sludge and the lowest density was recorded at 500°C temperature.

Next experiments were conducted to select a lubricant and to analyze its effectiveness of that. Coconut oil was selected as the lubricant for the dry powder production process considering the preferable intrinsic properties and availability. Dry powdered sludge samples were prepared by burning wet sludge mixtures after mixing with coconut oil at ten different temperatures. According to the test results, coconut oil has the potential to overcome alum activation without effect on the color of the sample. The density of each sample was measured and all the values of heated sludge samples with oil are significantly lower than the density values of heated sludge samples without oil. According to the results, the lowest density was recorded at 700°C temperature and it is concluded as the optimum temperature to produce dry powdered sludge. And also,

the optimum coconut oil mixing percentage was identified as 8% by weight, for a productive dry powder production process according to the test results.

Next experiments were conducted to identify the putty properties of developed sludge powders; to identify optimum dry powder to develop the wall putty. Therefore, an observational shrinkage test was conducted for each sample. According to the test results, drying cracks were observed on the samples burned at temperatures from $200^{\circ}C$ to 600°C, and no drying cracks were observed on the samples burned at 700°C and 800°C. Therefore, the dry sludge powder sample burned at 700°C temperature was the optimum dry sludge powder sample to develop wall putty. However, the sample dried with zero cracks it was scaled off from the surface; and the putty properties of the mix should be enhanced. Next experiments were conducted to select an effective stabilizer; lime and cement were selected to analyze their effectiveness as stabilizers. According to the results, it can be concluded that a wall putty mix can be developed with the sludge dry powder with a stabilizer. Both stabilizers effectively enhance the putty properties of the dry powdered sludge and also add cement to darker the color of the mix while adding lime to lighter the color of the mix. In this chapter, an effective method was developed and optimum conditions were identified to produce dry powdered sludge from Biyagama DWTP sludge. The developed putty properties should be compared with existing putty products and the real-scale performances of the developed putty mix should be studied, and the relevant experiments were conducted in the next chapter.

CHAPTER 07: DEVELOPMENT OF WALL PUTTY AND REAL-SCALE APPLICATION

7.1. General

The shelter is a main requirement of humans and from prehistoric times shelters were used to prevent extreme natural conditions; such as rain, wind, and UV radiation, and to protect from animals. In the beginning, primitive structures were used as shelters and with civilization building constructions were developed and improved with the technology development. In prehistoric times, a shelter was a protective structure but present structured buildings are designed with wide verities of activities including living, entertaining and manufacturing [61]. In conventional building construction concepts, walls are constructed using different walling materials and applied with three coatings; a sealer, primer and topcoat, as a wall finisher and a protector (Figure 1.1) [172],[162].

Walling materials are for building wall construction and there are solid walling materials and lightweight walling materials available in the world. In Sri Lanka, solid walling materials are used as the country has tropical climatic conditions [80]. In building constructions commonly used walling materials are burnt clay brick, cement block, Cabook block, cement stabilized earth block and mud concrete block (MCB) [61]. A sealer is a coat designed to inhibit capillary action of the porous substrate and sand cement plaster is the most commonly applied sealer. The wall plaster provides a uniform base for the primer while preventing the migration of extractives from the wall into the primer and top coat. The primer is a coating applied to the sealer to ensure the adhesion of the top coat to the sealer. Wall care putty and lime are commonly applied as a primer in Sri Lanka. The primer protects the plaster while providing uniform base for the top coat. The top coat is the finish coat and wall paints are applied as the top coat. Top coat is the final protective coating and present an aesthetically appealing decorative effect to the wall [162].

After a series of experiments, a method was developed to produce dry powdered alum sludge. The developed dry powdered sludge is applicable as a wall putty for the primer.

And also, it was identified that lime and cement are workable stabilizers for the putty mix. The developed putty mix has natural soil color. Most common commercial putty products consist of white color but color wall putty is also available; which was developed to add aesthetics to the wall. Color putty applies according to the architectural purposes and client requirements. However, the properties of the developed wall putty should be studied and compared with the properties of existing wall putty products. Therefore, experiments were conducted to identify the properties including physical properties and durability performances of the developed wall putty and the existing wall finishes. With the findings, the real-scale behavior of the developed wall putty can be analyzed. And the experimental procedures and results will be discussed in this chapter.

7.2. Selection of Stabilizer

The dry powder developed using DWTP waste sludge can apply as a wall putty. As discussed in the previous chapter, a stabilizer should be added to improve the putty properties of the mix. According to the literature, wall putty materials are developed with the characteristics of; high bonding strength, moisture barrier, high durability and low bio-receptivity. To improve those putty properties, the existing wall putty products are also applied after mixing with cement as a stabilizer. Next experiments were conducted to identify a workable stabilizer and the optimum mixing ratio for the sludge wall putty.

Stabilizers such as, cement, lime and styrene acrylic binder (Figure 7.1) were selected for the study. Cement was selected, as it is the stabilizer used with the existing wall putty products. The existing wall putty products are made of lime and styrene acrylic binder is developed with a styrene-acrylic polymer which is used to develop primers; so, lime and styrene acrylic binder were included in the experiments. Ten putty mixtures were prepared by mixing sludge dry powder with selected stabilizers according to the mix design as shown in Table 7.1. Mix 1 was the controller was prepared without a stabilizer. Dry powder and stabilizers mixing proportions were prepared by weight basis according to the mix design and mixed by a mechanical mixer. Experiments were conducted to test the bonding strength, bio receptivity and durability of mixtures; and three replicates from each mixture were used for the experiments.



Figure 7.1: Stabilizers selected for the experiments

Lime

Styrene Acrylic Binder

Mix	Sludge Putty (%)	Cement (%)	Lime (%)	Binder (%)
1	100	-	-	-
2	95	5	-	-
3	90	10	-	-
4	85	15	-	-
5	90	-	10	-
6	75	-	25	-
7	50	-	50	-
8	90	-	-	10
9	75	-	-	25
10	50	-	-	50

Table 7.1: Mix design for sludge putty mixtures with stabilizers

7.2.1. Bonding strength variation with stabilizers

Bond strength refers to the amount of adhesion between bonded surfaces which generally involves a determination of the stress required to rupture a bond formed by an adhesive between the two surfaces. Bond strength is measured in the terms of force needed to separate a layer of material from the base where it is bonded. The nature of the coating surface affects the bonding strength [173]. The bonding strength of a wall finisher can be described as the adhesion of the wall finishing layer to the plaster surface. When the wall finisher is applied to a plaster surface there are several physical, mechanical and chemical forces created in between the finishing coat and the plaster surface. Those forces bonded the two surfaces together and the bonding strength measured the failure point of those forces where the material was subjected to scaled off from the surface [174],[175].

When considering the bonding strength, cohesion and adhesion are important interactions to be determined. Cohesion refers to the bonding forces of the molecules of the same material. And adhesion refers to the bonding forces between two different surfaces. Both bonding strengths are important to consider when developing a wall finisher [176]. There are different bonding strength test methods for materials with different purposes.

A lap shear test was conducted to test the bonding strength of the mixtures, according to the ASTM D5868-014 test standards [116]. Plaster panels were prepared with the dimensions of 15cm×5cm×2.5cm (Figure 7.2 (a)) and putty mixtures were applied on one surface with 1mm thickness as shown in Figure 7.2 (b). After drying putty samples, a high solid epoxy adhesive was applied to the putty surface and two samples with the same mix were bonded together as shown in Figure 7.2 (c). Samples were cured and after the adhesive dried completely the lapse shear test was conducted using a universal testing machine (Figure 7.3 (a)). A force is applied to the directions as shown in Figure 7.3 (b), and increased until the bond fails. The highest possible joint strength; where the materials separate, is identified, and also the bond-breaking pattern was observed. There are four possible crack patterns identified according to the literature as shown in Figure 7.3 (c) [116]. By identifying the bond-breaking crack patterns adhesiveness of the putty mix can be critically analyzed as shown in Table 7.2 [177].



(a) Plaster panel preparation

(b) Putty mixtures application

(c) Samples fixing with adhesive

Figure 7.2: Sample preparation for bonding strength test of sludge putty mixtures with stabilizers



Figure 7.3: Bonding strength test of sludge putty mixtures with stabilizers

 Table 7.2: Analysing the nature of bonds between putty mix and plaster according to the crack pattern [116]

Result	Nature of the bond			
Fiber-Tear Failure	Bond strength of putty < Bond strength of plaster material			
Stock-Break Failure	Bond strength of putty > Bond strength of plaster material			
Adhesive Failure	The low bond strength between putty and plaster material			
Cohesive Failure	The high bond strength between putty and plaster material			

Three replicates of each mix were tested and the average bonding strength value was calculated. Results of the test were statistically analyzed (Appendix – IV) and the variation is shown in Figure 7.4. According to the test results, mix 1 (sludge putty without stabilizer) has the lowest strength value, and mix 10 (putty mix stabilized with 50% styrene acrylic binder) has the highest strength value. All ten mixtures' bonding strength values are significantly different from each other. All the mixtures mixed with stabilizer have a higher bonding strength value than the sludge putty mix without stabilizer; also, the bonding strength value increase with the increase of stabilizer proportion in each mixture. When comparing the three stabilizers, a significant increase in bonding strength value is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. However, it can be concluded that stabilizers increase the bonding strength of the sludge putty mix and the crack pattern should be analyzed to select the most workable stabilizer.



*Vertical bars indicate the standard errors of the mean bonding strength value (n = 3)*Means do not share a same letter are significantly different

Figure 7.4: Bonding strength variation of putty mixtures with stabilizers

In the bonding strength test, three crack patterns were observed from the four possible crack patterns and the frequency of each crack pattern was statistically analyzed as shown in Figure 7.5. According to the observations fiber-tear failure, cohesive failure and adhesive failure are observed within the samples and stoke-break failure is not observed. Cohesive failure is observed with the highest frequency (50%) and fiber-tear failure is observed with the lowest frequency (20%).

The nature of the bond of each mix was analyzed according to the crack pattern as shown in Table 7.3. Adhesive failure is observed in mix1, mix2 and mix 5; so, there is a low bond strength between the putty and the plaster panel of each mix. Fiber-tear failure is observed in mix3 and mix6; so, the bond strength of putty mix is less than the bond strength of plaster. Cohesive failure is observed in mix4, mix7, mix8, mix9 and mix10; so, there is a high bond strength between the putty and the plaster panel of each mix. Therefore, the putty mixtures observed cohesive failure is tightly bound to the plaster panel, and the most workable stabilizers are included also the sludge putty

mix do not tightly bound to the plaster panel without a stabilizer. All the putty mixtures prepared by mixing styrene acrylic binder are tightly bound to the plaster panel and other mixtures are tightly bound with the highest percentages only (cement 15% and lime 50%). Therefore, cement and lime should be added in high amounts to increase the bonding strength of the putty mixture while the bonding strength can be increased with a lower amount of styrene acrylic binder; styrene acrylic binder is the most workable stabilizer for sludge putty compared to cement and lime.



Figure 7.5: Crack patterns variation of putty mixtures with stabilizers

Mix	Sludge Putty (%)	Cement (%)	Lime (%)	Binder (%)	Observed crack pattern	Nature of the bond of the mix
1	100				Adhesive	Low bond strength between
T	100	-	-	-	Failure	putty and plaster material
ſ	05	5			Adhesive	Low bond strength between
4	95	5	—	—	Failure	putty and plaster material
					Fiber-tear	Bond strength of putty <
3	90	10	_	_	Failure	Bond strength of plaster
					Tantare	material
4	85	15			Cohesive	High bond strength between
_	05	15	-	-	Failure	putty and plaster material
5	00		10		Adhesive	Low bond strength between
3	90	—	10	—	Failure	putty and plaster material
6	75		25		Fiber-tear	Bond strength of putty <
0 /3	-	25	-	Failure	material	
-	50		50		Cohesive	High bond strength between
/	50	_	50	_	Failure	putty and plaster material
o	00	-		10	Cohesive	High bond strength between
o	90	-	-	10	Failure	putty and plaster material
0	75			25	Cohesive	High bond strength between
9	15	-	-	23	Failure	putty and plaster material
10	50			50	Cohesive	High bond strength between
10	50	-	-	30	Failure	putty and plaster material

Table 7.3: Analysis of the bond of putty mixtures variation with stabilizers

7.2.2. Bio receptivity variation with stabilizers

The ability of a material to be colonized by a living organism is defined as the 'Bioreceptivity' [125]. Bio-receptivity of the walling materials is depending on the intrinsic properties of the material [126],[127]. High humid and moist conditions facilitate the bio-receptivity of building walls. Water is the main requirement for living organisms for their metabolic activities. Walling materials absorb water when they are exposed to water during rain or flood conditions or exposed to a high relative humidity environmental condition. Building walls become wet when expose to heavy rain conditions under tropical climatic conditions; and wetted wall finishing materials increase the material bio-receptivity [128],[129]. Fungus growth on wall putty under wet conditions can be identified as the most common effect when exposed to highly humid environments and rain.

A fungus is a microscopic level living creature that is not visible to the human eye. But the fungus is visible to human eyes when they are colonized. Funguses are classified under the kingdom fungi; as eukaryotes. Fungus derives nutrients through the degradation of biological residues containing the surface where the fungus is attached [130],[131]. The fungus can colonize most of the habitats in the world; including land, underground, soil, surface waters and oceans. Fungus growth is accelerated by high humid and moist conditions [131],[130]. Therefore, the fungus can successfully colonize building surfaces under highly humid and moist conditions. Fungus growth results in damage to the building walls while resulting in health problems for the building occupants. The sludge putty is developed to apply to a wall as the primer and the primer should be a protective layer for the wall. Therefore, a fungus growth test was conducted to study the bio receptivity variation of the sludge putty with stabilizers.

Bio receptivity of putty mixtures was tested by conducting a fungus growth test according to the ASTM D3273-94 test standards [168],[62]. Cement plaster panels were cast according to the size of 5cmx15cmx2.5cm as shown in Figure 7.6. Putty mixtures were applied on one surface of each sample with 1mm thickness and cured the putty surface until completely dry.



Figure 7.6: Sample preparation for fungus test of sludge putty mixtures with stabilizers

After the preparation of samples, a fungus test chamber was prepared. To do that fungus, culture was prepared as shown in Figure 7.7. Potato dextrose agar media was used as the culture media and fungus spores; from existing fungus samples on a mud

brick, were transformed into the growth media using an inoculating loop and incubated for 7 days. After seven days [124],[62].



(a) Transformation of fungus stock culture

(b) Incubation

(c) Fungus culture after 7 days

Figure 7.7: Fungus culture preparation

A glass environmental chamber was prepared for the test as shown in Figure 7.8. The chamber was filled with wetted fungus growth media; which contains topsoil, compost and coir dust, up to 1 inch from the bottom. Then the chamber was incubated for 24 hrs. After that, the prepared fungus culture was transferred to the growth media in the environmental chamber and incubated for 14 days.



Figure 7.8: Fungus test chamber preparation

The putty surfaces of the test samples were captured before hanging them in the test chamber. After 14 days of incubation period test samples were vertically hung in the chamber according to the ASTM D3273-94 test standards as shown in Figure 7.9. Samples were hanged for four weeks (one month) and after that samples were taken out from the chamber and the putty surfaces were captured. Putty surfaces were compared before and after to identify fungus growth on the putty surface and to measure the fungus growth area. The results of the fungus growth test are shown in Figure 7.10. According to the observations, fungus growth is not observed on any putty surface. Therefore, it can be concluded that sludge putty mixtures do not support fungus growth and it is not affected by selected stabilizers.



Figure 7.9: Fungus growth test of sludge putty mixtures with stabilizers



Figure 7.10: Fungus growth comparison of sludge putty mixtures with stabilizers

7.2.3. Durability variation with stabilizers

Durability can be defined as the safe performance of a structure or a portion of a structure for its expected design life. A protective wall finishing surface that resists surfacing deterioration such as wetting, abrasion and drying is required for the durability of the building. The ability to resist deterioration due to cycles of wetting,

abrasion and drying characteristics of the humid tropics is the main property depicted by the durable walling surfaces. Abrasion in humid tropical climates is mainly a process of erosion by water flow and by rain impact rather than by wind. These factors cause undesirable properties such as loss of strength when saturated with water, erosion due to wind or driving rain leading to poor dimensional stability in building structures. Therefore, it is important to study the wall putty surfaces in the sense of durability; so, the durability of the putty mixtures was tested considering the rain impact on the surface. Accelerated erosion test was conducted [84],[86],[120].

Samples were prepared to conduct the spray erosion test as shown in Figure 7.11. Cement plaster panels were prepared with the dimensions of 1feet×1feet×1inch and putty mixtures were applied on one surface of the panel. Three replicates from each mix were applied and cured until the putty surface was completely dry. After completion of the test, samples were weighed to record the dry weight of the sample before exposure for the spray erosion test.



(a) Cement plaster panels

(b) Putty applied plaster panels

c) Samples weighing before and after the test

Figure 7.11: Sample preparation for spray erosion test of sludge putty mixtures with stabilizers Once the samples were weighted, a spray erosion test was conducted as shown in Figure 7.12 [86]. In the test, pressurized water was sprayed horizontally onto the material surface for one hour (60 min). Samples were placed at a distance of 500 mm away from a spray nozzle and water was sprayed with a pressure of 50kPa. Each sample was exposed to water for one hour with a surface area of 150 mm diameter. After completion of the test, the sample was dried completely and the weight of the dry sample was recorded to calculate the mass loss of the sample to determine the scaled-off factor (SOF).



Figure 7.12: Spray erosion test of sludge putty mixtures with stabilizers

SOF was calculated to determine the number of particles that scaled off from the putty surface when expose to heavy rain conditions [75],[120]. SOF was calculated using equation 7.1; where W1 refers to the weight (g) of the sample before the test, W2 refers to the weight (g) of the sample after the test and A refers to the area (m^2) of the sample.

$$SOF(gm^{-2}) = \frac{W_1 - W_2}{A} = \dots$$
 Equation 7.1

Three replicates of each mix were tested and the average SOF value was calculated for each mix. Results of the test were statistically analyzed (Appendix - V) and the variation is shown in Figure 7.13. According to the test results, mix 1 (sludge putty without stabilizer) has the highest SOF value, and mix 10 (putty mix stabilized with 50% styrene acrylic binder) has the lowest SOF value. All ten mixtures' SOF values are significantly different from each other. All the mixtures mixed with stabilizer have lower SOF values than the sludge putty mix without stabilizer; also, the SOF value increase with the increase of stabilizer proportion in each mixture. When comparing

the three stabilizers, a significant decrease in SOF value with a higher rate is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. However, it can be concluded that stabilizers improve the durability of sludge putty mixtures against rain surfaces and styrene acrylic binder is the most workable stabilizer for that.



*Vertical bars indicate the standard errors of the mean SOF value (n = 3)*Means do not share a same letter are significantly different

Figure 7.13: Spray erosion test results of sludge putty mixtures with stabilizers

7.3. Comparison with Existing Wall Finishes

With the increase in building constructions, the demand for wall finishes is increasing. Wall care putty is the most commonly applied material as the primer layer for both interior and exterior wall surfaces. Wall care putty is applied as a leveling material on the wall plaster and acts as a protective layer for the wall plaster while providing a strong and durable base for the top coat. Wall putty materials are developed with the characteristics of high bonding strength, moisture barrier, high durability and low bio-receptivity. And also the putty materials are developed with the ability to fill dents and hair cracks to level the rough surface of the plaster [108].

Lime was the traditional wall putty material widely applied in past constructions. At present, there are different types of wall putty products available in the market; such as white cement putty, acrylic putty and Plaster of Paris (POP). White cement wall putty is a putty product commercially available in powder form; and it is developed with a mineral, white cement and polymer. The powder is applied after mixing with water. There are different white cement-based wall putty products are available in the market from different commercial brands. Each brand has different properties. Acrylic wall putty is a ready-use putty mix, developed with acrylic polymers and other chemicals. POP is a powder material developed with gypsum and the putty mix is prepared by mixing the powder with water [109]. Talcum, quartz, gypsum and cement are the main raw materials for most commercial wall putty materials and the material should consist of the engineering properties of high bonding strength, good performance of crack resistance, excellent resistance to water, shrinkage as well as flexibility [108].

In this research study, a wall putty was developed from DWTP sludge which is applicable as a primer. And styrene acrylic binder was selected as the stabilizer for the sludge putty. Wall care putty developed from calcium carbonate and lime slurry is the existing wall finishes used for the primer in Sri Lanka. Therefore, the performance of the developed wall putty was tested and compared with existing wall finishes. Therefore, the most commonly used three putty products and lime were selected to test properties along with the developed sludge putty.

7.3.1. Material selection

In the research, commercially available primers were collected from the market. As the research study was conducted at the University of Moratuwa, hardware stores located around the university premises were selected to purchase wall putty products. An area with a 1.5km distance around the university was identified using Google Maps as shown in Figure 7.14; the locations of hardware stores within the area were collected. Therefore, fifteen hardware stores were identified and after contacting those, details were collected regarding available wall putty products and their sales record for one year time period. According to the collected data, nine commercial putty products were identified which are commercially available around the area where the research study is conducted. And the properties of each product were collected from material specifications (Appendix – VI). Those three putty products with the highest sales records were selected for the study as shown in Table 7.4. And lime was also selected for the study as it is a widely used traditional material applied as a primer in Sri Lanka. Selected materials are shown in Figure 7.15.



Figure 7.14: Hardware stores selected to purchase commercially available wall putty products

Putty	Putty form	Raw Materials Used	Water Demand by Volume	Maximum Thickness	Coverage
P1	Powder	White cement, re- dispersible polymer, functional additives	36 - 40 %	1.5 mm (2 Coats)	20 - 25 sqfeets/kg on Single Coat (Varies with Surface Quality)
P2	Powder	White cement, Dolomite	30 - 35%	3 mm (2 Coats)	16 sqfeets/kg on Single Coat (Varies with Surface Quality)
Р3	Powder	CaCO ₃ , white cement	35 - 40%	3 mm	12.5 sqfeets/kg onsingle Coat(Varies with SurfaceQuality)

Table 7.4: Properties of selected commercial putty products

Five mixtures were prepared for the study by mixing materials according to the mix design shown in Table 7.5. Mix 1 was prepared by mixing lime with water. Selected three putty products were mixed with 10% cement by weight and sludge putty mix was prepared by mixing with 25% styrene acrylic binder by weight. The mixtures were mixed in the mechanical mixer. Experiments were conducted to test the bonding strength, bio receptivity and durability of mixtures; and three replicates from each mixture were used for the experiments.



Figure 7.15: Wall finishes selected for the comparison

Mix	Lime (%)	P1 (%)	P2 (%)	P3 (%)	Sludge Putty (%)	Cement (%)	Binder (%)
1	100	-	-	-	-	-	-
2	-	90	-	-	-	10	-
3	-	-	90	-	-	10	-
4	-	-	-	90	-	10	-
5	-	-	-	-	75	-	25

Table 7.5: Mix design for preparation of putty mixtures

7.3.2. Bonding strength comparison of different wall finishes

To test the bonding strength of the mixtures, the lap shear strength test was conducted according to the ASTM D5868-014 test standards [116]. Plaster panels were prepared with the dimensions of 15cm×5cm×2.5cm (Figure 7.16 (a)) and mixtures were applied on one surface with 1mm thickness as shown in Figure 7.16 (b). After drying the samples wall filler was applied on mix2, mix3 and mix4, and sealer was applied on mix5 as the top coat. After completion of sample preparation, a high solid epoxy adhesive was applied on the surface and two samples with the same mix were bonded together as shown in Figure 7.16 (c). Samples were cured and after the adhesive dried completely the lapse shear test was conducted using a universal testing machine (Figure 7.17 (a)). A force is applied to the directions as shown in Figure 7.17 (b), and increased until the bond fails until the materials separate and highest possible joint strength is identified and also the bond breaking pattern was observed. There are four possible crack patterns identified according to the literature as shown in Figure 7.17 (c) [116]. By identifying the bond-breaking crack patterns adhesiveness of the putty mix can be critically analyzed as shown in Table 7.2 [177].



Figure 7.16: Sample preparation for bonding strength comparison of different wall finishes



Figure 7.17: Bonding strength test conducted for comparison of different wall finishes

Three replicates of each mix were tested and the average bonding strength value was calculated. Results of the test were statistically analyzed (Appendix – VII) and the variation is shown in Figure 7.18. According to the test results, mix 1 (lime mix) has the highest strength value and mix 3 (P2 putty mix stabilized with 10% cement) has the lowest strength value. The bonding strength of mix1 and mix2 is not significantly different from each other and significantly different from the other three mixtures. The bonding strength of mix2 and mix5 is not significantly different from each other and significantly different from the other three mixtures. The bonding strength of mix3, mix4 and mix5 is not significantly different from each other and significantly different from the other two mixtures. Therefore, the bonding strength of sludge putty stabilized with 25% styrene acrylic binder by weight is not significantly different from lime. In the test cohesive failure is the only crack pattern observed among the samples. So, there is a high bonding strength between all mixtures and plaster panels. Therefore, all the selected wall finishes were tightly bound to the plaster surface.



Mix 1 – Lime, Mix 2 – P1, Mix 3 – P2, Mix 4 – P3, Mix 5 – Sludge Putty *Vertical bars indicate the standard errors of the mean bonding strength value (n = 3) *Means do not share a same letter are significantly different

Figure 7.18: Bonding strength variation of different wall finishes

7.3.3. Bio receptivity comparison of different wall finishes

Bio receptivity of the mixtures was tested by conducting a fungus growth test according to the ASTM D3273-94 test standards. Cement plaster panels were cast according to the size of 5cmx15cmx2.5cm as shown in Figure 7.19 (a). Mixtures were applied on one surface of each sample with 1mm thickness and cured until completely dry. Wall filler was applied on the samples of mix2, mix3 and mix4 and a sealer was applied on samples of mix5 as the top coat. After the preparation of samples, the surface appearance of each sample was captured before being hung in the fungus test chamber. Samples were hanged in the chamber (Figure 7.19 (b)) for four weeks (one month) duration and after that samples were taken out from the chamber and the surfaces were captured. Surfaces were compared before and after to identify fungus growth on the samples. The results of the fungus growth is not observed on any sample surface as all the selected wall finishes do not support fungus growth.


(a) Sample preparation

(b) Samples hang in fungus test chamber

Figure 7.19: Sample preparation for comparison of fungus growth of different wall finishes



Figure 7.20: Surface appearance of different wall finishes before and after the fungus test

7.3.4. Durability comparison of different wall finishes

The durability of the mixtures was tested considering the rain impact on the material. An accelerated erosion test was conducted and samples were prepared as shown in Figure 7.21. Cement plaster panels were prepared with the dimensions of 1feetx1feetx1inch and mixtures were applied on one surface of the panel. Three replicates from each mix were applied and cured until the putty surface was completely dry. Wall filler was applied on the samples of mix2, mix3 and mix4 and sealer was applied on samples of mix5 as the top coat. After completion of the test, samples were weighed to record the dry weight of the sample before exposure for the spray erosion test. Once the samples were weighted, a spray erosion test was conducted as shown in Figure 7.12. After completion of the test, the sample was dried completely and the weight of the dry sample was recorded. SOF was calculated to determine the number of particles that are scaled off from the surface using equation 7-1.



Figure 7.21: Sample preparation and spray erosion test for comparison of different wall finishes

Three replicates of each mix were tested and the average SOF value was calculated for each mix. Results of the test were statistically analyzed (Appendix – VIII) and the variation is shown in Figure 7.22. According to the test results, mix 1 (lime mix) has the lowest SOF value and mix 3 (P2 putty mix stabilized with 10% cement) has the highest SOF value. SOF of mix2, mix3 and mix4 is not significantly different from each other and significantly different from the other two mixtures. SOF of mix2, Mix4 and mix5 is not significantly different from each other and significantly different from the other two mixtures. SOF of mix1 is significantly different from the other and significantly different from each other and significantly different from each other and significantly different from the other and significantly different from each other and significantly different from the other and significantly different from the other and significantly different from the other and significantly different from each other and significantly different from the other four mixtures. SOF of sludge putty stabilized with 25% styrene acrylic

binder by weight is not significantly different from P2 and P4 commercial putty products and it is significantly different from lime and P3.



Mix 1 – Lime, Mix 2 – P1, Mix 3 – P2, Mix 4 – P3, Mix 5 – Sludge Putty *Vertical bars indicate the standard errors of the mean SOF value (n = 3) *Means do not share a same letter are significantly different

Figure 7.22: SOF variation of different wall finishes

7.4. Real Scale Application

After a series of experiments, a wall putty was developed from DWTP waste sludge. A dry powder was developed from the wet sludge and a workable wall putty mix was developed. The developed wall putty mix was tested under laboratory conditions and acquired results. To analyze the real scale behavior of the developed wall putty mix it was applied on a few building walls including both interior and exterior surfaces as shown in Figure 7.23. Sludge putty was produced in large amounts and provided to skilled laborers and applied to walls following the procedure mentioned in Figure 7.24. During the real-scale application comments and suggestions were collected from skilled laborers for further improvements to the product. And also, the putty surface variation, when exposed to real environmental conditions, was observed for a one-year time duration.



Figure 7.23: Real-scale applications of the sludge putty



Figure 7.24: Sludge putty application procedure

7.4.1. Sludge dry powder production

To apply the putty on a real scale large amount of sludge dry powder was required. So 10kg of sludge dry powder was produced following the dry powder production steps as shown in Figure 7.25.



Figure 7.25: Sludge dry powder production for real-scale application

7.4.2. Wall preparation

The Block wall should be plastered before applying sludge putty. So, the selected block walls were plastered with sand cement plaster as shown in Figure 7.26. Plaster mix was prepared by mixing cement and sand with a ratio of 1:5. Plastered wall was cured for 24 hours to completely dry and sludge putty was applied.



Figure 7.26: Wall preparation for putty application

7.4.3. Material mixing

Putty mix was prepared by mixing sludge dry powder and styrene acrylic binder with a ratio of 3:1 adding water. The natural soil color gradient of the developed powder is the unique property of the product; therefore, styrene acrylic binder was selected as the stabilizer. Styrene acrylic binder is white but colorless when dry, and there is no influence on the natural color of the product. Putty mixtures were prepared by mixing by hand and using a putty mixer as shown in Figure 7.27. According to the skilled labor feedback mixing using a putty mixer is the most effective mixing option.



Mixing with mechanical mixer

Figure 7.27: Mixing the sludge putty mixture

7.4.4. Putty application

The sludge putty mix was applied to the wall plaster as shown in Figure 7.28. For the putty application putty trowel and putty scrapers (Figure 7.29) were used. Sludge putty was applied with a 1mm thickness. After application, the putty surface was cured for two days until completely dry after that the first putty coating can be sanded, and apply the second putty coating where necessary.



Figure 7.28: Sludge putty application on wall plaster



7.4.5. Sanding putty surface

After complete drying of the sludge putty coating, the surface can be sanded with sandpaper as shown in Figure 7.30. Sandpapers with 180 and 200 grit numbers were used in the sanding process. The putty surface should be sanded to obtain a smooth finish surface.



Figure 7.30: Sanding the putty surface with sandpapers

7.4.6. Application of top coat

After sanding the sludge putty surface, a sealer was applied as the finishing top coat as shown in Figure 7.31. The sealer is applied as a protective coating for the sludge putty; which is act as a moisture barrier and increases the durability of the putty surface. And also, the sealer with transparent color was used to preserve the natural color of the sludge putty. The sealer was applied using a sponge to obtain a smooth surface finish avoiding brush marks. The total putty application was done by skilled laborers and at the end of each step, their feedback and comments were recorded. At the end of the process, suggestions were made as shown in Table 7.6



Figure 7.31: Applying sealer on sludge putty surface as a top coat

Putty form	Powder - 300µm
	•
Best mixing option	Mix using a putty mixer
Best mixing ratio	Dry powder 3: Binder 1
	5 I
Coverage	$10-15 \text{ft}^2/\text{kg}$ on single coat(vary with surface quality)
Maximum thickness	1mm
Water demand by volume	35 - 40%
Colour	Natural soil color
Recommended applications	For both interior and exterior walls
······································	

 Table 7.6: Product details developed with skilled labor feedback and observations

7.4.7. Real-scale performance analysis

After the application of the sludge putty in real-scale building walls, three walls were selected to analyze the real-scale performance of the putty. Three walls, which were exposed to different conditions were selected for the study. One interior wall and two exterior walls were selected as shown in Figure 7.32. One of the exterior walls (Figure 7.32 (b)) is completely exposed to the exterior environment and the other one (Figure 7.32 (c)) is under shade.



Figure 7.32: Selected sludge putty applied walls to analyze performances; (a) Interior wall (b) exterior wall (c) Exterior wall with shade

Putty surfaces of the selected walls were observed for one year time period. Data were recorded after analyzing the surface monthly. Color variation, scaling off, fungus growth and other variations were observed monthly. Scale off and fungus growth was not observed for the first one-year time duration after putty application. Photos of the putty surface were taken monthly and compared the images to identify the color variation with the time. Color variation of sludge putty applied on the interior wall is shown in Figure 7.33 and a considerable color deterioration was not observed after eight months. The exterior wall was directly exposed to sunlight, solar radiations and rainy conditions; but scale-off or fungus growth was not observed. Color variation of sludge putty applied on the exterior of sludge putty applied on the exterior sludge putty applied on the exterior wall was directly exposed to sunlight, solar radiations and rainy conditions; but scale-off or fungus growth was not observed. Color variation of sludge putty applied on the exterior wall with shade is shown in Figure 7.35 and a considerable color deterioration for variation of sludge putty applied on the exterior wall with shade is shown in Figure 7.35 and a considerable color deterioration the figure 7.35 and a considerable color deterioration for variation of sludge putty applied on the exterior wall with shade is shown in Figure 7.35 and a considerable color deterioration for fungus putty applied on the exterior wall with shade is shown in Figure 7.35 and a considerable color deterioration.



Figure 7.33: Colour variation of the sludge putty applied to an interior wall



Figure 7.34: Colour variation of sludge putty applied on exterior wall



Figure 7.35: Colour variation of sludge putty applied on exterior wall with shade



- Scale-off or fungus growth was not observed
- Scale-off or fungus growth was not observed
- Scale-off or fungus growth was not observed



7.5. Summary

In the previous chapter, experiments were conducted to develop a dry powdered sludge with putty properties. DWTP wet alum sludge is harder when dry, due to the activation of alum and it is difficult to powder. However, an effective method was developed and optimum conditions were identified to produce dry powdered sludge. The putty properties should be compared with existing wall finishes and the real-scale performances of the developed putty mix should be studied. In this chapter, experiments were conducted to identify the properties including physical properties and durability performances of the developed wall putty and the existing wall finishes, and analyzed the real-scale behavior of the developed wall putty.

As discussed in the previous chapter, a stabilizer should be added to improve the putty properties of the mix. Therefore, as the first step of this chapter experiments were conducted to identify a workable stabilizer and the optimum mixing ratio for the sludge wall putty. To that three stabilizers were selected; cement, lime and styrene acrylic binder, and performances were tested. Ten different putty mixtures were prepared by mixing sludge putty and stabilizers according to the mix design and bonding strength test, fungus growth test and spray erosion tests were conducted. Sludge putty without mixing stabilizer was remain as the controller for all experiments.

To test the bonding strength of the mixtures, a lap shear strength test was conducted according to the ASTM D5868-014 test standards. All the mixtures mixed with stabilizer have a higher bonding strength value than the sludge putty mix without stabilizer; also, the bonding strength value increase with the increase of stabilizer proportion in each mixture. When comparing the three stabilizers, a significant increase in bonding strength value is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. According to the observations fiber-tear failure, cohesive failure and adhesive failure are observed within the samples and stoke-break failure is not observed. Therefore, the putty mixtures observed cohesive failure is tightly bound to the plaster panel and the most workable stabilizers are included also the sludge putty mix do not tightly bound to the plaster panel without a stabilizer. All the putty mixtures prepared by mixing styrene acrylic binder are tightly

bound to the plaster panel and other mixtures are tightly bound with the highest percentages only (cement 15% and lime 50%). Therefore, cement and lime should be added in high amounts to increase the bonding strength of the putty mixture while the bonding strength can be increased with a lower amount of styrene acrylic binder; styrene acrylic binder is the most workable stabilizer for sludge putty compared to cement and lime.

Bio receptivity of putty mixtures was tested by conducting a fungus growth test according to the ASTM D3273-94 test standards. According to the observations, fungus growth is not observed on any putty surface. Therefore, it can be concluded that sludge putty mixtures do not support fungus growth and it is not affected by selected stabilizers. The durability of the putty mixtures was tested considering the rain impact on the surface. An accelerated erosion test was conducted. All the mixtures mixed with stabilizer have lower SOF values than the sludge putty mix without stabilizer; also, the SOF value increase with the increase of stabilizer proportion in each mixture. When comparing the three stabilizers, a significant decrease in SOF value with a higher rate is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. However, it can be concluded that stabilizers improve the durability of sludge putty mixtures against rain surfaces and styrene acrylic binder is the most workable stabilizer for that.

The next experiments of this chapter were conducted to compare the properties of the sludge putty mix with existing wall finishes. The most commonly used three putty products and lime was selected to test properties along with the developed sludge putty. Five mixtures were prepared according to the mix design and bonding strength test, fungus growth test and spray erosion tests were conducted. According to the test results, the bonding strength of sludge putty stabilized with 25% styrene acrylic binder by weight is not significantly different from commercial putty products and it is significantly different from lime. In the test cohesive failure is the only crack pattern observed among the samples. So, there is a high bonding strength between all mixtures and plaster panels. Therefore, all the selected wall finishes were tightly bound to the plaster surface. According to the observations of the fungus growth test, there is no

fungus growth observed on any sample surface as all the selected wall finishes do not support fungus growth. According to the spray erosion test results, SOF of sludge putty stabilized with 25% styrene acrylic binder by weight is not significantly different from P2 and P4 commercial putty products and it is significantly different from lime and P3.

Up to the previous section, the developed wall putty mix was tested under laboratory conditions only and acquired results. Therefore, the next step of this chapter was conducted to analyze the real-scale behavior of the developed wall putty mix. To that, the sludge putty mix was applied on a few building walls including both interior and exterior surfaces with skilled labor. After the application of sludge, putty feedback was taken from skilled laborers. According to the feedback 300µm is identified as the effective fineness for the sludge putty powder and dry powder 3: binder 1 is the best mixing ratio for the putty mixture. And also, it was found that there is no considerable color deterioration, scale off or fungus growth observed in the sludge putty for one-year time duration and the putty can be recommended for both interior walls and exterior walls. All the findings of this research will be concluded in the next chapter including recommendations for further research studies.

CHAPTER 08: CONCLUSIONS AND RECOMMENDATIONS

8.1. Conclusions and Recommendations

- This research was conducted with the main aim of the development of wall putty for the tropics using DWTP waste alum sludge. As the first step DWTP was selected to collect sludge samples for the study. DWTPs practice different sludge removal processes and removing sludge has different properties identical to the sludge removal process. Sludge samples were collected from Ambathale, Biyagama and Kandana DWTPs; as each plant practice different sludge removal methods.
- Laboratory experiments were conducted to identify the properties of each sludge sample to select a DWTP to collect sludge for the study. According to the test results, the Ambathale DWTP sludge sample has the highest moisture content and the lowest solid content. Biyagama DWTP sludge; which was collected from the sludge treatment plant of the Biyagama DWTP, has the lowest moisture content, highest solid content and lowest volumetric shrinkage values; and Biyagama DWTP was selected to collect sludge for the study.
- The chemical composition of Biyagama DWTP sludge was identified by conducting an XRF analysis and according to the analysis, Si (42.23%) is identified as the main chemical element includes in DWTP sludge.
- According to the XRD analysis test results, DWTP sludge chemical composition consists of kaolinite, quartz, calcite and ettringite. And also, the chemical composition is significant to the findings in the literature.
- According to the heavy metal analysis test results Cadmium (Cd) (<0.1ppm), Lead (Pb) (4.3ppm), Mercury (Hg) (<0.1ppm) and Arsenic (As) (0.9ppm) were identified in the sludge sample as heavy metals; but the available quantities are lower than the maximum permissible level define for wall top coatings in Sri Lanka. Therefore, there is no harmful impact on human health or the environment when developing a wall putty with the sludge

- Laboratory tests were conducted to identify the shrinkage and moisture content
 of DWTP sludge. Volumetric shrinkage of DWTP sludge varies between the
 values of 59.82% and 93.62% with an average of 83.91%. Observational
 shrinkage is not affected by the applied area and the DWTP sludge layer is
 completely cracked and detached from the surface. DWTP sludge moisture
 content varies between the values of 62.31% and 84.72% with an average of
 76.26%. Humidity is the significant climatic factor that affects on shrinkage of
 DWTP sludge with a positive correlation.
- Wall putty cannot be developed with DWTP wet sludge according to the high shrinkage. Therefore, experiments were conducted to optimize the putty properties of wet sludge. To develop wall putty using wet sludge, adhesiveness should be increased while reducing the drying shrinkage of the mix and additives can be used to improve the putty properties of wet sludge.
- The addition of sand reduced the volumetric shrinkage and the shrinkage becomes zero at 70% sand percentage, but it is not practically applicable in developing a putty mixture. According to the ball drop test results, it was identified that wet sludge mixtures with a sand percentage of between 30% to 60% have optimum clay and sand contents to develop wall putty mix. Although the mix has optimum shrinkage performances, the adhesiveness of the mix should be developed.
- Three natural fibers; banana fiber, coir fiber and sugarcane fiber, were selected for the study, and mixtures were prepared by mixing fiber with wet sludge and a volumetric shrinkage test was conducted. According to the test results addition of fiber reduced the volumetric shrinkage and banana fiber is the most effective natural fiber type to reduce volumetric shrinkage of wet sludge compared to coir fiber and sugarcane fiber.
- The effectiveness of binders was tested. Three plant-based natural binders and a synthetic binder were selected. Crack patterns of wet sludge mixture vary with the binder type and the moisture content of the mix increased with the

addition of binders. But fungus growth in the mix increase with natural binders. Therefore, natural binders are not effective as an additive to develop wall putty with DWTP sludge. Addition of synthesized binder; styrene acrylic binder reduces volumetric shrinkage and observational shrinkage without facilitating fungus growth.

- Physical additives reduced drying shrinkage while binders increase the adhesiveness of wet sludge mixtures. According to that the most effective method to enhance the putty properties of wet sludge is adding both synthetic binder and physical additives. The optimum wet sludge mix was identified as 57% wet sludge, 30% sand, 3% sugarcane fiber and 10% styrene acrylic binder. The wet sludge mixture consists of low volumetric shrinkage, low drying cracks and high adhesiveness; but the mix has few defects as the presence of small surface cracks and fibers pop up from the surface when apply as a wall putty.
- DWTP wet alum sludge is harder when dry, due to the activation of alum and it is difficult to powder. As wall putty cannot be developed with wet sludge, experiments were conducted to develop a process to produce dry powdered sludge. Therefore, the thermal properties of DWTP sludge were studied. The Colour of the sludge is variants with the temperature, and from 100°C to 400°C the color becomes darker in both dry and wet conditions. Beyond 400°C color become lighter up to 700°C and become constant up to 1000°C.
- The highest density was achieved at the temperature of 900°C and the lowest density was achieved at 500°C. Density values of the heated sludge samples are gradually reduced from 100°C up to 500°C and beyond the 500°C, density again gradually increases up to 900°C temperature. Density increased when applying heat to sludge and the lowest density was recorded at 500°C temperature.
- To overcome alum activation coconut oil was selected as the lubricant; Coconut oil has the potential to overcome alum activation without effect on the color of the sample.

- The density values of heated sludge samples with oil are significantly lower than the density values of heated sludge samples without oil. The lowest density was recorded at 700°C temperature and the optimum coconut oil mixing percentage was identified as 8% by weight, for a productive dry powder production process.
- Drying cracks were observed on the samples burned at temperatures from 200°C to 600°C, and no drying cracks were observed on the samples burned at 700°C and 800°C. Therefore, the dry sludge powder sample burned at 700°C temperature was the optimum dry sludge powder sample to develop wall putty.
- A stabilizer was added to improve the putty properties of the dry powdered sludge mix. A lap shear strength test was conducted according to the ASTM D5868-014 test standards Bonding strength increase with the increase of stabilizer proportion in each mixture. When comparing the three stabilizers, a significant increase in bonding strength value is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. All the putty mixtures prepared by mixing styrene acrylic binder are tightly bound to the plaster panel and other mixtures are tightly bound with the highest percentages only (cement 15% and lime 50%). Therefore, cement and lime should be added in high amounts to increase the bonding strength of the putty mixture while the bonding strength can be increased with a lower amount of styrene acrylic binder; styrene acrylic binder is the most workable stabilizer for sludge putty compared to cement and lime.
- Bio receptivity of putty mixtures was tested by conducting a fungus growth test according to the ASTM D3273-94 test standards. According to the observations, fungus growth is not observed on any putty surface. Therefore, it can be concluded that sludge putty mixtures do not supportive of fungus growth and it is not affected by selected stabilizers. The durability of the putty mixtures was tested considering the rain impact on the surface. An accelerated erosion test was conducted. All the mixtures mixed with stabilizer have lower SOF values than the sludge putty mix without stabilizer; also, the SOF value

increase with the increase of stabilizer proportion in each mixture. When comparing the three stabilizers, a significant decrease in SOF value with a higher rate is observed in the mixtures stabilized with styrene acrylic binder compared to other mixtures. However, it can be concluded that stabilizers improve the durability of sludge putty mixtures against rain surfaces and styrene acrylic binder is the most workable stabilizer for that.

- Experiments were conducted to compare the properties of the sludge putty mix with existing wall finishes. The most commonly used three putty products and lime was selected to test properties along with the developed sludge putty. Five mixtures were prepared and bonding strength tests, fungus growth tests and spray erosion tests were conducted. According to the test results, the bonding strength of sludge putty stabilized with 25% styrene acrylic binder by weight is not significantly different from commercial putty products and it is significantly different from lime. In the test cohesive failure is the only crack pattern observed among the samples. So, there is a high bonding strength between all mixtures and plaster panels.
- According to the observations of the fungus growth test, there is no fungus growth observed on any sample surface as all the selected wall finishes do not supportive of fungus growth. According to the spray erosion test results, SOF of sludge putty stabilized with 25% styrene acrylic binder by weight is not significantly different from P2 and P4 commercial putty products and it is significantly different from lime and P3.
- After testing the developed sludge putty under laboratory conditions, the putty mix was applied on a few building walls including both interior and exterior surfaces with skilled labor to analyze real-scale behavior. After the application of sludge, putty feedback was taken from skilled laborers. According to the feedback 300µm is identified as the effective fineness for the sludge putty powder and dry powder 3: binder 1 is the best mixing ratio for the putty mixture. And also, it was found that there is no considerable color deterioration, scale, or fungus growth observed in the sludge putty for one-year

time duration and the putty can be recommended for both interior walls and exterior walls.

• Finally, a dry powder was developed from dewatered alum sludge collected from Biyagama DWTP. Wall putty mix can be developed by mixing the dry powder with a styrene acrylic binder. The sludge putty is applicable on both exterior and interior wall surfaces and a sealer is applied on the putty surface as a protective coating.

8.2. Future Research

- Development of a standard for soil-base wall finishes
- Lifecycle costing of newly invented sludge putty compared to existing wall finishes
- Development of a natural binder to use as the stabilizer for the sludge putty mix
- Development of a natural polymer as a protective top coating for the sludge putty mix.
- Investigation of possible technologies to develop white color sludge putty powder.
- Analyzing the potential of alternative applications of dry powdered alum sludge.
- Investigating the potential to develop, a useful dry powder from other industrial wastes.

References

- Ministry of Water Supply and Drainage, "National drinking water policy," Sri Lanka, 2012.
- T. Ahmad, K. Ahmad, and M. Alam, "Sludge quantification at water treatment plant and its management scenario," *Environ. Monit. Assess.*, vol. 2, no. 189, p. 453, 2017, doi: 10.1007/s10661-017-6166-1.
- G. R. Xu, J. L. Zou, and G. B. Li, "Ceramsite obtained from water and wastewater sludge and its characteristics affected by (Fe2O3+CaO+MgO)/ (SiO2+Al2O3)," *Water Res.*, vol. 43, no. 11, pp. 2885–2893, 2009, doi: 10.1016/j.watres.2009.03.046.
- [4] K. Anjithan, "Management practices of water treatment sludge in Sri Lanka," University of Moratuwa, 2015.
- [5] National Water Supply and Drainage Board, "Sludge Management Policy for Water Treatment Plants," Sri Lanka, 2012.
- [6] H. H. Galkanda and R. U. Halwatura, "Analysis the Performance of Chemical and Physical Additives to Reduce Shrinkage of Drinking Water Treatment Sludge (DWTS)," *Int. J. Environ. Sci. Dev.*, vol. 11, no. 4, pp. 175–179, 2020, doi: 10.18178/ijesd.2020.11.4.1247.
- [7] A. Ruuska and T. Häkkinen, "Material Efficiency of Building Construction," *Buildings*, vol. 4, no. 3, pp. 266–294, 2014, doi: 10.3390/buildings4030266.
- [8] S. Hermann, D. Arent, P. Steduto, and R. S. J. Tol, "Considering the energy, water and food nexus: Towards an integrated modelling approach," *Energy Policy*, 2011, doi: 10.1016/j.enpol.2011.09.039.
- [9] Y. Gong and D. Song, "Life cycle building carbon emissions assessment and driving factors decomposition analysis based on LMDI-A case study of Wuhan city in China," *Sustain.*, vol. 7, no. 12, pp. 16670–16686, 2015, doi: 10.3390/su71215838.
- [10] J. Park, J. Yoon, and K. H. Kim, "Critical review of the material criteria of building sustainability assessment tools," *Sustain.*, vol. 9, no. 2, 2017, doi: 10.3390/su9020186.
- [11] A. A. Sofi, T. A. Sheikh, R. A. Wani, and A. Manzoor, "Cement stabilized earth

blocks (CSEB): An economic and eco- friendly building material .," *IOSR J. Mech. Civ. Eng.*, vol. 13, no. 6, pp. 6–11, 2016, doi: 10.9790/1684-1306050611.

- C. Udawattha, F. R. Arooz, and R. U. Halwatura, "New earth walling material: Integrating modern technology onto ancient mud wall," *7th Int. Conf. Sustain. Built Environ.*, vol. 7, no. March 2017, pp. 24–31, 2016.
- [13] M. I. Gomes, P. Faria, and T. D. Gonçalves, "Earth-based mortars for repair and protection of rammed earth walls. Stabilization with mineral binders and fibers," *J. Clean. Prod.*, vol. 172, no. November, 2017, doi: 10.1016/j.jclepro.2017.11.170.
- [14] J. Krantz, J. Larsson, W. Lu, and T. Olofsson, "Assessing Embodied Energy and Greenhouse Gas Emissions in Infrastructure Projects," *Buildings*, vol. 5, no. 4, pp. 1156–1170, 2015, doi: 10.3390/buildings5041156.
- [15] E. Quagliarini, A. Stazi, E. Pasqualini, and E. Fratalocchi, "Cob construction in Italy: Some lessons from the past," *Sustainability*, vol. 2, no. 10, pp. 3291–3308, 2010, doi: 10.3390/su2103291.
- [16] M. S. Lekshmi, S. Vishnudas, and D. G. N. Nair, "Experimental Study on the Physical Properties of Mud Mortar in Comparison with the Conventional Mortars," *Civ. Eng. Urban Plan. An Int. J.*, vol. 3, no. 2, pp. 127–135, 2016, doi: 10.5121/civej.2016.3211.
- [17] M. Emiroğlu, A. Yalama, and Y. Erdoğdu, "Performance of ready-mixed clay plasters produced with different clay/sand ratios," *Appl. Clay Sci.*, vol. 115, pp. 221–229, 2015, doi: 10.1016/j.clay.2015.08.005.
- [18] J.K. Cement Limited, "Growth with resilience Sustainable Report 2015-16," Uttara Pradesh, India, 2016.
- [19] GMP+ International B.V., "Risk assessment: Calcium carbonate," The Netherlands, 2015.
- [20] P. Janiszewski, "How long can we survive without food or water?," 2015.
- [21] A. H. Sequeira, R. Malik, P. Pandey, R. Chandra, and P. Baishya, "Study on Drinking Water Habits of Residents of a Campus: A Case Study.," SSRN Electron. J., vol. May, pp. 4–28, 2014, doi: 10.2139/ssrn.2392765.
- [22] B. C. O. Kelly, "Geotechnical properties of a municipal water treatment sludge incorporating a coagulant," *Can. Geotech. J.*, vol. 45, pp. 715–725, 2008, doi:

10.1139/T07-109.

- [23] S. De Carvalho, J. L. Zhou, W. Li, and G. Long, "Progress in manufacture and properties of construction materials incorporating water treatment sludge : A review," *Resour. Conserv. Recycl.*, vol. 145, pp. 148–159, 2019, doi: 10.1016/j.resconrec.2019.02.032.
- [24] H. Moayedi, B. Huat, A. ASADI, Z. Salleh, and M. Moghaddas, "Surface Water Treatment Process; A Review on Various Methods," *Electron. J. Geotech. Eng.*, vol. 16, 2011.
- [25] A. I. Zouboulis and I. A. Katsoyiannis, "Recent Advances in Water and Wastewater Treatment with Emphasis in Membrane Treatment Operations," *Water*, vol. 11, no. 45, pp. 1–6, 2019, doi: 10.3390/w11010045.
- [26] M. Farhaoui and M. Derraz, "Review on Optimization of Drinking Water Treatment Process," J. Water Resour. Prot., vol. 8, pp. 777–786, 2016, doi: 10.4236/jwarp.2016.88063.
- [27] A. M. Hidalgo, M. D. Murcia, M. G'omez, and E. G'omez, "Possible Uses for Sludge from Drinking Water Treatment Plants," *J. Environ. Eng.*, vol. 143, no. 04016088, pp. 1–7, 2016, doi: 10.1061/(ASCE)EE.1943-7870.0001176.
- [28] K. M. Hassan, K. Fukushi, K. Turikuzzaman, and S. M. Moniruzzaman, "Effects of using arsenic – iron sludge wastes in brick making," *Waste Manag.*, vol. 34, no. 6, pp. 1072–1078, 2014, doi: 10.1016/j.wasman.2013.09.022.
- [29] S. R. Teixeira, G. T. A. Santos, A. E. Souza, P. Alessio, S. A. Souza, and N. R. Souza, "The effect of incorporation of a Brazilian water treatment plant sludge on the properties of ceramic materials," *Appl. Clay Sci.*, vol. 53, no. 4, pp. 561–565, 2011, doi: 10.1016/j.clay.2011.05.004.
- [30] S. E. Hagemann, A. L. G. Gastaldini, M. Cocco, S. L. Jahn, and L. M. Terra, "Synergic effects of the substitution of Portland cement for water treatment plant sludge ash and ground limestone : Technical and economic evaluation," *J. Clean. Prod.*, vol. 214, pp. 916–926, 2019, doi: 10.1016/j.jclepro.2018.12.324.
- [31] S. De Carvalho Gomes, J. L. Zhou, W. Li, and G. Long, "Progress in manufacture and properties of construction materials incorporating water treatment sludge: A review," *Resour. Conserv. Recycl.*, vol. 145, no. February, pp. 148–159, 2019, doi: 10.1016/j.resconrec.2019.02.032.

- [32] C. Huang, J. R. Pan, and Y. Liu, "Mixing Water Treatment Residual with Excavation Waste Soil in Brick and Artificial Aggregate Making," *J. Environ. Eng.*, vol. 131, no. February, pp. 272–277, 2005.
- [33] National Water Supply and Drainage Board Sri Lanka, "National Water Supply and Drainage Board - Annual Report 2019," 2019.
- [34] S. Sumanaweera, "Kelani Right Bank Water Treatment Plant Sri Lanka," 2015.[Online]. Available: http://www.jwrc-net.or.jp/aswin/en/newtap
- [35] A. Sales, F. Rodrigues, D. Souza, and C. Rosa, "Mechanical properties of concrete produced with a composite of water treatment sludge and sawdust," *Constr. Build. Mater.*, vol. 25, no. 6, pp. 2793–2798, 2011, doi: 10.1016/j.conbuildmat.2010.12.057.
- [36] T. Ahmad, K. Ahmad, and M. Alam, "Sustainable management of water treatment sludge through 3'R' concept," J. Clean. Prod., vol. 124, pp. 1–13, 2016, doi: https://doi.org/10.1016/j.jclepro.2016.02.073.
- [37] J. Komlos, A. Welker, M. Asce, V. Punzi, R. Traver, and M. Asce, "Feasibility Study of As-Received and Modified (Dried / Baked) Water Treatment Plant Residuals for Use in Storm-Water Control Measures," *J. Environ. Eng.*, vol. 139, no. 10, pp. 1237–1245, 2013, doi: 10.1061/(ASCE)EE.1943-7870.0000737.
- [38] B. C. O'Kelly and M. E. Quille, "Compressibility and consolidation of water treatment residues," in *Proceedings of the Institution of Civil Engineers Waste* and Resource Management, 2009, no. 162, pp. 85–97. doi: 10.1680/warm.2009.162.2.85.
- [39] M. C. Wang, J. Q. Hull, M. Jao, B. A. Dempsey, and D. A. Cornwell, "Engineering behavior of water treatment sludge," *J. Environ. Eng.*, vol. 118, no. 6, pp. 848–864, 1992.
- [40] B. M. C. Wang, J. Q. Hull, M. Jao, B. A. Dempsey, and D. A. Cornwell, "Engineering behavior of w at er treatment sludge," *J. Environ. Eng.*, vol. 118, pp. 848–864, 1992.
- [41] A. M. Hidalgo, M. D. Murcia, M. Gomez, E. Gomez, C. Garcia-Izquierdo, and C. Solano, "Possible Uses for Sludge from Drinking Water Treatment Plants," *J. Environ. Eng.*, pp. 1–7, 2016, doi: 10.1061/(ASCE)EE.1943-7870.0001176.

- Y.-E. Lee, I.-T. Kim, and Y.-S. Yoo, "Stabilization of High-Organic-Content Water Treatment Sludge by Pyrolysis," *Energies*, vol. 11, no. 3929, pp. 1–14, 2018, doi: 10.3390/en11123292.
- [43] E. Nimwinya, W. Arjharn, and S. Horpibulsuk, "A sustainable calcined water treatment sludge and rice husk ash geopolymer," *J. Clean. Prod.*, vol. 119, pp. 128–134, 2016, doi: 10.1016/j.jclepro.2016.01.060.
- [44] S. Horpibulsuk, C. Suksiripattanapong, W. Samingthong, R. Rachan, and A. Arulrajah, "Durability against Wetting Drying Cycles of Water Treatment Sludge Fly Ash Geopolymer and Water Treatment Sludge Cement and Silty Clay Cement Systems," *J. Mater. Civ. Eng.*, pp. 1–9, 2015, doi: 10.1061/(ASCE)MT.1943-5533.0001351.
- [45] P. Verlicchi and L. Masotti, "reuse of drinking water treatment plants sludges in agriculture: problems, perspectives and limitations," in *Technology transfer*. *Proceedings of the 9th International Conference on the FAO ESCORENA Network on recycling of agricultural, municipal and industrial residues in agriculture, Gargano, Italy*, 2001, pp. 67–73.
- [46] K. Chiang, P. Chou, C. Hua, K. Chien, and C. Cheeseman, "Lightweight bricks manufactured from water treatment sludge and rice husks," *J. Hazard. Mater. J.*, vol. 171, pp. 76–82, 2009, doi: 10.1016/j.jhazmat.2009.05.144.
- [47] K. Lin and C. Lin, "Hydration characteristics of waste sludge ash utilized as raw cement material," *Cem. Concr. Res.*, vol. 35, no. 2005, pp. 1999–2007, 2005, doi: 10.1016/j.cemconres.2005.06.008.
- [48] Y. Lee, S. Lo, J. Kuo, and C.-C. Tsai, "Beneficial Uses of Sludge from Water Purification Plants in Concrete Mix," *Environ. Eng. Sci.*, vol. 29, no. 4, pp. 284– 289, 2012, doi: 10.1089/ees.2010.0479.
- [49] R. M. R. Zamora, F. E. Ayala, L. C. Garcia, A. D. Moreno, and R. Schouwenaars, "Optimization of the preparation conditions of ceramic products using drinking water treatment sludges," *J. ofEnvironmental Sci. Heal. Part A Toxic/Hazardous Subst. Environ. Eng.*, vol. 43, pp. 1562–1568, 2008, doi: 10.1080/10934520802293750.
- [50] L. V. Cremades, J. A. Cusido, and F. Arteaga, "Recycling of sludge from drinking water treatment as ceramic material for the manufacture of tiles," *J.*

Clean. Prod., vol. 201, pp. 1071–1080, 2018, doi: 10.1016/j.jclepro.2018.08.094.

- [51] A. O. Babatunde and Y. Q. Zhao, "Constructive Approaches Toward Water Treatment Works Sludge Management : An International Review of Beneficial Reuses," *Crit. Rev. Environ. Sci. Technol.*, vol. 37, no. 2, pp. 129–164, 2007, doi: 10.1080/10643380600776239.
- [52] F. R. Arooz and R. U. Halwatura, "Mud-concrete block (MCB): mix design & amp; durability characteristics," *Case Stud. Constr. Mater.*, vol. 8, no. December 2017, pp. 39–50, 2018, doi: 10.1016/j.cscm.2017.12.004.
- [53] C. Udawattha and R. Halwatura, "Embodied energy of mud concrete block (MCB) versus brick and cement blocks," *Energy Build.*, vol. 126, pp. 28–35, 2016, doi: 10.1016/j.enbuild.2016.04.059.
- [54] K. G. Ramirez, E. Possan, B. G. dos Santos Dezen, and M. Colombo, "Potential uses of waste sludge in concrete production," *Manag. Environ. Qual. An Int. Journa*, vol. 28, no. 6, pp. 821–838, 2017, doi: 10.1108/MEQ-09-2015-0178.
- [55] P. Ponkarthikeyan, R. Ganesh, S. F. A, and A. Clay, "Experimental Study on Bricks Using Water Treatment Sludge," *Int. J. Res. Appl. Sci. Eng. Technol.*, vol. 4, no. Xi, pp. 485–493, 2016.
- [56] E. M. M. Ewais, R. M. Elsaadany, A. A. Ahmed, and N. H. Shalaby, "INSULATING REFRACTORY BRICKS FROM WATER TREATMENT SLUDGE AND RICE HUSK ASH," *Refract. Ind. Ceram.*, vol. 58, no. 2, pp. 136–144, 2017, doi: 10.1007/s11148-017-0071-6.
- [57] E. M. Silva, D. M. Morita, A. C. M. Lima, and L. G. Teixeira, "Manufacturing ceramic bricks with polyaluminum chloride (PAC) sludge from a water treatment plant," *Water Sci. Technol.*, vol. 71, no. 11, pp. 1638–1645, 2015, doi: 10.2166/wst.2015.132.
- [58] M. Dahhou, M. El, M. Muhammad, and A. Arshad, "Synthesis and characterization of drinking water treatment plant sludge - incorporated Portland cement," *J. Mater. Cycles Waste Manag.*, 2017, doi: 10.1007/s10163-017-0650-0.
- [59] M. Alqam, A. Jamrah, and H. Daghlas, "Utilization of Cement Incorporated with Water Treatment Sludge," *Jordan J. Civ. Eng.*, vol. 5, no. 2, pp. 268–277,

2011.

- [60] J. Byrne, "G REEN B UILDINGS IN D ELAWARE :," no. September, 2008.
- [61] C. Udawattha and R. Halwatura, "Life cycle cost of different Walling material used for affordable housing in tropics," *Case Stud. Constr. Mater.*, vol. 7, no. April, pp. 15–29, 2017, doi: 10.1016/j.cscm.2017.04.005.
- [62] C. Udawattha, H. Galkanda, I. S. Ariyarathne, G. Y. Jayasinghe, and R. Halwatura, "Mold growth and moss growth on tropical walls," *Build. Environ.*, vol. 137, no. January, pp. 268–279, 2018, doi: 10.1016/j.buildenv.2018.04.018.
- [63] R. Dissanayake, "GREEN BUILDINGS FOR SUSTAINABLE BUILT ENVIRONMENT IN SRI LANKA," no. Id, pp. 1–7, 2015.
- [64] M. Rana, S. Azad, A. H. Akhi, G. Sharfaraz, and I. Azad, "Effect of Environmental Impact in Building Constructions," vol. 5, no. 6, pp. 339–343, 2017, doi: 10.11648/j.ajce.20170506.14.
- [65] U. A. Umar, D. Adaptive, M. Customization, and F. Building, "sustainable building material for green building construction, sustainable building material for green building construction, conservation and refurbishing," no. December, 2012.
- [66] R. Rahardjati, M. F. Khamidi, and A. Idrus, "Green building rating system : The need of material resources criteria in green building assessment," in 2011 2nd International Conference on Environmental Science and Technology, 2011, no. February 2011.
- [67] W. Fei *et al.*, "The critical role of the construction industry in achieving the sustainable development goals (Sdgs): Delivering projects for the common good," *Sustainability*, vol. 13, no. 9112, 2021, doi: 10.3390/su13169112.
- [68] S. Goubran, "On the Role of Construction in Achieving the SDGs," J. Sustain. Res., vol. 1, no. 2, 2019, doi: 10.20900/jsr20190020.
- [69] E. M. Elias and C. Khai, "The empirical study of green buildings (residential) implementation : perspective of house developers," *Procedia Environ. Sci.*, vol. 28, no. SustaiN 2014, pp. 708–716, 2015, doi: 10.1016/j.proenv.2015.07.083.
- [70] J. Zuo and Z. Zhao, "Green building research current status and future agenda : A review Green building research – current status and future agenda : A review," *Renew. Sustain. Energy Rev.*, vol. 30, no. February 2014, pp. 271–

281, 2018, doi: 10.1016/j.rser.2013.10.021.

- [71] Brick Industry Association, "Sustainability and Brick, Technical Notes on Brick Construction," *Brick Ind. Assoc.*, vol. 48, no. November, pp. 1–10, 2015.
- [72] S. A. A. Gunawardana, S. A. D. S. S. Maheepala, and G. Y. Jayasinghe, "Current Status, Requirements and Opportunities for Green Building in Sri Lanka," *Int. For. Environ. Symp. 2017*, no. 73, p. 2017, 2017.
- [73] B. A. W. P. Bombugala and A. Atputharajah, "Sustainable development through green building concept," in *International Conference on Sustainable Built Environment-2010*, 2010, no. December 2010, pp. 13–14.
- [74] K. G. A. S. Waidyasekara and W. N. J. K. Fernando, "benefits of adopting green concept for," 2013.
- [75] F. R. Arooz and R. U. Halwatura, "Case Studies in Construction Materials Mudconcrete block (MCB): mix design & durability characteristics," *Case Stud. Constr. Mater.*, vol. 8, no. November 2017, pp. 39–50, 2018, doi: 10.1016/j.cscm.2017.12.004.
- [76] C. Jayasinghe, "Embodied energy of alternative building materials and their impact on life cycle cost parameters," 2011.
- [77] J. M. Crow, "The concrete conundrum," *Chem. World*, no. March, pp. 62–66, 2008.
- [78] World Business Council for Sustainable Development, "The cement sustainability initiative progress report 2005," Corp. Soc. Responsib. Environ. Manag., vol. 11, no. 2, 2005, doi: 10.1002/csr.59.
- [79] C. Meyer, "Concrete Materials and Sustainable Development in the USA," *Struct. Eng. Int.*, vol. 14, no. 3, pp. 203–207, 2004, doi: 10.2749/101686604777963757.
- [80] C. Udawattha, F. R. Arooz, and R. U. Halwatura, "Energy content of walling materials- A comparison of Mud-Concrete Blocks, Bricks, Cabook and Cement Blocks on tropics," *7th Int. Conf. Sustain. Built Environ.*, vol. 7, no. December, pp. 30–42, 2016.
- [81] N. H. V. T. N. Nanayakkara, C. Udawattha, and R. Halwatura, "Investigation on Elements and their Fraction of Building Construction Cost," *Moratuwa Eng. Res. Conf.*, pp. 277–282, 2017.

- [82] R. U. Halwatura and N. M. Nishad, "Effects of greenery on city comfort in different micro climatic conditions," 2013.
- [83] British Standard Institution, "Durability of buildings and building elements, products and components," London, 1992.
- [84] C. D. Udawattha, G. A. H. H. Galkanda, and R. U. Halwatura, "A Study on Natural Rain Surface Erosion of Different Walling Materials in Tropics," 2018 Moratuwa Eng. Res. Conf., no. i, pp. 84–89, 2018, doi: 10.1109/MERCon.2018.8421938.
- [85] H. Galkanda, C. Udawattha, and R. Halwatura, "Improve intrinsic properties of walling materials to create occupant comfort," in "Sustainability for people envisaging multi disciplinary solution": Proceedings of the 11th International Conference of Faculty of Architecture Research Unit (FARU), 2018, pp. 66–73.
- [86] S. A. A. Gunawardana, H. H. Galkanda, R. U. Halwatura, and G. Y. Jayasinghe, "Investigation of rain surface erosion and bonding strength of different wall care putty materials along with different walling materials," *J. Build. Eng.*, vol. 34, no. September 2020, p. 101872, 2021, doi: 10.1016/j.jobe.2020.101872.
- [87] A. N. Askar and L. P. Raut, "International journal of engineering sciences & research technology design of automatic wall plastering machine," *Int. J. Eng. Sci. Res. Technol. Des.*, vol. 6, no. 3, pp. 543–555, 2017.
- [88] M. M. Saleh, "characterization of qarh's wall plasters, al-ulla, saudi arabia. A case study," *Int. J. Conserv. Sci.*, vol. 4, no. 1, pp. 65–80, 2013.
- [89] N. Zainudeen and J. Jeyamathn, "Cement and its effect to the environment : A case study in SriLanka," *Dep. Build. Econ. Univ. Moratuwa*, pp. 1408–1416, 2004.
- [90] K. S. Devi, V. V Lakshmi, and A. Alakanandana, "Impacts of Cement Industry on Environment - An Overview impacts of cement industry on environment – an overview," *Asia pacific J. Res.*, vol. 1, no. 157, 2018.
- [91] M. Stajanca and A. Estokova, "Environmental Impacts of Cement Production," *Tech. Univ. Kosice, Civ. Eng. Fac. Inst. Archit. Eng.*, pp. 296–302, 2012.
- [92] R. U. K. Piyadasa, "River sand mining and associated environmental problems in Sri Lanka," in *River sand mining and associated environmental problems in Sri Lanka*, 2011, no. September 2009, pp. 148–153.

- [93] M. P. Gray, "Cave Art and the Evolution of the Human Mind," Victoria University of Wellington, 2010.
- [94] R. J. Forbes, *Studies in Ancient Technology Volume I.* Leiden, The Netherlands: E.J. Brill, 1955.
- [95] M. Singh, S. Vinodh Kumar, and S. Waghmare, "Mud Plaster Wall Paintings of Bhaja Caves: Composition and Performance Characteristics," *Indian J. Hist. Sci.*, vol. 51, no. 3, pp. 431–442, 2016, doi: 10.16943/ijhs/2016/v51i3/48846.
- [96] G. Browne, "stabilised interlocking rammed earth blocks: alternatives to cement stabilisation." Southampton Solent University, 2009.
- [97] S. Fang, H. Zhang, B. Zhang, and G. Li, "A study of Tung-oil-lime putty A traditional lime based mortar," *Int. J. Adhes. Adhes.*, vol. 48, pp. 224–230, 2014, doi: 10.1016/j.ijadhadh.2013.09.034.
- [98] P. Zhao, M. D. Jackson, Y. Zhang, G. Li, P. J. M. Monteiro, and L. Yang, "Material characteristics of ancient Chinese lime binder and experimental reproductions with organic admixtures," *Constr. Build. Mater.*, vol. 84, pp. 477–488, 2015, doi: 10.1016/j.conbuildmat.2015.03.065.
- [99] S. Fang, H. Zhang, B. Zhang, and G. Li, "A study of Tung-oil-lime putty A traditional lime based mortar," *Int. J. Adhes. Adhes.*, vol. 48, pp. 224–230, 2014, doi: 10.1016/j.ijadhadh.2013.09.034.
- [100] S. Fang, K. Zhang, H. Zhang, and B. Zhang, "A study of traditional blood lime mortar for restoration of ancient buildings," *Cem. Concr. Res.*, vol. 76, pp. 232– 241, 2015, doi: 10.1016/j.cemconres.2015.06.006.
- [101] J. N. Cooray, The Sigiriya Royal Gardens; Analysis of the Landscape Architectonic Composition, vol. 2012, no. 06. 2012. doi: 10.7480/a+be.vol2.diss6.
- [102] D. B. Dhanapala, "A short note on the Technique of the Sigiriya Pictures," 2015.
- [103] M. Ranaweera and H. Abeyruwan, "Materials Used in the Construction, Conservation, and Restoration of Ancient Stupas in Sri Lanka," pp. 2573–2586, 2004.
- [104] E. Navrátilová, E. Tihlaříková, V. Neděla, P. Rovnaníková, and J. Pavlík, "Effect of the preparation of lime putties on their properties," 2017. doi: 10.1038/s41598-017-17527-3.

- [105] S. Holmes and B. Rowan, "Lime Stabilized Construction, A Manual and Practical Guide." IOM, Pakistan, 2015.
- [106] C. Weddikkara and K. Devapriya, "Demand and Supply Trends and Construction Industry Development: A Case Study in the Sri Lankan Construction Industry," *Aust. J. Constr. Econ. Build.*, vol. 1, no. 1, pp. 91–105, 2001, doi: http://dx.doi.org/10.5130/AJCEB.v1i1.2283.
- [107] A. Windapo, "Examining The Trends In Building Material Prices: Built Environment Stakeholders' Perspectives," Univ. Cape Town, South Africa Abstr., vol. 3, no. 2, pp. 187–201, 2012.
- [108] G. Zhang, L. Mo, J. Sun, J. Chen, and J. Liu, "Preparation and Experimental Study on a New Type of Exterior Wall Putty," *Adv. Mater. Res.*, vol. 671–674, pp. 1914–1917, 2013, doi: 10.4028/www.scientific.net/AMR.671-674.1914.
- [109] E. Navrátilová, E. Tihlaříková, V. Neděla, P. Rovnaníková, and J. Pavlík,
 "Effect of the preparation of lime putties on their properties OPEN," 2017. doi: 10.1038/s41598-017-17527-3.
- [110] C. Udawattha and R. Halwatura, "Embodied Energy of Mud Concrete Block (MCB) Versus Brick and Cement Blocks," *Energy Build.*, vol. 126, no. May, pp. 28–35, 2016, doi: 10.1016/j.enbuild.2016.04.059.
- [111] P. Melià, G. Ruggieri, S. Sabbadini, and G. Dotelli, "Environmental impacts of natural and conventional building materials: A case study on earth plasters," J. *Clean. Prod.*, vol. 80, pp. 179–186, 2014, doi: 10.1016/j.jclepro.2014.05.073.
- [112] O. Ruskulis, "Mud Plasters and Renders," *Appropriate Technology*, vol. 26, no.1, Burnham, United Kingdom, Jun. 1999.
- [113] M. Emiroglu, A. Yalama, and Y. Erdogdu, "Performance of ready-mixed clay plasters produced with different clay / sand ratios," *Appl. Clay Sci.*, vol. 115, pp. 221–229, 2015, doi: 10.1016/j.clay.2015.08.005.
- [114] R. Deliniere, J. E. Aubert, F. Rojat, and M. Gasc-Barbier, "Physical, mineralogical and mechanical characterization of ready-mixed clay plaster," *Build. Environ.*, vol. 80, pp. 11–17, 2014, doi: 10.1016/j.buildenv.2014.05.012.
- [115] T. Santos, L. Nunes, and P. Faria, "Production of eco-efficient earth-based plasters: influence of composition on physical performance and biosusceptibility," J. Clean. Prod., vol. 167, pp. 55–67, 2017, doi:

10.1016/j.jclepro.2017.08.131.

- [116] R. Garcia and P. Prabhakar, "Bond interface design for single lap joints using polymeric additive manufacturing," *Compos. Struct.*, vol. 176, no. September, pp. 547–555, 2017, doi: 10.1016/j.compstruct.2017.05.060.
- [117] M. S. Lekshmi, S. Vishnudas, and D. G. Nair, "experimental study on the physical properties of mud mortar in comparison with the conventional mortars," *Int. J. Civ. Eng. Urban Plan.*, vol. 3, no. 2, pp. 127–135, 2016, doi: 10.5121/civej.2016.3211.
- [118] G. Frohnsdorff, L. W. Masters, and J. W. Martin, "An Approach to Improved Durability Tests for Building Materials and Components," in An Approach to Improved Durability Tests for Building Materials and Components, 1980, pp. 1–35.
- [119] A. G. Kerali and T. H. Thomas, "Simple durability test for cement stabilized blocks," *Build. Res. Inf.*, vol. 32, no. 2, pp. 140–145, 2004, doi: 10.1080/0961321032000148479.
- [120] G. A. H. H. Galkanda, S. A. A. Gunawardana, and R. U. Halwathura, "Analysis of the Performance of Existing Wall Finishers to Enhance the Durability of Different Walling Materials Against Rain Surface Erosion," *MERCon 2019 -Proceedings, 5th Int. Multidiscip. Moratuwa Eng. Res. Conf.*, pp. 400–405, 2019, doi: 10.1109/MERCon.2019.8818875.
- [121] M. Kameni and J. A. Orosa, "Case Studies in Thermal Engineering Building construction materials effect in tropical wet and cold climates : A case study of office buildings in Cameroon," *Case Stud. Therm. Eng.*, vol. 7, pp. 55–65, 2016, doi: 10.1016/j.csite.2016.01.007.
- [122] F. Wagner *et al.*, "Pan-tropical analysis of climate effects on seasonal tree growth," *PLoS One*, vol. 9, no. 3, pp. 20–22, 2014, doi: 10.1371/journal.pone.0092337.
- [123] X. Zhou, Y. Fu, L. Zhou, B. Li, and Y. Luo, "An imperative need for global change research in tropical forests," *Tree Physiol.*, vol. 33, no. 9, pp. 903–912, 2013, doi: 10.1093/treephys/tpt064.
- [124] G. A. H. H. Galkanda, C. Udawattha, G. Y. Jayasinghe, and R. U. Halwatura,"Comparative study of fungal and moss growth effect on different walling

materials in tropics," in *3rd Wayamba International Conference (WinC- 2018)*, 2018, p. 30.

- [125] O. Guilliitte, "Bioreceptivity: a new concept for building ecology studies," *Sci. Total Environ.*, vol. 167, no. 1–3, pp. 215–220, 1995, doi: 10.1016/0048-9697.
- [126] M. D'Orazio *et al.*, "Effects of water absorption and surface roughness on the bioreceptivity of ETICS compared to clay bricks," *Build. Environ.*, vol. 77, pp. 20–28, 2014, doi: 10.1016/j.buildenv.2014.03.018.
- [127] A. M. Pranjic, J. Mulec, T. Muck, A. Hlandnik, and A. Mladenovic, "The bioreceptivity of building stone," *Geophys. Res. Abstr.*, vol. 17, 2015.
- [128] C. P. Hoang, K. A. Kinney, R. L. Corsi, and P. J. Szaniszlo, "International Biodeterioration & Biodegradation Resistance of green building materials to fungal growth," *Int. Biodeterior. Biodegradation*, vol. 64, no. 2, pp. 104–113, 2010, doi: 10.1016/j.ibiod.2009.11.001.
- [129] N. Karagiannis, M. Karoglou, A. Bakolas, and A. Moropoulou, "Building Materials Capillary Rise Coefficient: Concepts, Determination and Parameters Involved," in *New Approaches to Building Pathology and Durability*, vol. 6, J. Delgoda, Ed. Singapore: Springer, 2016, pp. 27–44. doi: 10.1007/978-981-10-0648-7.
- [130] D. E. James and J. E. Kylander, *Techniques for Studying Bacteria and Fungi*, 8th editio. Carolina: Carolina Biological Supply Company, 1986.
- [131] Mycological Society of Japan, "Introduction to the World of Fungi," *Polymers* (*Basel*)., p. 466, 2015.
- [132] H. Viitanen *et al.*, "Moisture and biodeterioration risk of building materials and structures," *J. Build. Phys.*, vol. 33, no. 3, pp. 201–224, 2010, doi: 10.1177/1744259109343511.
- [133] M. Safiuddin, "Concrete Damage in Field Conditions and Protective Sealer and Coating Systems," *Coatings*, vol. 7, no. 90, pp. 1–22, 2017, doi: 10.3390/coatings7070090.
- [134] S. T. Methods, "Standard Test Methods for Physical Testing of Quicklime, Hydrated Lime, and," vol. i, 2016, doi: 10.1520/C0110-15E01.2.
- [135] ASTM International, ASTM C110-15, Standard Test Methods for Physical Testing of Quicklime, Hydrated Lime, and Limestone. West Conshohocken,

2016. doi: 10.1520/C0110-15E01.

- [136] G. A. H. H. Galkanda and R. U. Halwatura, "feasibility of incorporating biyagama drinking water treatment plant solid sludge in a construction material production process," in 9TH YSF Symposium, 2020, no. November, pp. 37–41.
- [137] C. Polakowski, M. Ryżak, A. Sochan, M. Beczek, R. Mazur, and A. Bieganowski, "Particle size distribution of various soil materials measured by laser diffraction—the problem of reproducibility," *Minerals*, vol. 11, no. 5, 2021, doi: 10.3390/min11050465.
- [138] J. R. Rosell, L. Haurie, A. Navarro, and I. R. Cantalapiedra, "Influence of the traditional slaking process on the lime putty characteristics," *Constr. Build. Mater.*, vol. 55, pp. 423–430, 2014, doi: 10.1016/j.conbuildmat.2014.01.007.
- [139] F. A. De Andrade, H. A. Al-qureshi, and D. Hotza, "Measuring and Modeling the Plasticity of Clays," *Mater. Res.*, vol. 13, no. 3, pp. 395–399, 2010.
- [140] M. Dahhou, M. El Moussaouiti, A. Benlalla, A. El Hamidi, M. Taibi, and M. A. Arshad, "Structural Aspects and Thermal Degradation Kinetics of Water Treatment Plant Sludge of Moroccan Capital," *Waste and Biomass Valorization*, 2016, doi: 10.1007/s12649-016-9513-5.
- [141] E. Nimwinya, W. Arjharn, and S. Horpibulsuk, "A sustainable calcined water treatment sludge and rice husk ash geopolymer," *J. Clean. Prod.*, vol. 119, pp. 128–134, 2016, doi: 10.1016/j.jclepro.2016.01.060.
- [142] A. L. G. Gastaldini, M. F. Hengen, M. C. C. Gastaldini, F. D. Amaral, M. B. Antolini, and T. Coletto, "The use of water treatment plant sludge ash as a mineral addition," *Constr. Build. Mater.*, vol. 94, pp. 513–520, 2015, doi: 10.1016/j.conbuildmat.2015.07.038.
- [143] B. E. E. Hegazy, H. A. Fouad, and A. M. Hassanain, "Brick Manufacturing From Water Treatment Sludge And Rice Husk Ash," *Aust. J. Basic Appl. Sci.*, vol. 6, no. 3, pp. 453–461, 2012.
- [144] N. H. Rodríguez *et al.*, "Re-use of drinking water treatment plant (DWTP) sludge: Characterization and technological behaviour of cement mortars with atomized sludge additions," *Cem. Concr. Res.*, vol. 40, pp. 778–786, 2010, doi: 10.1016/j.cemconres.2009.11.012.
- [145] R. A. Wuana and F. E. Okieimen, "Heavy Metals in Contaminated Soils: A

Review of Sources, Chemistry, Risks and Best Available Strategies for Remediation," *ISRN Ecol.*, 2011, doi: 10.5402/2011/402647.

- [146] T. A. Adagunodo, L. A. Sunmonu, and M. E. Emetere, "Data in Brief Heavy metals ' data in soils for agricultural activities," *Data Br.*, vol. 18, pp. 1847– 1855, 2018, doi: 10.1016/j.dib.2018.04.115.
- [147] J. Ogilo, O. Anam, and A. O. Yusuf, "Assessment of Levels of Heavy Metals in Paints from Interior Walls and Indoor Dust from Residential Houses in Nairobi City County, Kenya," *Chem. Sci. Int. J.*, vol. 21, no. 1, pp. 1–7, 2017, doi: 10.9734/CSJI/2017/37392.
- [148] O. Joseph, E. Emmanuel, and Y. Perrodin, "Ecological risk assessment of heavy metals in paint manufacturing effluents," in *Conference*, 2005, no. June 2014.
- [149] C. Huang, J. R. Pan, K. Sun, and C. Liaw, "Reuse of water treatment plant sludge and dam sediment in brick-making," *Water Sci. Technol.*, vol. 44, no. 10, pp. 273–277, 2001.
- [150] G. XU, J. ZOU, and G. LI, "Ceramsite Made with Water and Wastewater Sludge and its Characteristics Affected by SiO2 and Al2O3," *Environ. Sci. Technol.*, vol. 42, no. 19, pp. 7417–7423, 2008.
- [151] H. Chen, X. Ma, and H. Dai, "Reuse of water purification sludge as raw material in cement production," *Cem. Concr. Compos.*, vol. 32, no. 6, pp. 436–439, 2010, doi: 10.1016/j.cemconcomp.2010.02.009.
- [152] N. Waijarean, S. Asavapisit, and K. Sombatsompop, "Strength and microstructure of water treatment residue-based geopolymers containing heavy metals," *Constr. Build. Mater.*, vol. 50, pp. 486–491, 2014, doi: 10.1016/j.conbuildmat.2013.08.047.
- [153] H. M. Owaid, R. Hamid, and M. R. Taha, "Influence of thermally activated alum sludge ash on the engineering properties of multiple-blended binders concretes," *Constr. Build. Mater.*, vol. 61, pp. 216–229, 2014, doi: 10.1016/j.conbuildmat.2014.03.014.
- [154] P. Wang and D. Liu, "Physical and Chemical Properties of Sintering Red Mud and Bayer Red Mud and the Implications for Beneficial Utilization," *Materials* (*Basel*)., vol. 5, pp. 1800–1810, 2012, doi: 10.3390/ma5101800.
- [155] Y. Gapak, G. Das, U. Yerramshetty, and T. V. Bharat, "Laboratory

determination of volumetric shrinkage behavior of bentonites : A critical appraisal," *Appl. Clay Sci.*, 2016, doi: 10.1016/j.clay.2016.10.038.

- [156] A. Sales, F. Rodrigues, D. Souza, W. Nunes, A. Mendes, and C. Rosa, "Lightweight composite concrete produced with water treatment sludge and sawdust: Thermal properties and potential application," *Constr. Build. Mater.*, vol. 24, no. 12, pp. 2446–2453, 2010, doi: 10.1016/j.conbuildmat.2010.06.012.
- [157] P.-C. Aïtcin and R. J. Flatt, Science and Technology of Concrete Admixtures. Elsevier, 2016.
- [158] T. C. Fu, W. Yeih, J. J. Chang, and R. Huang, "The Influence of Aggregate Size and Binder Material on the Properties of Pervious Concrete," *Adv. Mater. Sci. Eng. Vol.*, vol. 2014, pp. 1–17, 2014.
- [159] S. S. Kamath, D. Sampathkumar, and B. Bennehalli, "A review on natural areca fibre reinforced polymer composite materials," *Ciência Tecnol. dos Mater.*, vol. 29, no. 3, pp. 106–128, 2017
- [160] A. Lotfi, H. Li, D. V. Dao, and G. Prusty, "Natural fiber-reinforced composites: A review on material, manufacturing, and machinability," J. Thermoplast. Compos. Mater., vol. 34, no. 2, pp. 238–284, 2019
- [161] M. Ali, A. Liu, H. Sou, and N. Chouw, "Mechanical and dynamic properties of coconut fibre reinforced concrete," *Constr. Build. Mater.*, vol. 30, pp. 814–825, 2012, doi: 10.1016/j.conbuildmat.2011.12.068.
- [162] F. McGonigle and P. A. Ciullo, "Paints & Coatings," in *industrial minerals and their uses*, Noyes Publication, USA, 1996, pp. 99–159.
- [163] S. R. Branch and Y. Branch, "Synthesis and Characterization of Styrene-Acrylic Binders and Their Application on Pigment Printing of Cotton and Polyester Textile Fabrics," *Asian J. Chem.*, vol. 21, no. 6, pp. 4871–4880, 2009.
- [164] C. Udawattha, D. E. De Silva, H. Galkanda, and R. Halwatura, "Performance of natural polymers for Stabilizing earth blocks," *Materialia*, vol. 2, no. August, pp. 23–32, 2018, doi: 10.1016/J.MTLA.2018.07.019.
- [165] H. Galkanda, H. Galabada, D. Dhammika.P, M. Malsha, N. Renuka, and H. Rangika, "Enhance Biyagama water treatment plant sludge properties with natural and synthetic binders," pp. 0–4, 2020.
- [166] K. V. Ramana and A. J. S. Raju, "Traditional and commercial uses of Litsea
glutinosa (Lour .) C . B . Robinson (Lauraceae)," *ournal Med. Plants Stud.*, vol. 5, no. 3, pp. 89–91, 2017.

- [167] L. C. B. Rob, N. T. Hien, T. D. Thang, D. N. Dai, and T. H. Thai, "Chemical composition of the leaf oil of Litsea glutinosa," *VNU J. Sci. Nat. Sci. Technol.*, vol. 26, pp. 161–164, 2010.
- [168] S. A. A. Gunawardanaa, G. A. H. H. Galkanda, and R. U. Halwatura, "Investigation of Fungus Growth and Moss Growth on Different Wall Care Putty Materials along with Different Walling Materials," *MERCon 2020 - 6th Int. Multidiscip. Moratuwa Eng. Res. Conf. Proc.*, pp. 96–101, 2020.
- [169] S. N. Araya, M. Meding, and A. A. Berhe, "Thermal alteration of soil physicochemical properties : a systematic study to infer response of Sierra Nevada climosequence soils to forest fires," SOIL, vol. 2, pp. 351–366, 2016.
- [170] G. A. H. H. Galkanda and R. U. Halwatura, "Thermal alteration of water treatment plant alum sludge: a systematic study of Biyagama water treatment plant alum sludge," 2020.
- [171] C. E. Boyd, "Aluminum Sulfate (Alum) for Precipitating Clay Turbidity from Fish Ponds," *Trans. Am. Fish. Soc.*, vol. 108, pp. 307–313, 1979.
- [172] C. Fewins, "The Pros and Cons of Different Construction Systems," pp. 1–11, 2006.
- [173] V. Vliet, "Shear tests on masonry panels; Literature survey and proposal for experiments," *Test*, 2004.
- [174] K. Uehara and M. Sakurai, "Bonding strength of adhesives and surface roughness of joined parts," J. Mater. Process. Technol., vol. 127, no. 2, pp. 178– 181, 2002.
- [175] B. Istegun and E. Celebi, "Triplet Shear Tests on Retrofitted Brickwork Masonry Walls," vol. 11, no. 9, pp. 1250–1255, 2017.
- [176] T. Tachibana *et al.*, "Cell membrane perforation with photosensitizer and a brushshaped soft-polymer sheet using a malignant glioma cell line," *Anticancer Res.*, vol. 35, no. 11, pp. 6069–6074, 2015.
- [177] M. Derikvand and H. Pangh, "A modified method for shear strength measurement of adhesive bonds in solid wood," *BioResources*, vol. 11, no. 1, pp. 354–364, 2016.

0	3	1		Sludge	1	Plant Pr	roduction	ion						Weather condition													
20	120 We	ek Da	Moistun Conteni	Shrinkag	Dry Density	Water	Sludge to decanter	A	Jum		Ŭ	me			Chl	orine			Poly	mer		Temp	(High)	Temp	o (Low)	Hun	nidity
			*	%	g/cm ³	m ³ /day	m ³ /day		7	P	re	Po	st	P	re	P	ost	Raw	Water	Sludge (Decanter	°c	°c	°c	°c	%	%
1	+	1		- 2/25	State.	186874	72	g/m ³ 10.8	g/m′	g/m²	g/m	g/m ³ 9.32	g/m°	g/m ² 2.16	g/m°	g/m ³ 1.89	g/m°	g/m ³ 0.09	g/m²	g/m ³ 32.71	g/m°	30	1250	26		76	
	-	2	_			188312 192990	84 133	11.8 12.18		0		7.77		2.11 2.05		1.73		0.1		33.8 26.13		31 31		25 24		78.5 75	
		4				193618	88 125	11.4		0		11.08		2.05		2.36		0.08		30.37		31		26		76	
		6				192430	63	12.15		0		9.48		2.07		2.12		0.09		37.65		32		25		71	
	We 0	8	68.48	69.48	1.27	191887	163	11.32	11.505714	0	0	10.92	9.777	1.99	2.01857	2.09	2.04429	0.09	0.087143	38.74	29.9243	32	32	26	24.71	58.5	65.5
		10	5			194506	94	10.62		0		9.29		1.97		1.94		0.09		33.12		31		24		68	
		13			-	193109	65	10.9		0		9.84		1.97		1.93		0.06	-	36.31		33		24	8	58	-
	W.S	1			1.11.21	191334 189640	170 94	10.83	Contraction (1997)	0		8.41		1.98	an services	1.75		0.06	i.	16.71 34.55	i Internet	32		25 24		50 49.5	-
1	an 0.	15	75.66	78.38	1.32	193627 193925	126	10.89 9.8	10.385714	0	0	6.73 7.98	7.813	1.94 1.94	1.95286	1.72	1.81	0.06	0.057143	24.11 36.82	31.18	31	31.43	22 23	24.57	60.5 64	60.357
		17				194238 194624	118 145	9.82 10.12	-	0		7.3 8.91		1.94 1.94		1.87		0.05		43.8 25.96		30 31		27 26		72	
	2	19				194135 193047	109 80	10.34 9.9		0		9.5 8.66		1.96 1.94	-	2.18		0.05		40.43		32		26		71 67	
	We	ek 21	78.53	81.17	1.34	194708 194558	132	9.67 9.34	9.27	0	0	7.39	8.454	1.93 1.94	1.94714	2.04	2.05143	0.05	0.05	24.83 38.57	35,7671	33	32.43	27	26	66.5 68	67.786
	0.	23				192458	112	8.86		0		10.03		1.97	1	1.91		0.05		38.4		32		26	a 8	67	
	-	25			_	195290	134	8.46		0		7.36	_	1.96	<u> </u>	2.02		0.05		35.26	<u> </u>	33		25		67.5	-
	260	27				194275	115	8.42		0		5.74		1.95		2.22		0.05		40.25		33		24		66.5	
	We	ek 28	78.32	84.8	1,32	194029 194585	191 174	8.46	8.405	0	0	9.77	7.738	2.01	2.00167	2.27	2.18	0.05	0.05	31.72 37.12	38.0383	31	31.17	24	24.33	64	64.333
		30				191288 193366	157 199	8.51 8.09		0		8.16		2.11 2.03		2.15		0.05		41.22 35.58		31 30	100	25	8 8	66 68.5	
45	-	2				194243 194428	139 126	8.02 8.01		0		8.63 8.59		2.05		2.11 2.15		0.05		32.45 28.59		33 32		25 25		65	
	1	3				193603 194763	79 88	8.05 7.96		0		7.69		2.04 2.03		2.16		0.05		34.28 42.76		33 34		25 25		66.5 55	
	We	ek 5	78.58	82	1 37	192510	78	8.18	8.75	0	0	7.04	8 296	2.11	2 13571	2.19	2 03143	0.05	0.045714	44.84	42 1471	33	32.43	26	24.85	63	63 786
	1	7			1.5050	194120	107	9	10508760	0		8.44		2.2		1.87		0.04		44.62		31	-	25		67	
		9				193025	158	9.60		0		9.78	<u> </u>	2.19		1.92		0.05		28.11		31		24		68	
		11				194714	152	9.43		0		9.08		2.19		2.05		0.05		31.32		34		23		61	
	We 2	ek 13	80.9	80.62	1.32	194743	113	9.51	9.6885714	0	0	7.09	7.884	2.16	2.20571	1.94	2,1	0.05	0.052857	26.61	30.9743	36	33.29	26	25.43	60 56.5	62.643
F	eb	14				194610 194969	250	9.87 9.93		0		7.73		2.2		2.18		0.05		22.01 28.6		33		26		67 72.5	
	-	16			-	194174 194647	98 96	9.6 9.49		0		6.97 8.07		2.21 2.28		2.14		0.05		39.31 37.69		33		26 26	8 8	70 65	
		18				180518 195855	99 232	10.17 9.68		0		9.54 7.59		2.63		2.31		0.05		39.27 17.47		33		24		59 60.5	
	3	20	63.39	59.82	1.3	195445 182172	155 92	9.21 10.42	9.4671429	0	0	7.9	8.167	2.6	2.58143	1.88	1.94714	0.05	0.05	29.03 31.87	29.8071	33	33.29	25 26	25.43	62	61.214
		22				195236 194830	129	9.25 8.05		0		8.19		2.62		1.76		0.05		28.63 24.69		34		27		59.5 59.5	
		24	-			194031 195132	105	9.11		0		5.5		2.53		2.05		0.04		29.69		33		26	1	61	
	We	ek 20	70.98	70.71	1,39	194811	120	9.25	9.1	0	0	7.79	7.202	2.52	2.515	1.81	1.97333	0.04	0.04	25.44	27.4733	32	32.33	27	26.33	69.5	64.167
		28				195179	146	8.41		0		6.58		2.5		1.81		0.04		26.44		32		26		66.5	
		1				195216	95	9.16		0		6.2		2.5		1.78		0.04		42.91		32		25		66	
		3				194573	124	9.69		0		10.9		2.57		2.02		0.06	į.	30.62		31		26		/0.5 69.5	
	We	ck 5	76,46	77.24	1.37	195579 197654	121	9.09 9.62	10.354286	0	0.0029	6.93 8.44	8.641	2.62 2.59	2.61714	2	2.08	0.05	0.055714	31.48 26.29	30.01	31 32	31.71	25 26	25.71	67 71	67
		7				196795 195575	150 135	9.92 12.33		0.01		8.49 9.07		2.59		2.04		0.06		25.51 25.95	1	32		27 26		68 60.5	
		8			-	196008 196303	117 116	10.79 10.05		0		9.29 8.63	-	2.71 2.62	\vdash	2.17		0.05		42.99 22.61	-	32	_	24 26		62.5 69.5	-
		10				148934 196804	177	10.23		0.01		9.68 7.23		2.91 2.58		2.12		0.05		22.85 27.58		32		27		67 72.5	
	We 2	tk 12	80.12	88.02	2.01	197482	127	11.08	10.718571	0	0.0029	8.91	7.989	2.58	2.63143	2.19	2.09714	0.06	0.058571	26.19	23.6557	32	32	27	27	68.5	69.143
		14				196760	141	11.01		0		7.35		2.59		2.03		0.07		25.89		33		28		67.5	
•	lar	16			-	196361	89	11.77		0		6.87		2.7		2.55	-	0.08	-	24.35		33		27		65.5	_
	Wo	17	in and in a		11000	190954 196802	224	12.21		0		9.76		2.6	arrend et	2.99	-	0.06	Constanting of	24.28	i.	34		26		68	
	3	19	78.61	80.21	1.43	194658 195161	142 200	11.36 10.75	11.402857	0	0	8.88 8.96	8.464	2.61	2.59143	2.23	2.45857	0.06	0.064286	23.24 23.4	24.2157	32	32.71	26	25.71	73	66.286
		21				193501 195122	191 78	10.77 10.98		0		6.98 8.95		2.58 2.62		2.46 2.45		0.07		26.01 24.5		33 32		25 25	8	58.5 64.5	
		23				195146 195062	133 128	10.89		0		12.89 10.44		2.6		2.37		0.06		16.36 23.15		32		26		66 70.5	6
	We	25 ak 26	87 18	83.29	1.45	192830 194798	110	10.8	10.682857	0	0.0029	8.23	9,046	1.81	1,89429	2.44	2.12286	0.06	0.06	22.58	24,6329	32	32 29	27	26.43	65 67.5	67 571
	4	27				195627	131	10.77		0		8.68		1.69		1.87		0.05		38.34		32		26		74	
	L	28				195930	139	10.85		0.01		7.28		1.68		1.96		0.06		28.19		33		28	8	63	
L	-	30				195440	214	11.17		0.01		5.77		1.67		1.92		0.05		27.94 28.05		33		26		69	

APPENDIX – I: Recorded data of physical properties, chemicals and climatic factors

				Sludge		Plant Pr	oduction	tion							W	eathe	r condit	tion	-								
2020) Week	Day	Moisture Content	Shrinkage	Dry Density	Water	Sludge to decanter	A	lum		Ú	me			Chl	orine			Poly	mer		Temp	(High)	Tem	p (Low)	Hum	nidity
			%	%	g/cm ³	m³/dav	m³/dav	<u> </u>	4	P	re	Po	st	Р	re	Р	ost	Raw	Water	Sludge (Decanter	°c	°c	°C	°c	%	%
		1						g/m²	g/m³	g/m³	g/m³	g/m³	g/m²	g/m³	g/m³	g/m°	g/m°	g/m³	g/m°	g/m³	g/m°	32		26		73	
	Week 1	3	84.72	85.09	1.53	194976 194449	137 203	12.58 13.04	12.886667	0	0.01	7.74 8.07	8.223	1.68 1.65	1.66667	2.64 2.67	2.63667	0.04	0.04	25.54 25.77	21.74	33 32 33	32.6	26 26 26	25.8	67 64.5 68	68.3
	-	5		-	-	193599	239	13.04		0.02		8.86		1.67		2.6		0.04		13.91		33		25		69 68	
		7				194841	118	12.43		0.02		7.85		1.68		3.15		0.05		25.28		32		27		67.5	
	Week	8			1.50	193650	134	11.6		0.02	0.0074	10.06		1.64	4.00	3.22		0.06		22.03	22.00	33		26		70	co 053
	2	10	84.57	84.14	1.53	195/74	132	10.99	11.511429	0	0.0071	8.67	8.929	1.61	1.68	2.93	3.02714	0.05	0.051429	22.2	23.99	31	32.14	25	25.71	66	69.857
		11	1			158967	170	11.75		0		9.41		1.91		3.07		0.05	1	25.59		32		25	1	68	
		12				194754	121	11.7		0		9.42		1.63		2.67		0.05		26.17		32		25		74.5	
		13				193858	150	11.89		0	8	9.38	3	1.65		2.46	a 1	0.05		25.26		32		25		69 70 F	
	20 8	15				192767	100	11.63		0	2	9.95	5	1.69		2.76		0.07		42.51		32		25		69.5	
Apr	Week 3	16	83.6	83.51	1.46	192589	129	11.16	12.254286	0	0	8.62	9.454	1.69	1.66571	2.65	2.65	0.07	0.067143	40.32	32.4129	32	32.14	25	25.14	67	72
		17	-			193704	98	12.55	-	0	8	9.82		1.66		2.67		0.07	1	40.65	1	31		25		79 72 E	
		19				194224	81	12.97		0	S.	9.25		1.67		2.03	8 8	0.07		26.02		32		25		75.5	
		20				193491	137	14.38		0.04		9.69		1.72		2.79		0.06		23.24		32		25		78	
		21				193595	160	13.52	-	0.04	8	12.1		1.72		2.98		0.06		26.84		33		25		76 82.5	
	Week	23	76.47	93.31	4.44	189760	139	13.28	12.751429	0.01	0.0129	10.9	10.52	1.75	1.70429	2.52	2.71571	0.06	0.062857	25.67	25.3643	32	32	25	25	79	80.429
	4	24				193980	114	12.7	1	0		10.95		1.67		2.6		0.07]	24.37		31		24		84.5	
		25				193906	225	11.89	-	0	8	9.95		1.68		2.34		0.06		25.32		33		26		82.5	
		27				194498	141	12.47		0.01		9.68		1.67		2.55		0.07		23.15		33		27		80	
	Week	28	77.97	92.63	4.23	194268	118	12.75	12.67	0.02	0.0175	8.75	8.645	1.69	1.67	2.51	2.445	0.06	0.0625	25.68	25.78	33	33	27	27	78.5	78.875
	5	29				195669	85	12.79		0.02		7.66		1.66		2.37		0.06		27		33		27		78.5	
	-	1				195398	93	12.59		0.02		7.67		1.67		1.45		0.06		27.29		33		27		75	
		2				194285	114	12.65		0.02		8.19		1.69		1.1		0.06		27.57		33		27		73	
	-	3		Î		195681	164 0.9	12.29		0.02		12.86		1.69		2.41	-	0.06		27.34		33		2/		73	
		5				194327	140	12.36		0	İ	9.48		1.69		2.42		0.06		27.68		34		27		72	
	Week	6				193346	114	12.11		0		8.57		1.69		2.49		0.06		26.93		30		26		80	
	1	7	80.7	93.62	4.39	191270	142	11.87	12.165714	0	0	9.78	9.821	1.68		2.49	2.46	0.06	0.06	27.33	27.41	31	31.86	26	26.43	72.5	74.929
		9				193806	113	12.35		0		9.62		1.68		2.46		0.06	1	27.47		32		27	1	78	
		10			-	195108	106	12.16		0		8.84		1.67		2.44		0.06		27.67		33		26		74	
		11				193965	109	12.7		0.01	1	8.67		1.7		2.35		0.05		26.91		32		26		74	
	Wook	13	Sectore 1	1 children		194366	106	13.01		0	a and the second	10.9		1.7		2.8		0.08		27.07		31		25		61	
	2	14	69.17	88.8	3.77	193171	146	11.93	13.182857	0	0.9729	13.58	11.17	1.64	1.66143	2.77	2.68714	0.08	0.081429	26.23	26.4929	32	30.71	27	26.86	72	77.929
May		16				194246	152	16.29		2.82		12.64		1.6		2.73		0.08		25.9		31		20		79	
12242543		17				190979	121	13.82		3.97		9.08		1.7		2.69		0.11		25.71		31		29		82.5	
		18				195668	106	13.15	-	3.79		0.93	-	1.65		2.89		0.08	-	26.3		31		29		58	
		20	_			195633	171	13.2		5.09		10.94		1.78		2.32	2 2	0.00		25.09		32		29		71	
	3	21	62.31	86.34	4.06	192601	135	13.28	13.197143	5.03	4.2343	11.93	13.39	1.88	1.86	2.44	2.51286	0.08	0.082857	24.84	25.8114	32	31.43	28	28	76	73.857
		22				194983	128	12.89	-	3.37		23.12		2.07		2.17		0.08		25.1		31		25		79	
		24				195899	200	12.86		4.55		13.19		2.17		2.22		0.08	1	24.83		32		29		81	
		25				193825	132	14.08	-	3.7		10.08		2.21		2.06		0.11		26.68		31		29		84	
		20				195125	168	14.24	-	3.57		12.36	5	2.10		1.69		0.1		27.07		30		20		83.5	
	Week 4	28	75	90.95	4.05	192734	221	14.71	13.83	3.65	3.54	11.17	11.73	2.2	2.26571	2.05	1.86143	0.11	0.092857	9.1	24.4914	30	30.86	24	26.71	78.5	79.143
	8	29	_			195272	189	13.91		3.18	6	12.12		2.3		2.23		0.08		26.78		31		28		77	
		31				194939	153	13.28	-	3.65		9.1		2.39		1.35		0.08		25.94		32		27		75	
		1				194603	228	11.78		4.74	8	7.54		2.41		1.18		0.08		25.06		31		27		74.5	
		2				194547	206	12.22		1.8	ŝ	10.77	8	2.55		1.16	8 8	0.07	-	23.91		31		27		72.5	
	Week	4	75.24	91.72	4.05	198277	163	10.67	11.58	0.03	0.9471	8.98	10.08	2.42	2.43	0.82	1.56286	0.07	0.075714	24.88	25.1157	32	31.43	26	26.57	69.5	74.5
	1	5				193168	191	9.94]	0.02		11.17		2.44		1.24		0.08	1	25.58		32		28		70.5	
		6				192366	123	12.1	-	0	8	9.58		2.37		2.32		0.08		25.65		31		27		80	
		8				193504	133	12.89		0	_	12.07		1.7		1.35		0.08	-	25.45	-	30	-	24		79.5	
		9				195723	122	12.98		0.05		11.08		1.72		1.59		0.08]	25.6		32		28		77	
	Week	10	79.45	93.47	4.24	193058	133	12.16	12.49	0.02	0.0329	9.73	12.95	2.02	1.87714	1.3	1.45857	0.08	0.082857	25.98	25,8557	31	31	28	27.43	79.5	77.071
	2	12	13.13	23.07		194158	166	12.56		0.02	0.0525	10.48	12100	1.92		0.77	215057	0.09	0.002037	26.08	23.0357	31	-	27	27112	80	11.011
		13				193969	137	11.85		0.05		17.98		2.06		1.1		0.09		25.8		31		28		74	
Jun	\vdash	14		-		194/45	215	12.51	-	0.06	-	20.41	-	2.31	-	2.98	-	0.08	-	25.86	-	31		29	-	73.5	-
		16				195185	156	10.77	1	0		19.18		2.37		1.56	1	0.08	1	26.29		30		28	1	81	
	Week	17	76.47	01.07	2.07	195724	116	10.4	11 202057	0	0.0111	17.12	12.25	2.28	2 27442	1.45	0.04574	0.08	0.00	25.71	15 3730	31	21.20	28	10.14	78.5	75 074
	3	18	/6.47	91.34	3.97	195239	136	11.35	11.392857	0	0.0114	8.19	13.35	2.13	2.2/143	0.98	u.94571	0.08	0.08	25.84	25.1729	32	51.29	28	28.14	73.5	/5.071
		20	1			196046	130	11.97	1	0	1	8.16		2.29	1	0.18	1	0.08	1	26.02	1	31		28	1	73	
	\vdash	21			-	196913	134	12.01		0		7.53		2.26		0.23	-	0.08	-	25.99		31		28		72	
		22				19601/	215	11.01		0.07		9.15		2.23		2		0.08		25.88		32		28		69	
	Wool	24	Same	Non-	Same	197036	121	10.45		0.09		7.96	-	2.29		1.88	-	0.09		24.1		31		27		78	
	4	25	68.49	87.12	3.74	198222	134	10.07	11.461429	0.11	0.0829	7.44	8.136	2.16	2.14857	1.89	1.75	0.08	0.087143	24.21	25.4486	31	30.86	28	27.43	79	74.929
		27				194318	136	13.38	1	0.05		7.58		2.17		1.75	1	0.09	1	24.55		31		27		72.5	
	1	28	1	1	1	194862	206	11.99	1	0.07	1	8.42	1	1.98]	1.85	Ĩ	0.09	1	26.62		31		27	1	79	

APPENDIX – II: Regression analysis of shrinkage

Regression Analysis: Sludge Shrinkage versus Chemicals Apply in DWTP (Alum, Lime, Chlorine & Polymer), Highest Temperature of the day, Lowest Temperature of the day & Humidity

Regression Equation

Shrinkage = 62.7 - 0.31 Alum - 0.77 Lime - 2.90 Chlorine + 0.030 Polymer - 1.26 Highest Temperature of the day + 0.07 Lowest Temperature of the day + 1.163 Humidity

Coefficients

Term	Coef	SE Coef	T-Value	P-Value	VIF
Constant	62.7	86.9	0.72	0.480	
Alum	-0.31	1.80	-0.17	0.866	6.05
Lime	-0.77	1.08	-0.71	0.486	3.10
Chlorine	-2.90	2.37	-1.22	0.238	1.31
Polymer	0.030	0.420	0.07	0.944	3.37
Highest Temperature of the day	-1.26	1.86	-0.68	0.507	1.55
Lowest Temperature of the day	0.07	1.81	0.04	0.971	2.97
Humidity	1.163	0.374	3.11	0.006	4.17

Model Summary

S	R-sq	R-sq(adj)	R-sq(pred)
5.44689	69.42%	56.83%	39.32%

Analysis of Variance

Source	DF	Adj SS	Adj MS	F-Value	P-Value
Regression	7	1145.13	163.590	5.51	0.002
Alum	1	0.87	0.875	0.03	0.866
Lime	1	15.05	15.053	0.51	0.486
Chlorine	1	44.35	44.346	1.49	0.238
Polymer	1	0.15	0.149	0.01	0.944
Highest Temperature of the day	1	13.65	13.646	0.46	0.507
Lowest Temperature of the day	1	0.04	0.042	0.00	0.971
Humidity	1	286.12	286.125	9.64	0.006
Error	17	504.37	29.669		
Total	24	1649.49			

Fits and Diagnostics for Unusual Observations

Obs Shrinkage Fit Resid Std Resid

7 59.82 72.10 -12.28 -2.60 R

R Large residual

Durbin-Watson Statistic

Durbin-Watson Statistic = 1.55653

APPENDIX – III: Regression analysis of moisture content

Regression Analysis: Sludge Moisture content versus Chemicals Apply in DWTP (Alum, Lime, Chlorine & Polymer), Highest Temperature of the day, Lowest Temperature of the day & Humidity

Regression Equation

Moisture content = 85 - 0.67 Alum - 0.78 Lime - 3.20 Chlorine - 0.316 Polymer + 1.88 Highest Temperature of the day - 2.38 Lowest Temperature of the day + 0.432 Humidity

Coefficients

Term	Coef	SE Coef	T-Value	P-Value	VIF
Constant	85	101	0.84	0.411	
Alum	-0.67	2.09	-0.32	0.754	6.05
Lime	-0.78	1.26	-0.62	0.542	3.10
Chlorine	-3.20	2.75	-1.16	0.261	1.31
Polymer	-0.316	0.488	-0.65	0.526	3.37
Highest Temperature of the day	1.88	2.16	0.87	0.397	1.55
Lowest Temperature of the day	-2.38	2.10	-1.13	0.273	2.97
Humidity	0.432	0.435	0.99	0.334	4.17

Model Summary

S	R-sq	R-sq(adj)	R-sq(pred)
6.32508	22.83%	0.00%	0.00%

Analysis of Variance

Source	DF	Adj SS	Adj MS	F-Value	P-Value
Regression	7	201.217	28.745	0.72	0.658
Alum	1	4.065	4.065	0.10	0.754
Lime	1	15.481	15.481	0.39	0.542
Chlorine	1	54.023	54.023	1.35	0.261
Polymer	1	16.782	16.782	0.42	0.526
Highest Temperature of the day	1	30.242	30.242	0.76	0.397
Lowest Temperature of the day	1	51.322	51.322	1.28	0.273
Humidity	1	39.538	39.538	0.99	0.334
Error	17	680.112	40.007		
Total	24	881.329			

Fits and Diagnostics for Unusual Observations

	Moisture			
Obs	content	Fit	Resid	Std Resid
7	63.40	76.92	-13.52	-2.46 R

R Large residual

Durbin-Watson Statistic

Durbin-Watson Statistic = 1.25002

APPENDIX – **IV: One-way ANOVA: Bonding strength of putty mixtures with stabilizers**

One-way ANOVA: Bonding strength versus Mix

Method

 $\begin{array}{ll} \mbox{Null hypothesis} & \mbox{All means are equal} \\ \mbox{Alternative hypothesis} & \mbox{Not all means are equal} \\ \mbox{Significance level} & \mbox{α} = 0.05 \\ \end{array}$

Equal variances were assumed for the analysis.

Factor Information

 Factor Levels Values

 Mix
 10 Mix 10, Mix 2, Mix 3, Mix 4, Mix 5, Mix 6, Mix 7, Mix 8, Mix 9

Analysis of Variance

Source	DF	Adj SS	Adj MS	F-Value	P-Value
Mix	9	4.18800	0.465333	×	×
Error	20	0.00000	0.000000		
Total	29	4.18800			

Model Summary

<u>S R-sq R-sq(adj) R-sq(pred)</u> 0 100.00% 100.00% 100.00%

Means

 Mix
 N
 Mean
 StDev
 95% Cl

 Mix 1
 3
 1.000
 0.000
 (1.000, 1.000)

 Mix 1
 3
 2.300
 0.000
 (2.300, 2.300)

 Mix 2
 3
 1.300
 0.000
 (1.300, 1.300)

 Mix 3
 3
 1.500
 0.000
 (2.000, 2.000)

 Mix 4
 3
 2.000
 0.000
 (1.200, 1.200)

 Mix 5
 3
 1.200
 0.000
 (1.200, 1.200)

 Mix 6
 3
 1.500
 0.000
 (1.500, 1.500)

 Mix 7
 3
 1.800
 0.000
 (1.400, 1.800)

 Mix 8
 3
 1.400
 0.000
 (1.400, 1.400)

 Mix 9
 3
 1.800
 0.000
 (1.400, 1.400)

Pooled StDev = 0

Tukey Pairwise Comparisons

Grouping Information Using the Tukey Method and 95% Confidence

Mix	Ν	Mean	<u>}</u>		Gro	bu	ping	
Mix 10	3	2.300	А					
Mix 4	3	2.000		В				
Mix 9	3	1.800			С			
Mix 7	3	1.800			D			
Mix 6	3	1.500				Ε		
Mix 3	3	1.500					F	
Mix 8	3	1.400					G	
Mix 2	3	1.300					н	
Mix 5	3	1.200					1	Ľ.
Mix 1	3	1.000						J

APPENDIX - V: One-way ANOVA: SOF of putty mixtures with stabilizers

One-way ANOVA: Scale off versus Mix

Method

Null hypothesis All means are equal Alternative hypothesis Not all means are equal Significance level $\alpha = 0.05$

Equal variances were assumed for the analysis.

Factor Information

Factor Levels Values 10 Mix 1, Mix 10, Mix 2, Mix 3, Mix 4, Mix 5, Mix 6, Mix 7, Mix 8, Mix 9 Mix

Analysis of Variance

Source DF Adj SS Adj MS F-Value P-Value *

Mix 9 83015.7 9223.97 Error 20 0.0 0.00 Total 29 83015.7

Model Summary

S R-sq R-sq(adj) R-sq(pred) 0 100.00% 100.00% 100.00%

Means

Mix	Ν	Mean	StDev	95% CI
Mix 1	3	265.1	0.0	(265.1, 265.1)
Mix 10	3	107.5	0.0	(107.5, 107.5)
Mix 2	3	257.7	0.0	(257.7, 257.7)
Mix 3	3	244.2	0.0	(244.2, 244.2)
Mix 4	3	232.4	0.0	(232.4, 232.4)
Mix 5	3	239.7	0.0	(239.7, 239.7)
Mix 6	3	184.4	0.0	(184.4, 184.4)
Mix 7	3	153.8	0.0	(153.8, 153.8)
Mix 8	3	243.2	0.0	(243.2, 243.2)
Mix 9	3	143.2	0.0	(143.2, 143.2)

Pooled StDev = 0

Tukey Pairwise Comparisons

Grouping Information Using the Tukey Method and 95% Confidence

Mix	Ν	Mean	Grouping
Mix 1	3	265.1	A
Mix 2	3	257.7	В
Mix 3	3	244.2	С
Mix 8	3	243.2	D
Mix 5	3	239.7	E
Mix 4	3	232.4	F
Mix 6	3	184.4	G
Mix 7	3	153.8	н
Mix 9	3	143.2	1
Mix 10	3	107.5	J

APPENDIX -	- VI : 2	Properties	of exi	sting v	vall j	putty	products
-------------------	-----------------	------------	--------	---------	--------	-------	----------

	Product Deta	ils		1	01			Pro	perties		0	
Wall care putty	Raw Materials Used	Putty form	pН	Wet Density	Transverse Strength	Compressiv e Strength (28 days)	Whiten ess Index	Pot Life	Water Demand By Volume	Surface Hardness	Maximum Thickness	Coverage
P1	White cement, redispersible pollymer, functional additives	Powder							36 - 40 %		1.5 mm (2 Coats)	20 - 25 sqfeets/kg On Single Coat (Varies with Surface Quality)
P2	White cement, Dolomite	Powder			1.98 N/mm	11 kg/cm	94%	180 min	30 - 35 %	2.8 N	3 mm (2 Coats)	16 sqfeets/kg On Single Coat (Varies with Surface Quality)
Р3	CaCO ₃ , White cement	Powder							35 - 40%		3 mm	12.5 sqfeets/kg On Single Coat (Varies with Surface Quality)
P4	CaCO ₃ , White cement, CMG	Powder							35 - 40 %		3 mm	12.5 sqfeets/kg On Single Coat (Varies with Surface Quality)
P5	Filler - Graded silica sand, CaCO3 Binder - White cement, powder polymers Additives - Water soluble powder polymers to improvr workability, water retantion, adhesion and for more durability	Powder	7	1700 kg/m ³				90 - 120 min	40 -45 %		2 - 2.5 mm	16 - 21.5 sqfeets/kg On Single Coat (Varies with Surface Quality)
P6	White cement, redispersible pollymer, functional additives	Powder							40 - 50 %		1.5 mm (2 Coats)	10 - 15 sqfeets/kg On two Coats (Varies with Surface Quality)
P7	Birla White Cement, high quality extra HP polymers, specialty chemicals and mineral fillers etc.	Powder				3.5 - 7.5 N/mm ²			36 - 40 %		1.5 mm (2 Coats)	20 - 25 sqfeets/kg On Single Coat (Varies with Surface Quality)
P8	White cement	Powder									1.5 mm (1st coat) 1mm (2nd coat)	14 sqfeets/kg On two Coats (Varies with Surface Quality)
Р9	Filler - White cement Pigmentation - Titanium Dioxide, Magnesium, Aluminium Silicate Solvent - Water, Ethylene Glycol	Paste							No need to dialute		1.5 mm	20 - 25 sqfeets/kg On Single Coat (Varies with Surface Quality)

*Commercial product name of the wall care putty is not mentioned

APPENDIX – VII: One-way ANOVA: Bonding strength of putty mixtures

One-way ANOVA: Bonding strength versus Mix

Method

Equal variances were assumed for the analysis.

Factor Information

 Factor Levels Values

 Mix
 5 Mix 1, Mix 2, Mix 3, Mix 4, Mix 5

Analysis of Variance

 Source
 DF
 Adj
 SS
 Adj
 MS
 F-Value
 P-Value

 Mix
 4
 3.7227
 0.93067
 11.44
 0.001

 Error
 10
 0.8133
 0.08133
 0.08133

 Total
 14
 4.5360
 4.5360
 4.5360

Model Summary

 S
 R-sq
 R-sq(adj)
 R-sq(pred)

 0.285190
 82.07%
 74.90%
 59.66%

Means

 Mix
 N
 Mean
 StDev
 95% Cl

 Mix 1
 3
 3.000
 0.000
 (2.633, 3.367)

 Mix 2
 3
 2.6667
 0.1528
 (2.2998, 3.0335)

 Mix 3
 3
 1.800
 0.500
 (1.433, 2.167)

 Mix 4
 3
 1.800
 0.265
 (1.433, 2.167)

 Mix 5
 3
 1.933
 0.252
 (1.566, 2.300)

Pooled StDev = 0.285190

Tukey Pairwise Comparisons

Grouping Information Using the Tukey Method and 95% Confidence

Mix N Mean Grouping

 Mix 1
 3
 3.000 A

 Mix 2
 3
 2.6667 A
 B

 Mix 5
 3
 1.933
 B
 C

 Mix 4
 3
 1.800
 C

 Mix 3
 3
 1.800
 C

APPENDIX - VIII: One-way ANOVA: SOF of putty mixtures

One-way ANOVA: Scale off versus Mix

Method

Equal variances were assumed for the analysis.

Factor Information

 Factor Levels Values

 Mix
 5 Mix 1, Mix 2, Mix 3, Mix 4, Mix 5

Analysis of Variance

 Source
 DF
 Adj
 SS
 Adj
 MS
 F-Value
 P-Value

 Mix
 4
 76834
 19209
 15.81
 0.000

 Error
 10
 12150
 1215
 1215

 Total
 14
 88985
 1215

Model Summary

 S
 R-sq
 R-sq(adj)
 R-sq(pred)

 34.8574
 86.35%
 80.88%
 69.28%

Means

Mix	Ν	Mean	StDev	95% CI
Mix 1	3	11.3110	0.0018	(-33.5300, 56.1521)
Mix 2	3	162.5	42.5	(117.7, 207.4)
Mix 3	3	224.19	14.45	(179.35, 269.03)
Mix 4	3	173.1	54.4	(128.2, 217.9)
Mix 5	3	120.7	33.1	(75.8, 165.5)

Pooled StDev = 34.8574

Tukey Pairwise Comparisons

Grouping Information Using the Tukey Method and 95% Confidence

 Mix
 N
 Mean Grouping

 Mix 3
 3
 224.19 A

 Mix 4
 3
 173.1 A
 B

 Mix 2
 3
 162.5 A
 B

 Mix 5
 3
 120.7
 B

 Mix 1
 3
 11.3110
 C